# POLITECNICO DI TORINO

Doctoral School Production Systems and Industrial Design

Ph.D. Thesis

# Manufacturing Execution Systems for lean, adaptive production processes



Tutors: prof. Franco Lombardi dr. Frédéric Segonds

> Ph.D. Candidate: Gianluca D'Antonio

April 2016

# Contents

1	Intr	oduction	9
	1.1	The need for information	10
	1.2	Lean manufacturing	11
	1.3	Information tools in manufacturing	14
		1.3.1 Enterprise Resource Planning	14
		1.3.2 Product Lifecycle Management	15
		1.3.3 Manufacturing Execution Systems	17
	1.4	New requirements for future factories	19
	1.5	Aim of this work	20
2	Ma	nufacturing Execution Systems	23
	2.1	Definition of MES	24
	2.2	State of the art	26
		2.2.1 MES market	27
		2.2.2 MES intellectual property rights	27
		2.2.3 MES scientific literature	29
3	Res	earch methodology	33
	3.1	Introduction	34
	3.2	General description of a manufacturing process	34
	3.3	Methodology for data analysis	38
	3.4	A schematic tool for the methodology	40
4	ME	S for monitoring and control of a finishing operation	43
	4.1	Introduction	44
	4.2	State of the art	45
	4.3	Description of the process	46
	4.4	Monitoring and control system	47
		4.4.1 The mathematical technique	47
	4.5	Results	52
	4.6	Alternative positioning technique	55
	4.7	The role of MES.	58

### Contents

		4.7.1	Integration between the monitoring and control	
			system and a MES	. 61
		4.7.2	Integration between MES and PLM	. 61
	4.8	The s	upport to lean manufacturing	. 62
	4.9	Concl	usions	. 63
5	MES	5 for m	nonitoring of product geometry failure	. 65
	5.1	Introd	luction	. 66
	5.2	State	of the art	. 67
		5.2.1	AM processes	. 67
		5.2.2	AM methodologies	. 69
		5.2.3	MES for Additive Manufacturing	. 70
	5.3	Descr	iption of the process	. 72
	5.4	Moni	toring and control system	. 73
	5.5	Resul	ts	. 75
	5.6	The re	ole of MES	. 78
		5.6.1	Integration between MES and DFAM	. 78
	5.7	The s	upport to lean manufacturing	. 80
	5.8	Concl	lusions	. 80
6	MES	5 for st	oring processes and warehouse management	. 83
	6.1	Introc	luction	. 84
	6.2	State	of the art	. 85
		6.2.1	Techniques for performance evaluation	. 85
		6.2.2	Information tools for warehouses	. 86
	6.3	Descr	iption of the process	. 87
	6.4	Analy	tical evaluation of warehouse performance	. 89
		6.4.1	Models for systems with one shuttle	. 93
		6.4.2	Models for systems with two shuttles	. 94
	6.5	Resul	ts of the analytical model	. 95
	6.6	Deplo	byment of Discrete Event Simulation	. 97
	6.7	Resul	ts of the simulation model	. 100
	6.8	The re	ole of MES	. 103
	6.9	The s	upport to lean manufacturing	. 104
	6.10	Concl	lusions	. 105
7	MES	5 for th	ne management of automated vehicles traffic	. 107
	7.1	Intro	luction	. 108
	7.2	State	of the art	. 109
	7.3	Descr	iption of the process	. 110
	7.4	Techn	ique to optimize a cellular layout	. 110
	7.5	Resul	ts of the layout optimization	. 115
	7.6	Discre	ete Event Simulation for shop-floor traffic management	. 117
	7.7	Resul	ts of the simulation	. 119
	7.8	The re	ole of MES	. 122

4

Contents 5			
	7.9 7.10	The support to lean manufacturing124Conclusions125	
8	Con	iclusions	
	8.1	Introduction	
	8.2	Summary of the work 128	
	8.3	Conclusive remarks	
	8.4	Future work	
	8.5	Vision for the Factory of the Future	
		8.5.1 The role of the human	
Α	Aut	omated warehouses: results of the simulations	
В	List	of abbreviations	
List of Figures			
List of Tables			
<b>References</b>			

# Abstract

In order to deal with global competition, industries have undertaken many efforts directed to improve manufacturing efficiency. From a broad perspective, the adopted approaches could be classified in two categories:

- the simplification of manufacturing processes and relative control systems, leading to lean manufacturing methododologies and techniques;
- 2. the massive deployment of information tools and computational algorithms, aiming to plan and control all the activities in detail, in spite of system complexity.

For several years, these two approaches have been assumed to be mutually exclusive; nevertheless, information collection and analysis are mandatory to define improvement strategies and assess their impact; therefore, the deployment of lean manufacturing methodologies cannot exclude the integration of Information Technology (IT) tools.

The aim of this work is to investigate on methodologies and techniques adoptable to improve the efficacy of Manufacturing Execution Systems (MES), a class of software that allows data exchange between the shop-floor and the organizational levels, enabling the implementation of the lean manufacturing approach.

Today, the feedback information in the available MES mainly consists in key performance indicators, such as cycle time, work in process and resources utilization. Beside this, MES requires the integration of functionalities for process monitoring and control, aiming at the reduction of wastes and supporting continuous improvement. Hence, mathematical techniques able to analyze data in real-time and provide useful information to adaptively control the process are studied in this work. To provide the evidence of the feasibility and effectiveness of the approach, as well as the independence from any specific manufacturing technology, different case studies, both in the fields of subtractive and additive manufacturing, have been developed. In the former, a technique for the automatic alignment of a spur gear has been studied: geometrical measurements are acquired and analyzed in realtime to provide the values for two feasible part rotations resulting in the gear configuration with minimum positioning error. Such gears are manufactured for applications in aeronautics, and the deployment of this automation system is particularly significant because of the tight tolerances to be satisfied. The latter case study deals with a Fused Deposition Modeling process: an algorithm able to monitor part surface accuracy and identify defects has been developed. This methodology allows to evaluate in real-time whether the quality of the part is satisfactory or not; in case of negative response, the process can be stopped avoiding material loss. The implemented techniques enable product quality improvement, as well as the reduction of wasted material and time. Nevertheless, the deployment of such information only for process control purposes is restrictive; a framework to use this knowledge for supporting the design and the continuous improvement of a product or a process is presented.

Furthermore, two case studies have been dealt to extend the application of MES tools from manufacturing operations to ancillary services. The first one is in the field of automated warehouses: a combined approach made of mathematical models and simulations has been developed. Analytical tools have been defined to evaluate the average performance of a system in simple, pre-determined situations; conversely, the simulation tool aims at a higher detail level of assessment, since in the real shop-floor deployment, different, composite storage and retrieval activities can take place. In the second case-study, mathematical models and simulation are used to support the re-design of a manufacturing process; the focus is on the transport of items through the line, performed by automated vehicles. The mathematical model has been developed to identify the optimal layout of the workstations; simulations are used to evaluate the tasks to be performed by the automated vehicles and the resulting performance. In both the applications, the deployment of simulation tools allows to evaluate complex or even unexpected scenarios by predicting the behavior of a system, preventing criticalities, and evaluating the impact of a change in the process. The management criteria can be adapted according to the features of the real situation to be faced; this leads to better exploit the available resources, to improve productivity and identify waste sources, consistently with the lean paradigm.

# Chapter 1 Introduction

Abstract Today manufacturing companies experience several challenges, such as the growing complexity of their processes and supply networks, cost pressures, increasing customer expectations for quality, lead time, and customization. In order to perform profitable production processes and improve competitiveness, different actions can be undertaken. Among them, one approach is the implementation of lean manufacturing practices, to identify and eliminate non-value added operations and sources of waste. Another strategy is the deployment of information tools, to better manage and control the production process. For long time, these two paradigms have been considered mutually exclusive; conversely, this research work aims at developing a framework for their integration. In this introductory Chapter, an overall review of the two approaches is provided. Further, the key directions enabling factories to remain competitive in future, are presented: they are all driven by the development of innovative IT tools.

# 1.1 The need for information

Today enterprises are driven by a market demand characterized by fierce competition, rapid pace of business and continually compressed time schedules. On the one hand, manufacturing is characterized by shortened production cycles and reduced batch sizes; on the other hand, the variety of product types and their customization is increasing, as well as customer demands rapidly change. Hence, to maintain and improve their competitive advantage, leading organizations in different industrial sectors need to improve process optimization and efficiency.

One of the initiatives that a company may undertake to improve its competitiveness is the implementation of lean manufacturing practices. The term *Lean manufacturing* has been first introduced by Womack et al. (1990) to describe the working philosophy deployed in Japanese companies, with particular concern for Toyota. The essence of this methodology is the elimination of waste and non-productive process, in order to focus on value added operations and produce high-quality products, at the customers demand pace, with little waste.

Another approach is the deployment of automation and IT tools, which allow to improve process planning and control, as well as to enhance the performance of each step of the manufacturing process. The landscape of the existing software classes and their purposes has been changing over the years, and is still evolving at a high pace. Today, the focus is on the integration and the communication between different information tools and among systems deployed by different companies (for example, among firms belonging to the same supply chain).

For several years, lean manufacturing has been considered as opposed to the deployment of IT tools and their integration within and between firms (Ward and Zhou, 2006). On one side, the philosophy of lean is "less is better": to improve the performance of a company, inventory, variability, material handling, options and choices must be reduced as much as possible. Conversely, the philosophy of IT is "more is better": IT tools allow to better manage more information, increased flexibility, functions and features.

According to Ward and Zhou, the two classes of instruments are complementary both in the concept and in the application: IT tools were considered a kind of higher-level planning system, while lean practices were related to shop-floor control and execution activities. Nevertheless, in order to define improvement strategies and assess their impact, the collection and analysis of information is mandatory: the deployment of methodologies for lean manufacturing cannot exclude the integration of IT tools. Hence, in the last years, IT instruments have been widely adapted, upgraded and expanded to deal with process monitoring and control activities.

The remainder of this Chapter is organized as follows: in Section 1.2 an introduction about lean manufacturing is provided. In Section 1.3 the infor-

#### 1.2 Lean manufacturing

mation tools mostly deployed in manufacturing are reviewed. Section 1.4 is devoted to the analysis of the requirements that factories have to satisfy in future to remain competitive. Finally, in Section 1.5 the aims of this work are presented.

# 1.2 Lean manufacturing

*Muda* is a Japanese word meaning "waste": it is referred to any human activity that needs dedicated resources, but does not create value. Taiichi Ohno, a Toyota executive, introduced the concept of *muda* in manufacturing, to label all the activities that require resources to take place, but do not add value to the process or to the product (Ohno, 1988). In particular, he defined seven classes of waste that typically affect a manufacturing process. Namely:

- 1. *Overproduction*: the production of unordered items, or manufacturing of goods before the customers demand. It is often considered as the worst of all the seven wastes: overproduced items have to be stored, leading to possible criticalities in handling materials or in movement through the plant. Further, storage is costly and items stored for long time risk to become obsolescent.
- 2. *Waiting*: the time that items spend idle while they are not processed or transported, in the form of raw material, work in progress, or finished product. This waste is mainly due to slow, not synchronized flow of material.
- 3. *Transport*: it is given by the unnecessary movement of materials and items (both semi-finished and finished) that does not add value to the product. Avoidable transport is a cost, due to items that spend time into a useless condition and to the resources performing the transport. Further, it is a source of risks, such as deterioration, damages or product loss.
- 4. *Extra processing*: it is given by the manufacturing operations performed to achieve a quality level not requested by the customer.
- 5. *Inventory*: it is the counterpart of waiting. Inventory is the quantity of components necessary to manufacture an item present in the process, in the form of raw material, work in progress, or finished product. Excess of materials in the process are money not producing income.
- 6. Motion: it is the counterpart of *Transport* concerning resources, rather than items. This waste is due to unnecessary movement of resources through the shop-floor without an increase of product value. These movements occur due to poor work planning or to a non optimal shop-floor layout. Beside the money cost, motion can result in ergonomic criticalities for personnel, leading to safety and health issues.
- 7. *Defects*: they are the most noticeable among all the wastes. Mismatching between the expected and the real quality of the produced parts require reworking operations leading, in turn, to additional cost and utilization of

resources. In case defects are found after product sale, image issues may arise and further costs are necessary to replace all the defective parts.

These wastes do not add value to the product, hence customers are not willing to pay for them. Manufacturers have to become less wasteful in order to be more profitable and improve their competitiveness. A systemic method to eliminate *muda* is *lean manufacturing* (Womack and Jones, 1996). It is an approach inspired by Japanese management methods, in particular by the Toyota Production System. The effectiveness of this approach led to interest among European and American companies. The first attempts in exporting Japanese production methods have been made in the 1970s. At that time, the experience was not successful; according to the prevalent opinion, this was due to cultural differences and to the unique social context of Japan. Nonetheless, in the early 1990s, deeper empirical and theoretical research has been performed and the basis for the first definition of lean manufacturing has been provided (Houy, 2005; Womack et al., 1990). This approach is structured in the following 5 principles.

- 1. *Specify value*. Value is created by the producer; nonetheless, it is defined by the ultimate customer, and refers to a specific product, a specific price, a specific time, a specific area. Hence, lean thinking must start through a dialogue with the target customers to define what is the value: providing the wrong product or service is a *muda*.
- 2. Identify the Value Stream. The value stream is the set of all the actions required to lead a product through the critical management tasks of a business: (i) problem-solving, i.e. the transformation of a concept into a physical product through detailed design; (ii) information management, i.e. transform orders into deliveries through detailed scheduling; (iii) physical transformation, i.e. the processes that transform raw materials into finished products. A careful analysis of the value stream is necessary to identify all the actions currently performed in the process that do not add value for the customer or lead to provide items that do not meet the customers request.
- 3. *Flow*. After the elimination of wasteful operations, the remaining valueadded steps have to be re-engineered to flow. The work flow has to be rethought and enabled to provide positive contribution to value creation; this step often involves a switch from departments and batches towards product teams and flows.
- 4. *Pull*. The change brought by *Flow* leads to a dramatic reduction in the time necessary to design and produce an item. This, in turn, enables inventory reduction and a shortened return of investment. Nonetheless, this is not sufficient to achieve lean manufacturing. A precise scheduling must be performed to produce items at the right time, i.e. when the customer pulls them. This approach is opposite to the push system, in which items are put into the market and may be unwanted, leading to *muda*.

- 1.2 Lean manufacturing
- 5. *Perfection*. The four steps presented above enable a company to perform a *kaikaku*, i.e. a radical improvement, due to the realignment of the value stream. Nonetheless, the four steps are not independent: they interact with each other and can be iterated in a virtuous circle: this is the *kaizen* approach, which is the philosophy of continual, incremental improvements aiming to achieve perfection.

Given these 5 principles, plenty of tools and techniques have been developed to improve parts of a process and achieve *lean manufacturing*. Among them, the most common are:

- 1. Single Minute Exchange of Dies (SMED): defined by Dillon and Shingo (1985), it consists in undertaking actions to minimize the setup times, which do not add value to the production. The quick changeover allows to profitably produce smaller size lots and, in turn, to reduce *waiting* and *inventory*.
- 2. Six Sigma: defined by Motorola in 1985, it consists in a strong reduction of process variability; the target of this methodology is to decrease the standard deviation of a given measure to a value lower than 1/12 of the specification limit. Six Sigma supports *defects* reduction: a process operating in these conditions produces, at most, 3.4 parts per million not matching the expected quality.
- 3. Kanban: it is a system of order cards grouped in two categories (Sugimori et al., 1977). A conveyance kanban is removed from a container when its content begins to be used, and is then attached to another container, upstream, containing another part. Similarly, a production kanban is used for containers with a part to be processed. In case no card is available, further parts cannot enter the line. Thus, the implementation of a kanban system supports the elimination of *Overproduction* and *Inventory*.
- 4. Value Stream Mapping (VSM): defined by Rother and Shook (2003), it is a tool to support the description of product path, from the supplier to the customer, through a careful representation of each step, in order to identify possible improvements and draw a map to describe how value should flow in future. It is particularly helpful in highlighting wastes in *transport* and *motion*.
- 5. Five-S (Osada, 1991) is a practice consisting in creating and maintaining well organized, clean, high effective and high quality workplaces. Namely, the five "S" are Japanese words that can be translated as Sort, Set in order, Shine, Standardize and Sustain. The result is a more effective organization, elimination of losses connected with failures and breaks, improvement of the quality and safety of work.

### 1.3 Information tools in manufacturing

Beside paradigms for waste reduction or elimination, the capacity of companies to competitively produce high quality products also depends on their ability to exploit IT solutions. Among the several information tools deployed within companies, two key classes of software are the Enterprise Resource Planning (ERP) and the Product Lifecycle Management (PLM). Each of them addresses different business needs for manufacturers and provides added value to the company. An introductory description is provided below. Each of these systems can be used independently; nonetheless, when integrated with each other, ERP and PLM result in a collaborative environment with increased impact on successful product development performance and the ability to maintain a competitive advantage (Omnify Software, 2007). A further tool is the Manufacturing Execution System (MES), which is the core of this research: it promotes information exchange between the ERP and the shop-floor.

# **1.3.1 Enterprise Resource Planning**

In the 1960s, companies were used to keep lots of just-in-case inventory, in order to be able to satisfy customers demand and remain competitive. Some customized software packages were developed to handle the inventory, based on traditional concepts. In the 1970s, firms could not afford any more large inventory quantities; hence, a new kind of software was introduced: the Material Requirements Planning (MRP). This package supported the inventory management to calculate the requirements of resources, based on the bill of materials, and the quantity of materials already available or booked.

Later, scheduling techniques were integrated in these systems, in order to deal with capacity planning and to better schedule both factory activities and suppliers deliveries. Furthermore, functionalities for the financial accounting and management systems were integrated into the MRP, leading to the development of Manufacturing Resources Planning (MRP II). This kind of systems allowed to have a more complete overview of the business system.

The MRP II has been then enriched, in the 1990s, to incorporate all resource planning of the entire enterprise, related, for example, to sales, marketing, manufacturing, logistics, accounting and staffing, product design, warehousing, human resources, and project management. These various departments are supported by delivering improved processes, e.g. through automated methods for order fulfillment or methods for information standardization.

#### 1.3 Information tools in manufacturing

The term Enterprise Resource Planning has been used to identify this complete IT tool (Omnify Software, 2007; Riezebos et al., 2009; Umble et al., 2003). The definition for ERP provided in the Eleventh Edition of the APICS dictionary is "a framework for organizing, defining and standardizing the business processes necessary to effectively plan and control and organization so that the organization can use its internal knowledge to seek competitive advantage". A synthesis of the modules and the corresponding functionalities that can be integrated into an ERP is shown in Figure 1.1 (Shehab et al., 2004).

The main advantage of an ERP is the integration of information. Before the introduction of ERP, a company would have a marketing information system, a production information system, and so on, each with its own hardware, software, and methods of processing data and information. Such unintegrated systems might work well within each individual functional area, but to be competitive, a company must share data among all the functional areas. When a company information systems are not integrated, costly inefficiencies can result, as well as the possibility of data errors, inconsistency or redundancy (Monk and Wagner, 2012). ERP systems have been widely adopted by large companies; nevertheless, the effort for their integration and maintenance into an organization is not trivial. This is the main reason for which ERP did not have a wide deployment in small and medium sized companies. The recent development of web 2.0 tools (i.e. platforms in which applications are no longer static, but are continuously modified by users, in a collaborative environment) and their integration in ERP software, made ERP suitable also for Small and Medium Enterprises (SMEs). Thus, beside big vendors such as SAP, Oracle, Microsoft and Sage, there also exist several open source ERP options: among them, Odoo, iDempiere, webERP, Openbravo.

### **1.3.2 Product Lifecycle Management**

CIMdata (2002) defines PLM as "a strategic business approach that applies a consistent set of business solutions in support of the collaborative creation, management, dissemination and use of product definition information across the extended enterprise from concept to end of life integrating people, processes, business systems, and information".

The origins of PLM systems rely in the Product Data Management (PDM) practices. PDM was a first step to satisfy the needs of information traceability by including more information about the product than just the geometric data. PLM extends PDM out of engineering and manufacturing into other areas like marketing, finance and after sale service and at the same time, addresses all the stakeholders of product throughout its lifecycle (Gunpinar and Han, 2008). PDM evolved during the 1990s to become PLM, providing

1 Introduction



**Fig. 1.1** A synthesis of the functionalities integrated into an ERP. Picture taken from (Shehab et al., 2004).

decision support at an enterprise level as well as continuing to handle traditional PDM functions (Abramovici, 2007; ATOS, 2010). The PLM concept integrates all the information produced throughout all the phases of a product life cycle to everyone within an organization, at every managerial and technical level, as well as with key suppliers and customers (Sudarsan et al., 2005).

Collaboration in PLM is made through the exchange of information related to the product; this exchange may regard different kinds of data such as design specifications, drawings, parameters, documentation, customers feedback, maintenance instructions, etc. (Garetti et al., 2007). From a production point of view, PLM is mainly about structuring product information in an orderly fashion, so that it is always available and can be accounted for on all levels in the manufacturing process and throughout the whole life cycle of each product. A PLM system works like a nervous system that communicates with all participants in a product manufacturing process, where the whole entirety leads to a developed product that can be verified in each step of the design and manufacturing process. A server holds the brain function and the communication is responsibility of the core services of the PLM system. PLM tracks and manages component data, bill of materials, documents, information for vendors and suppliers, and compliance data.

16

#### 1.3 Information tools in manufacturing

A key feature of PLM systems is the automated Change Management facility, that allows users to electronically propose changes to the information stored in the system. These proposals are then automatically routed to the appropriate resources for approval; finally, the PLM system automatically updates the information of the involved products with the suggested changes. If the PLM is integrated with an ERP, such information update is also communicated to the ERP. This electronic, integrated information update leads to a significant improvement in the performance of the engineering change: all the involved resources can simultaneously share information, reducing the cycle times for design and request of new parts; furthermore, the quality of transmitted data is increased, since issues due to handentering data are eliminated (Omnify Software, 2007).

PLM was first deployed in the automotive and aerospace industries: two sectors with complex, manufactured products (Abramovici, 2007). Despite its origins, PLM has now expanded to other industrial sectors (electronics, pharmaceutical, ships and buildings construction, etc.) and with a high diversity of products. Until recently, PLM solutions were designed exclusively for large, distributed manufacturing enterprises that had the extensive resources required to deploy and maintain them (CIMdata, 2011). However, SMEs are strongly motivated and they are attempting to integrate PLM into their business practice: they are a massive part of the world economy but a tiny part of the PLM marketplace. In the USA they contribute up to 30% of industrial output, while in countries such as Italy they form up to 95% of the industrial sector (PLM Interest Group, 2011).

Despite the promises made by some of the largest PLM software vendors, they have not yet delivered any PLM product to the market of small manufacturing companies. Nonetheless, there are open source solutions aimed for SMEs, such as Aras Innovator (CIMdata, 2011): companies can download, install, customize and use the software without any financial obligation to Aras.

### **1.3.3 Manufacturing Execution Systems**

The very first precursors of the MES are the data collection systems developed in the early 1980s: each area of a company (e.g. production planning, staff, quality monitoring, ...) had a dedicated acquisition system, which was independent from the others. The interdependencies of these areas began to arise with the emergence of the Computer Integrated Manufacturing (CIM) concept: production, personnel and quality were no longer seen as completely independent, and data crossovers were permitted from one task to another.

In the early and mid-1990s, these specialized data collection systems began to be upgraded and new features were added to integrate different fields (e.g. staff work time logging with PDA, PDA together with Machine Data Acquisition – MDA). Even with a small number of combination systems, a data collection (and sometimes a data evaluation) system able to take into account many functional areas of a manufacturing company could be defined. Nevertheless, the system components were already independent with each other and their synchronization required major work on interfacing. Over the course of time, combination systems performing several tasks arose from the independent data collection systems; their functionalities describe the functional scope of MES today:

- Production: from PDA, MDA, Distributed Numerical Control, control station;
- Resources: from staff work time logging, access control, short-term manpower planning;
- Quality assurance: from Computer Aided Quality Assurance, measured data acquisition.

However, these three task areas cannot be separated from each other: production accordingly needs suitable personnel to the quality it is producing. If mutually independent systems exchange their data or if data exchange is performed through the corporate level, too much time is lost and this can compromise the effectiveness of the reaction. Therefore the demand of more connected, or even horizontally integrated systems, arose (Kletti, 2007; Meyer et al., 2009).

Networked data acquisition and evaluation systems were developed to allow homogenized data exchange with the ERP system or with the automation level. Here data was exchanged with external systems via standardized interface mechanisms. Hence, data collection systems became closer to the MES concept. Systems of this kind support manufacturing operations by complying with the 6 R's rule which states: A product will not be created in the most economically efficient manner unless the Right resources are available in the Right quantity at the Right place at the Right time with the Right quality and with the Right costs throughout the entire business process.

If the networked data collection systems are integrated with elements of quality assurance, document management, document preparation and also performance analysis, the whole can already be regarded as a powerful MES system (Kletti, 2007). Several advantages can result from the MES deployment; among them, for example, improved product traceability, reduction of wastes and scraps, reduction of downtime, reduced production costs (Freedom Technologies, 2012).

Today, several commercial MES packages are available on the market, and the demand of such systems is increasing. A more detailed description about MES functionalities and the state of the art is provided in Chapter 2.

### 1.4 New requirements for future factories

Today manufacturing companies experience several challenges, such as the growing complexity of their processes and supply networks, cost pressures, increasing customer expectations for quality, lead time, and customization. In order to perform profitable production processes and improve competitiveness, different actions can be undertaken.

According to the European Commission (2014), the capacity of European companies to remain competitive while keeping the production phases in Europe strongly depends on their capacity in integrating state-of-the-art ICT solutions in their manufacturing plants. Among them, Smart Process Applications are a new class of software that combines the benefits of process applications and advanced analytics to help businesses and factories manage their resources, processes and systems more efficiently. They collect, process and analyze data by devices spread across production lines, logistic systems and plant sites, to provide meaningful information to decision makers.

Smart Process Applications are able to access and collect production data in real-time and to tag them to historical statistics, to obtain and analyze information collected directly on the shop-floor, to capture plant management information on production status, performance monitoring, and quality assurance. Furthermore, they are able to perform advanced computations to support the model creation and to test several scenarios and operating conditions in little time.

The market for Smart Process Applications was estimated to reach EUR 20.2 billion by 2015, and to grow at a CAGR<sup>1</sup> of roughly 18% towards 2018. Hence, one of the European research priorities focuses on factory design, data collection and management, operation and planning, from real-time to long-term optimization approaches. EFFRA (2013), the European Agency for the Factory of the Future, has identified the following key directions for the deployment of ICT tools in manufacturing:

- Solutions for factory floor and physical world inclusion. Real-world resources such as machines, robots, lines, items and operators are an integral part of the information structure of production processes. All of them need to be connected with each other and to back-end systems and, at the same time, to be self-aware of the surrounding environment.
- 2. Solutions for data storage and information mining. A huge amount of data from the shop-floor and the supply chain needs to be stored in a fault-tolerant way. Information embedded within these data has to be extracted and made available. New IT solutions should allow complex queries on distributed and heterogeneous data sources to be run (almost) in real-time to facilitate online decision-making across all the levels of the enterprise.

<sup>&</sup>lt;sup>1</sup> Compound Annual Growth Rate

- 3. Solutions for modeling and simulation tools. Complex environments need to be consistently described by semantic models in order to correlate information, describe the dynamics, and forecast their behavior. Knowledge from different sources must be made available and fully exploited by dedicated modeling and simulation tools.
- 4. Collaborative and decentralized application architectures and development tools. In extended enterprises and globalized markets, applications (e.g. life cycle management, supply chain management, monitoring and control, and customer relationship management) must no longer operate in closed monolithic structures. Stakeholders and customers must be able to cooperate on a common application platform implemented with the cloud approach for rapid development and deployment.

### 1.5 Aim of this work

In the previous Sections, two approaches to improve the performance of a process have been introduced: the deployment of lean manufacturing techniques and the integration of information tools. For several years, they have been considered mutually exclusive; nonetheless, recently the importance of the cooperation between the two techniques has been understood: IT tools can collect and analyze data useful to extract information significant for undertaking continuous improvement practices.

As introduced in Section 1.3.3 (and further detailed in Chapter 2), MES is in charge of collecting data, perform analyses and dispatch the resulting information. Nonetheless, the MES currently available on the market mainly focus on the top-down data flow (i.e. from the business level to the shop floor) through detailed jobs scheduling and dispatching. The opposite flow is mainly dealt through performance indicators. However, heterogeneous, rich data can be collected at the shop floor.

Further, current MES are static and are not able to adapt adequately to the evolvable production environments. The high dynamicity of future manufacturing systems requires a constant optimization of quality and resources usage, and the amount of knowledge extracted from the shop-floor should be fully exploited by MES (European Commission Business Innovation Observatory, 2014).

Hence, the aim of this research work is the development of mathematical techniques able to analyze the data collected by sensors systems and provide hints to reduce the sources of waste, thus improving the performance of the process and the quality of the product. These techniques are designed to be integrated into a MES and to cooperate with other tools, such as design tools. The findings of this work can support the development of a new MES generation capable to deal with highly dynamic environments and support a more sustainable manufacturing.

20

### 1.5 Aim of this work

The present work aims to address the following research questions:

- 1. In which manufacturing fields MES can be able to support lean practices?
- 2. Is it possible to identify a common approach for different areas or a specific path must be developed for each application?
- 3. Is the path for MES implementation dependent on the specific mathematical techniques to be used?
- 4. Is it possible to integrate MES with further information tools? Which benefits could result from such integration?

The first step of this work is the definition of a rigorous methodology. It consists in three steps, as shown in Chapter 3: (i) identify the sources of waste; (ii) deeply understand the process; (iii) develop the solution to deal with the target issue.

The methodology has been applied to different technologies: machining through traditional, subtractive manufacturing (Chapter 4) and additive manufacturing processes (Chapter 5). In both the cases, techniques to solve geometrical issues on the surface of the part have been developed. Further, two applications to ancillary services (with respect to the manufacturing process) have been studied: the tools presented in Chapter 6 allow to manage and evaluate the performance of automated warehouses, as well as to prevent possible issues. Conversely, the tools discussed in Chapter 7 aim at enhancing the transport of items along the shop-floor through automated vehicles.

The tools introduced in the following Chapters can be used with two deployment scale. The first one is the identification of actions enabling process improvement at short, medium and long term scales. The second one is the extraction of experience-driven knowledge that has to be formalized and integrated into design tools: in this way, it can be made available for further developments and shared by all the potential users.

# Chapter 2 Manufacturing Execution Systems

**Abstract** This Chapter is devoted to a deep review of Manufacturing Execution Systems, a class of software born in the early 1990s to support communication and data exchange between the business level of a company and the shop-floor. An exhaustive definition of such systems and their functionalities has been provided into the standard ISA 95 (2000). Here, the tasks in charge to a MES are reviewed, and the state of the art in the field is depicted from three points of view: (i) MES market is analyzed from the business perspective; (ii) the trend in intellectual property rights is shown to identify the top innovators; (iii) the most important research themes are synthesized to present the key directions identified by the scientific community.

# 2.1 Definition of MES

The first organization which defined the tasks to be dealt by a MES was the Manufacturing Enterprise Solutions Association (MESA), a US "global community of manufacturers, producers, industry leaders, and solution providers who are focused on driving business results from manufacturing information". MESA provided the following list of 11 functionalities (MESA International, 1997); combined with each other, they can form a MES solution.

- 1. **Resource allocation and status**. Manage and monitor resources, including staff, machines, tools and make available the documents necessary to start the working operations. Further, set up the equipment, and reserve resources and dispatch orders in order to meet the target objectives.
- 2. **Operations/Detail Scheduling**. Identify the optimal sequence planning based on priorities and resources availability, in order to minimize setups and downtime.
- 3. **Dispatching Production Units**. Manage the flow of production units (jobs, batches, lots, ...), and adjust it in real-time as events (e.g. reworking operations) occur on the shop-floor.
- 4. Document control. Manage and control the information significant for the production process (work instructions, drawings, specifications, environmental compliance requirements, safety instructions, etc.) as well as the "as planned" and the "as is" information. Historical data are saved; the information must be accessible to the staff at the right time and right place.
- 5. **Data collection/acquisition**. Data related to the production can be collected both automatically or manually, and used to track deviations.
- 6. Labor management. Provide the updated status of the personnel, store the staff working hours, the criteria to manage absences, holidays, etc, as well as the ability to perform tasks. This package can be used to evaluate the cost of activities, and may interact with the ERP to optimize resources allocation.
- Quality management. Measure production data and analyze them in real-time, aiming at ensuring product quality and identify in advance issues and criticalities. Actions to correct the issue can be included, as well as tools for process control (such as Statistical Process Control - SPC - or Statistical Quality Control - SQC) and for the management of inspections and offline analyses.
- Process management. Monitor the production process; alarm management functions can be included and automatic corrections or decision-support tools can be integrated to correct and improve process activities.
- Maintenance management. Track the use of operating material to plan periodic and preventive maintenance tasks, ensuring their availability according to the scheduled activities. The system also stores the chronology

### 2.1 Definition of MES

of past interventions to support problem diagnosis and the execution of maintenance actions.

- 10. **Product tracking and genealogy**. Record all the production data across the entire manufacturing chain, to ensure that the position of each item can be identified in real-time as well as its manufacturing history (e.g. components suppliers, lot and serial number, operators working on it, alarms, ...).
- 11. **Performance analysis**. Produce user-friendly, complete reports containing process and product information (e.g. resources availability and utilization, cycle times, noncompliances, ...) and a comparison with the past history and the expected performance, to support the assessment of production efficiency and the detection of issues.

Later, in the 2000s, the standard ISA 95 has been issued by the International Society of Automation. In this standard, a functional hierarchy model consisting in five levels is defined:

- Level 4: Business planning and logistics.
- Level 3: Manufacturing operations and control.
- Levels 2, 1, 0: Batch, Continuous, Discrete control.

Level 0 indicates the manufacturing process; Level 1 indicates manual sensing, sensors and actuators used to monitor the process; Level 2 indicates the control activities that keep the process stable or under control. These tasks are not addressed in the ISA 95 standard: this document is mainly concerned with the activities for Levels 3 and 4, and with their interface for data exchange. Level 4 activities include tasks for business management, which are usually performed by an ERP. Conversely, Level 3 is related to production management; the functionalities of this level correspond to the list of 11 tasks shown above, defined by MESA International (1997). The standard ISA 95 has been adopted and extended by the International Electrotechnical Commission, named "IEC62264: Enterprise-control system integration". Currently, this standard consists in 5 parts:

- **Part 1: Models and terminology**. It describes the Level 3 activities, and the interfaces within Level 3 and between Levels 3 and 4. The last edition has been issued in 2013.
- Part 2: Objects and attributes for enterprise-control system integration. It specifies generic interface content exchanged between manufacturing control functions and other enterprise functions. The last edition has been issued in 2013.
- Part 3: Activity models of manufacturing operations management. It defines activity models of manufacturing operations management that enable enterprise system to control system integration. The last edition has been issued in 2007.
- Part 4: Objects models attributes for manufacturing operations management integration. It defines object models and attributes exchanged

between the manufacturing operations management taking place in Level 3 and defined in the previous Part. The last edition has been issued in 2015.

• **Part 5: Business to manufacturing transactions**. It defines business to manufacturing transactions and manufacturing to business transactions. The last edition has been issued in 2011.

In Figure 2.1 a schematic of the hierarchy model levels, the associated company levels, and the corresponding supporting IT tool is provided. Further, a graphical indication about the time-scales and the detail of the transmitted information is given. The time scale for the planning provided by the ERP is in the order of weeks-months; the more detailed schedule elaborated by the MES involves events in the order of hours-days; the phenomena occurring at the shop-floor have lower time-scales, in the order of minutes or hours. Conversely, on the shop-floor a huge quantity of data can be acquired. Such data must be analyzed and transformed into a smaller amount of information to be transmitted to the business level, in order to have a complete and exhaustive picture and take proper tactical decisions.



Fig. 2.1 The functional levels defined in the standard ISA 95 and the supporting IT tools.

### 2.2 State of the art

In this Section, the state of the art concerning MES is depicted. The content is divided into three subsections, to analyze different perspectives: (i) the business point of view, by a synthetic market analysis and the identification of megatrends supporting the spread of MES; (ii) the industrial point of view, through a patent landscaping, to identify which are the trends in intellectual property rights and who are the innovation leaders; (iii) the academic point of view, through an analysis of the most recent scientific papers.

26

### 2.2.1 MES market

According to the last report of Markets and Markets (2015), the MES turnover is expected to reach USD 12.6 Billion by 2020 at a CAGR of 10.85% between 2015 and 2020. For a long period, Europe has been the main market for MES; nevertheless, in recent times the Asia Pacific region has superseded Europe as the biggest regional market. North and Latin America are, respectively, the 3rd and the 4th market (DKSH, 2014).

The major players in the MES market include ABB Ltd. (Switzerland), Andea Solutions (Poland), Dassault Systemes SA (France), Emerson Electric Co. (US), General Electric Co. (US), Honeywell International Inc. (US), Rockwell Automation, Inc. (US), SAP AG (Germany), Schneider Electric SE (France), Siemens AG (Germany), and Werum IT Solutions GmbH (Germany).

The market for MES is growing at a high pace, driven by the increasing demand of the following technologies: (i) The internet of things; (ii) Big Data; (iii) Cloud computing; (iv) Analytics. The combination of these technologies within a production management system makes feasible a representation of the production environment that goes beyond the boundaries of the plant, and is able to pervade the entire supply chain from the raw material to the final customer. Furthermore, the availability of real-time information concerning both the basic components and the finished products allows production processes to transform themselves constantly, adapting to the market conditions to optimize production times, reduce waste, maximize inventory turns, improve efficiency and, ultimately, ensure the satisfaction of the customer. The information generated by MES has a high value for companies: the ability to include information collected directly from the finished product in real time adds a whole new dimension to the analysis.

Nevertheless, the implementation of a MES requires a reorganization of production processes – due to technological investments – as well as a cultural change: each link in the production chain cannot independently determine its own strategy, but has to align its strategy with those of the other elements and must behave as one component in the full transformation of the entire system (De Bernardini, 2015).

# 2.2.2 MES intellectual property rights

In order to depict the inventive landscape in the field of MES, a patent search has been performed. A systematic analysis using the Orbit database has been made: the string "Manufacturing Execution System" has been searched into titles, abstracts and claims of patents deposited starting from 1990. The research, updated in January 2016, resulted in 660 patents.

### 2 Manufacturing Execution Systems

The first interesting result is the time trend for the first priority date<sup>1</sup>, plotted in Figure 2.2: the increasing pace of deposited patents shows that the interest in MES is increasing, as well as the necessity to provide innovative features. It must be highlighted that, due to the 18 months nondisclosure period, the data corresponding to 2014 and 2015 are not complete. The list of the top ten assignees is shown in Table 2.1: Siemens and Taiwan Semiconductor Manufacturing are the companies that most invested in intellectual property rights. Since the landscaping covers a wide time span, a narrower research, focused on the period 2009-2014, has also been performed. The last two columns in Table 2.1 show who have been the top innovators in MES in the last years. The list is made of companies involved in software and factory automation (Siemens, Rockwell Automation, IBM), as well as companies focused on specific manufacturing fields (materials and semiconductors). Nevertheless, the list of top vendors provided in the previous Section does not totally fit with this one: there exist some MES suppliers who do not own patents, as shown in Table 2.2. US and China are the countries which have invested more in the field of MES; the leader country in Europe is Germany.



Fig. 2.2 Time trend of the deposited patents in the field of MES.

28

<sup>&</sup>lt;sup>1</sup> The first application date for a given invention

### 2.2 State of the art

Table 2.1 List of the top patents assignees in the field of MES.

D 1.	Assistant	Nr. of patents		Dette
капк	Assignee	1990-2014	2009-2014	Katio
1	Siemens	78	45	58%
2	Taiwan Semiconductor Manufacturing	73	8	11%
3	Rockwell Automation	28	12	43%
4	Shangai Huali Microelectronics	27	27	100%
5	Applied Materials	22	0	0%
6	IBM	20	4	20%
7	Advanced Micro Devices	16	0	0%
8	Semiconductor Manufacturing International	15	10	67%
9	Powerchip Semiconductor	15	0	0%
10	Global Foundries	13	6	46%

Table 2.2 Patents portfolio of the top MES vendors.

Vendor	Nr. of patents		
ABB	9		
Andea Solutions	0		
Dassault Systemes	0		
Emerson Electric Co.	0		
GE	8		
Honeywell	2		
Rockwell Automation	28		
SAP	3		
Schneider Electric	0		
Siemens	78		
Werum IT solutions	0		

### 2.2.3 MES scientific literature

The third step performed to depict the state-of-the-art in the field of MES is an academic literature review. The content of this Section is grouped through keywords, to point out the directions identified by the scientific community.

**Traceability**. For a long time, bar codes have been used for product traceability; nevertheless, some issues arose as the adoption of this technology increased. Among them: (i) the low effectiveness, as only one bar code can be acquired through a scanning action; (ii) the low reliability, since bar codes can be easily damaged; (iii) the abundance of manual operations, for example to scan the codes. As the RFID technology became mature, it replaced the deployment of bar codes; several research works published in the last years present successful integration of RFID devices (tags, readers, and data transfer tools) along the manufacturing process, to support real-time data flow and analysis. Chen et al. (2009) introduced a system to integrate production management, labor management and warehouse operations (inventory tracking and visibility) for a mix-model assembly line. RFID devices are deployed to turn warehouses, conveyors, workstations, critical components and operators into traceable smart entities, and to facilitate intelligent operational functionalities of the system. Wang et al. (2012) presented a case study (a mold and die company) for a RFID-enabled MES to realize real-time, accurate control of a one-of-a-kind production processes Dai et al. (2012) introduced RFID devices to collect real-time data on the shop floors in a costeffective way, to transform it into meaningful information to enable both convenient operations for the operators (e.g. work-in-progress visibility and traceability) and efficient decision making for shop-floor supervisors (e.g. work-shop scheduling). Further, the authors aim to facilitate and rationalize shop-floor management. The case study was provided by an engine valve manufacturer. Fu and Jiang (2012) used RFID networks to collect quality data including static and dynamic parameters, based on the result of realtime status tracking for manufacturing. Zhong et al. (2013) deployed RFID tags to improve real-time data collection, planning and scheduling as well as to efficiently trace items.

**Support to lean manufacturing**. As stated in Section 1.2, for a long time the deployment of IT tools and the lean practices have been considered mutually exclusive. However, recently, the importance of deploying MES to support continuous improvement techniques has been shown. Cottyn et al. (2011a; 2011b) presented a method to align MES with lean objectives: the information provided by the MES and its standardized way of working can trigger and validate the lean decision-making process. The importance of MES-lean integration is explored through the case studies of a furniture firm and a food and beverage company.

MES design. The decision-making process of a MES can be hierarchical or distributed. In the former structure, the control is centralized; this kind of structure leads to efficient and robust results for stable and predictable manufacturing environments. However, a centralized MES is not efficient in dealing with unplanned disruptive events leading to reschedule tasks. Hence, the adoption of a distributed approach can be more convenient: in this case, shop-floor control is not carried out by a central unit, but is the consequence of the actions and interactions of local controllers in the system. An approach for a distributed architecture is the deployment of holonic MES: holons are autonomous and cooperative blocks of a manufacturing system, consisting in an information processing part and a physical processing part. A solution is proposed in (Simao et al., 2006). Valckenaers (2007) and Verstraete (2008) presented an holonic MES that uses a given schedule as a guideline to select among task execution alternatives but is also able to find solutions when the schedule is infeasible or unplanned events occur. This design, following the PROSA architecture (Van Brussel et al., 1998), is made of three agent-types: (i) resources (managing the resources on the shop-floor); (ii) products (that know how a product can be manufactured by

### 2.2 State of the art

the resources); (iii) orders (which consult the product agent to identify the necessary resources and looks for their availability). This approach is also used by Blanc (2008) to design a MES for the manufacturing of laminated bullet-proof security glasses, ready-to-assemble on vehicles. In (Rolón and Martínez, 2012), the autonomic MES is based on two classes of intelligent, open, self-managing units for simultaneous scheduling and control of an object: orders and resources, which interact according to a monitor-analyze-plan-execution loop.

**Collaborative environments**. The term *collaborative* can be used on two different scales. The first one is related to collaboration **within** the company. In 2004, the term C-MES (Collaborative MES) was coined, and a further role for MES was provided: it was considered not only as the communication layer between the business level and the shop-floor, but it was also seen as a real information hub. In this view, MES is a hub able to integrate information and dispatch it throughout the company; other functionalities (e.g. business, technical or logistical) are just users entering the MES platform to access data (MESA International, 2004). A schematic of C-MES approach is provided in Figure 2.3. On the other side, to approach agile manufacturing, collaboration is also required **among** the suppliers and the partners of a company. Thus, the information tools should reach high levels of integration. One way to approach this result is the use of cloud computing, to improve the effective-ness of information exchange among companies. One approach has been proposed by Helo et al. (2014).

Modeling and simulation. In order to make production planning dynamically adaptive to changing requirements, Rao et al. (2008) integrated simulation tools into the MES. To implement a real-time process control based on real data, an on-line simulation tool was deployed for decision making and evaluation. MES provides real-time data as input to the simulator which, in turn, provides the MES with the optimal production plan. Hence, simulation is no longer used as a long-term optimization tool, but is deployed to deal, over short-term time scales, with variability and unplanned events.



Fig. 2.3 The structure for C-MES model proposed by (MESA International, 2004).

# Chapter 3 Research methodology

**Abstract** In this Chapter, a three-steps methodology is defined: in order to accomplish the research aims of this work, the definition of a work methodology is mandatory. The first step is the identification of the process wastes, according to the classification of *muda*, that affect the process at stake and need to be eliminated. The second one is an exhaustive description of the process: this allows to identify the sources of the target wastes that must be kept in control. Finally, the strategy to acquire significant data and transform them into information useful to eliminate wastes is identified. Beside the description of the three steps, a schematic tool is also presented, to support the application of the methodology to the case studies dealt in the following Chapters.

# 3.1 Introduction

In the previous Chapter, a thorough introduction about MES has been provided: synthetically, these systems are in charge of taking data as input, analyze them through appropriate techniques and dispatch the results. This approach holds both for the top-down data flow (i.e. orders and targets provided by the business level to be transformed into manufacturing planning) and for the bottom-up data flow (the feedback information from the shopfloor).

In this work, new mathematical techniques to be integrated into a MES are presented, with the ultimate aim of achieving lean manufacturing. However, in order to best address the development work, the definition of a rigorous research methodology is necessary. The methodology deployed in this work consists of three steps:

- 1. Identification of waste classes to be faced: the kinds of waste that (can) affect the performance of the process are identified and classified according to the 7 *muda* defined in Section 1.2.
- 2. Description of the process: in order to identify the sources of waste and identify possible interventions to improve the performance, a wellstructure, exhaustive description of the process is necessary.
- 3. Data-analysis: it is the core of this research. In order to develop mathematical techniques, the source of input data, the target output information and the technique to transform data into information must be defined.

The first step is performed by the manufacturer; it can be made both onto an already existing process (thus, identifying wastes already affecting the process), or during the design phase, to evaluate possible process wastes and reduce them before the physical realization of the system. The two subsequent steps are further detailed in the following Sections. A graphical representation to synthesize the three steps is defined in Figure 3.1: this schematic is also used in the following Chapters to provide an exhaustive, easy to read description of each case study.

### 3.2 General description of a manufacturing process

In Figure 2.1 a layered model to describe a company has been provided. This structure can be adopted to describe a manufacturing process from the firm point of view, i.e. focusing on the business necessities, on the operations requirements or the shop-floor control. Each of these levels needs different information, which must be provided with different details depth and at different time-scales.

3.2 General description of a manufacturing process

An alternative approach consists in describing the process from the process perspective itself. The family of standards ISO 15531 (2004) has been published to define, in detail, the data structures to be exchanged within and between firms, to correctly perform a manufacturing process. In these standards, a distinction between resources and flows is given: the former class contains *any device*, *tool and means*, *except raw material and final product components*, *at the disposal of the enterprise to produce goods or services*; the latter consists in the *motion of a set of physical or informational objects in space and time*.

Nevertheless, in this work a structure detailed like the one provided in the standard is not necessary. The description used in this work is shown in this Section. It is a layered representation in which two classes of items cooperate to perform the manufacturing process: resources and components. Both the categories are made of physical objects and information, and can be provided as input for the process or obtained as output.

### Components

The term *Components* denotes all the items transformed by the manufacturing process to obtain a (semi-)finished object: they can be grouped into *Input components* and *Output components*, as shown in the following. Each of these categories consists of both information and physical objects, and can be split into further categories.

### Input components

This category includes all the components given as input to the manufacturing process. To better address the categorization, this group can be split into the following three subgroups:

- 1. **Suppliers**. The suppliers of a manufacturing process provide raw materials or semi-finished parts to be further processed. They can be external partners as well as upstream manufacturing processes within the same company. Together with the physical objects, a set of information must be provided. For example, the properties of the supplied parts (e.g. the composition of material, chemical, mechanical, electrical properties, ...) as well as their manufacturing history (when each part has been produced, where, which were the suppliers, ...). Furthermore, the constraints of the supplier must be known (e.g. supply capacity, cost, reactivity to new orders, ...). This information must be managed and dispatched by the ERP.
- 2. **Planning**. This input class only consists in information necessary to plan the production and, thus, to control the shop-floor. In a push production system, the interarrival time and variability for the input compo-

nents must be evaluated, as well as the size of the batch to be produced. Conversely, in a pull system, information about customers demand (average desired quantity per time unit and variability) must be provided. The MES is responsible for this information, since it is in charge of optimizing the production planning and flow.

3. **Design**. The third class of input is related to the instructions necessary to produce the parts: which are the materials necessary, which are the machines, the part-programs, the parameters required to produce the desired part. Furthermore, the positioning of the material into the machine, the tolerances to be satisfied and the final dimensions of the item must be known. This information is stored into the PLM.

### **Output components**

This category includes all the components resulting as output from the production process. Two subgroups can be identified:

- 1. **Performance**. The process provides, of course, the (semi-)finished products, along with a set of performance indicators to characterize the line: among them, the cycle time, the work in process, the throughput, the queues, the average utilization of the machines, their availability, the incidence of failures. These data can be stored and further analyzed (e.g. through time-series analyses) to synthesize the behavior of the line over the time-scales of interest.
- 2. Quality. Information about product quality is getting to be mandatory for manufacturers. It may result from a simple "pass or non-pass" test to quickly verify whether the tolerances are satisfied or not, or from a more complex monitoring system based on the deployment of sensors. Furthermore, quality information can be obtained both from on-line test and off-line verifications, through inspections performed in dedicated areas after the production process (e.g. metrological measurements performed in a controlled environment). Information concerning the incidence of reworking and scraps can be necessary.

Both the two classes of information are managed by the MES: this system must collect performance and quality information, analyze and merge it through proper mathematical techniques, and provide an exhaustive and synthetic report to the business level, in order to verify whether the process is working in a correct and profitable way, or if an intervention for performance improvement must be taken.
## Resources

The definition for *Resources* is adopted from the standard ISO 15531. Even in this case, a distinction between input and output resources can be performed; a further distinction can be made to distinguish *Reusable* and *Disposable resources*.

## Input resources

This category comprises all the resources necessary to run the manufacturing process. The following two groups can be defined:

- 1. **Reusable**. This group includes all the resources that can be re-used in the manufacturing process after the production of a part. Among such resources, there are the operators, the transportation means, and the machine; eventually, a setup operation may be necessary to restore the initial state of the resource (e.g. a break for the operator; battery-charge for a forklift; tool change for a machine, ...). The information concerning the state of the resources at the beginning of the manufacturing process must be stored.
- 2. Disposable. This group collects the resources which are used for the purposes of the production process and cannot be reused or restored: for example, the energy and the fluids (compressed air, lubro-refrigerants) used by the machine, or the tool, which must be changed after a finite number of manufacturing operations.

#### Output resources

This group can be divided into the following categories too:

- 1. **Reusable**. The physical output quantities are the same that were provided in input (the number of operators, the machine, the transportation means). Nevertheless, the manufacturing operation changed their state: hence, information about their state after the process must be collected.
- 2. **Disposable**. Given the nature of these components, nothing can be collected at the end of the process, except scraps. Information about the consumption of the process must be collected.

The data acquired before and after the process must be compared to evaluate its real impact and cost. The tool in charge of this task is the MES: it collects information on the shop-floor, analyzes it and provides a report to the business level, in order to check whether the process is operating in an economically sustainable condition or not.

## 3.3 Methodology for data analysis

One of the aims of this thesis is the development of smart mathematical techniques to be integrated into a MES, able to transform data into valuable information. In literature, several definitions for data and information are provided. Authors agree in stating that data are discrete observations which are unorganized and unprocessed, and hence without any specific meaning (Bocij et al., 2008; Groff and Jones, 2011; Valacich and Schneider, 2011). Conversely, information is given by formatted data that can be defined as a representation of reality (Valacich and Schneider, 2011); according to Bocij et al. (2008), information is: (i) data that have been processed so that they are meaningful; (ii) data that have been processed for a purpose; (iii) data that have been interpreted and understood by the recipient. Bocij et al. (2008) and Curtis and Cobham (2008) provide a list of the processes that allow to convert data into information. It consists in: (i) classification; (ii) rearranging/sorting; (iii) aggregating; (iv) performing calculations; (v) selection.

The most popular paradigm for the transformation of data into information is provided by the DIKW (Data – Information – Knowledge – Wisdom) hierarchy (Ackoff, 1989): it is often represented as a pyramid with the data at its base and the wisdom at the apex; each level of this hierarchy is the essential precursor for the above one. However, while the distinction between data and information is clear, there is less agreement about the processes that convert the former into the latter. Since information is obtained by organizing and structuring data, any scheme that has meaning and relevance for an individual, community or task, provides meaning to the data (Rowley, 2007). Hence, the rigorous definition of a technique to analyze and organize collected data plays a strategical role.

The methodology for data analysis introduced in this Chapter consists of the five key steps described in the following. It is an adaptation of the strategy defined by Abellan-Nebot and Romero Subirón (2010): their working area is in the field of intelligent monitoring systems; here, the methodology is extended and generalized in order to deal with monitoring and control systems integrated into manufacturing machines as well as with the analysis of any kind of data collected on the shop floor.

## Data source

The first step of the methodology is the identification of the data necessary to perform the analysis and the definition of the data collection devices.

On the shop-floor several kind of devices can be deployed to collect data. First, the PLC of the machine involved in the process can provide helpful data concerning, for example, axes position and errors, axes and spindle movement, the deployed tool and the content of the warehouse, the applied

#### 3.3 Methodology for data analysis

power and torque, and some key performance indicator (e.g. cycle times, throughput, the incidence of failures). Furthermore, different kind of sensors can be integrated into the machine to collect data related to the quality process and the state of the tool. Measurements performed by sensors can be classified into direct and indirect: the former are more accurate but usually expensive and difficult to implement in a machining environment; the latter are more economical and consist in inferring variables to have knowledge about the state of the process. In machining processes, the most deployed sensors are dynamometers, accelerometers, thermometers, acoustic emission and current sensors (Abellan-Nebot and Romero Subirón, 2010; Teti et al., 2010). Sensors can be used both online – while the process is occurring – or offline, for example to evaluate the quality of a finished part (e.g. geometrical dimensions, mechanical strength, electrical properties, ...); further, sensors collecting different kind of data can be used and their information can be integrated to have a more exhaustive picture.

Nevertheless, the shop-floor is not the only data source for MES. These systems also receive information from the design department, such as the bill of process, the bill of materials, the properties of the machines, the layout of the plant. The management level of the company provides the MES with data concerning the target to be reached (e.g. the bill of orders), or target levels to be respected (e.g. performance indicators). Such data must be considered to best plan and manage the production flow taking into account all the constraints. Furthermore, a part of the data used by the MES is generated by the MES itself, as the result of previous analyses: for example, time trends for productivity and performance indices, or the identification of statistical distributions to describe phenomena occurring at the shop-floor.

# Data processing and Feature generation

The second step of the methodology consists in choosing the mathematical technique to analyze the collected data. The aim of the data processing technique is to transform data, regardless of the source, into such information, through the generation of a finite set of features. Thus, the choice and the implementation of an appropriate processing technique is mandatory for a correct data interpretation and for a successful decision-making strategy. According to the specific case study, an approach already existing in literature of a technique developed ex-novo can be used.

Mainly, two classes of data processing techniques can be used. The first one consists in mathematical models, based on deterministic or statistic approaches. This technique is convenient when the analyzed system is not too complex and its behavior is fully known. In particular, the statistical approach is effective in dealing with a huge amount of data and is widely used, for example, with data acquired by a sensors set. Different indicators and graphical representations can be extracted to have information about the central tendency, the dispersion and the shape of the data. Deeper analyses can be performed through regression tools or interpolation techniques, or through analysis of variance, to identify which are the factors and their interactions which most impact the process. A data-fusion approach (Boström et al., 2007) can be used when heterogeneous data sources are used, in order to reciprocally integrate the collected data and investigate the relationships among variables, aiming at obtaining a more exhaustive and reliable description of the process.

The second class of data processing techniques consists of simulation tools: they are preferable when the analytical description of the system is too complex. Data provided in input to the simulation can provide from several sources: theoretical (or expected) data can be used to evaluate the behavior of the system in standard situations; real data, collected at the shop-floor are helpful to be aware of the reaction of the system in the current situation.

## Feature extraction and Decision making

The role of the data processing technique is to synthesize the collected data into a smaller set of information features; nevertheless, some of them may be not significant or reliable to take decisions and, thus, should be discarded. Furthermore, new significant features can be extracted by combining some parameters: overall indices can be obtained by averaging features, by generating response surfaces or by comparing the expected state with the real condition of a process or a product.

Finally, a strategy for decision making must be defined, based on the results of the feature extraction. The decision can be automatically taken by an algorithm able to choose the values of a set of parameters in order to optimize a given metric. Alternatively, the algorithm may provide hints to an operator and leave him free to act on the process. Furthermore, the decision making algorithm should also provide an estimation of the state of the process after such intervention, to evaluate the impact on the performance of the process.

# 3.4 A schematic tool for the methodology

The three steps described in the previous Sections have been synthesized into a unique schematic, shown in Figure 3.1, with a twofold aim. First, it can be used to point out ideas when dealing with a case study. The schematic must be filled in a clockwise direction: first, the sources of waste must be identified and highlighted; second, the process must be thoroughly 3.4 A schematic tool for the methodology

described; third, the technique for data analysis must be designed. The second purpose of this schematic is to provide the reader with a synthetic, exhaustive overview of the case study. In the Figure, black color represents information; green color is used for physical quantities.

		Process	de	scripti	DN
C	Suppliers Raw ma Materia Supply Planning Batch s Interarr	iterial Il properties constaints ize ival times & variability	I N P U		Reusable Operators Transportation means Machines Current state Disposable Enerau
M P O	Design Materia Parame Toleran	II, Machines, Part programs ters, Part positioning, ces, Part dimensions	T		Fluids E Tools S Current state
N E N T S	Performan Finisher Cycle ti Queves Machine Failures	ce I product me, work in process, throughput es utilization incidence	0 U T		ReusableROperatorsCTransportation meansCMachinesECurrent stateS
	Quality Test & Non-co Reworki	Inspections nFormities ng and Scraps	P U T		Disposable Waste Resource consumption
		Wastes	Τ		Data—analysis
Overp	roduction	Production of unordered items		Data source	The devices deployed to collect data
Waitir Transı	ng  port	The time that items spend idle Unnecessaru movement of items		Data processing	The technique to transform raw data into information
Extra	processing	Achievement of unnecessary quality		Feature generation	The information provided as output by the data processing
Invent	tory	The quantity of idle parts		Feature extraction	The features significant for the reduction of target waste
Defec	" ts	Poor item quality	····	Decision making	Strategy for process improvement

Fig. 3.1 The schematic of the methodology used to develop this work. Black color represents information; green color is used for physical quantities.

# Chapter 4 MES for monitoring and control of a finishing operation

Abstract Gears manufactured for aeronautical applications must meet very high quality, due to the tight tolerances required and the possible consequences of a failure. Further, the grinding process for such gears is a costly operation to be executed in the best conditions; the positioning of the gears into the machine is performed using two reference surfaces previously finished. Hence, great accuracy must be ensured in the finishing operation. A manual alignment of the gear in the finishing machine is not satisfactory any more; hence, an automation system has been developed. It is based on a monitoring and control system equipped with sensors for surface measurement; a mathematical technique has been studied to identify the current gear position in the finishing machine and provide the orientation parameters that minimize the residual errors. Beside online control, the integration of such monitoring and control system with a MES supports process stability as well as the evaluation of longer term analytics. Further, the information generated by the MES can be used as a feedback to redesign or revise manufacturing operations, in order to enhance the quality of the product and the performance of the production process. This experience-driven knowledge must be integrated in the PLM, to be available for future production, and shared in different places or among cooperating companies.

## 4.1 Introduction

The work shown in this Chapter deals with the development of a real-time monitoring and control system to be deployed in the manufacturing of aeronautical spur gears: such workpieces exhibit a high unitary production cost, and the accuracy of the final product must be high. During the manufacturing process, heat-treatment is performed, leading to high geometrical distortions. Hence, the general attitude of the gear must be recovered to guarantee the required form accuracy. The recovery process usually starts from a well-defined workpiece alignment with the grinding machine axis. For this purpose, a prototype machine has been developed to generate the gear axis by finishing the countersinks of the gear. According to the experience of the manufacturer, there exists a specific point located on the gear axis which is not affected by heat treatment distortion, and it can be deployed as a pivot for workpiece alignment, in agreement with the currently deployed manual centering operation.

The aim of this work is the development of a mathematical technique able to lead to the best workpiece alignment within the prototype machine. The technique described in this Chapter is based on measuring a set of surface points of the spur gear: to perform the measurement, the part is mounted on the machine rotary table and a suitable set of sensors acquires data by performing one workpiece revolution. Such data are used to calculate the values for two angular rotations of the gear axis about the pivot, leading to the best alignment of the workpiece with the machine axis. The rotary table and a set of actuators allow to perform the alignment; thereafter, the countersinks of the gear are finished, to define the workpiece axis for the subsequent grinding of the gear teeth.

The content of this Chapter is organized as follows. In Section 4.2 an exhaustive description of the existing scientific and technical literature is provided. In Section 4.3 the case study at stake is introduced. Both the hardware part of the monitoring system and the developed mathematical algorithm are described in Section 4.4. Numerical results are provided in Section 4.5: the technique is validated using simulated datasets and comparing the results obtained on data acquired through the monitoring system and a Coordinate Measuring Machine (CMM). An alternative alignment strategy is provided in Section 4.6, to emulate the results of the current manual procedure. The role of the MES is described in Section 4.7: its deployment on different time scales as well as its integration with the monitoring and control system and with a PLM are discussed. The contribution of MES to lean manufacturing is shown in Section 4.8. Finally, in Section 4.9 some conclusions and hints for future work are provided.

## 4.2 State of the art

A wide literature concerning the issue of workpiece localization and positioning is available: research works have been developed in several manufacturing fields to deal with this topic. The approaches can be grouped in two categories. The first is the definition of an error function, given by the sum, over all the sampled points, of the distances among the measured coordinates of the workpiece points and the corresponding nominal ones; through the least squares principle, transformations (translations and/or rotations) are evaluated such that the workpiece approaches the desired position as much as possible; this approach has been used, for example, by Anotaipaiboon et al. (2006); Li et al. (1998); Sun et al. (2009); Yau and Meng (1996). Another largely deployed algorithm, still belonging to this class of techniques, is the Iterative Closest Point, introduced by Besl and McKay (1992). The second kind of approaches consists in a minimax technique: the transformations are evaluated to minimize only the maximum value among the distances between the measured points and their expected positions (Chatelain and Fortin, 2001; ElMaraghy et al., 2004).

The industrial significance of this problem is confirmed by the number of patents concerning methods to identify the optimal positioning for axisymmetric workpieces that have been deposited. A first approach consists in employing mathematical means (eventually weighted). The technique proposed by Nagata et al. (1980) consists in measuring a set of internal and external points to identify the center of mass of a hollow cylinder. In (Kunugi and Sasaki, 1990), the coordinates of two points on the external surface of a workpiece are measured; the line joining these two points is supposed to be parallel to the symmetry axis. Similarly, in (Niewmierzycki, 1995), the circular sections orthogonal to the symmetry axis are approximated by the circle lying on the three measured points. In other applications, the Least-Squares (LS) principle is used. In (Noda et al., 2009), the diameters of several crosssections are measured, and a proper error function is minimized to reach the best workpiece positioning. In (Akerley et al., 1992), the line representing workpiece axis is obtained through a LS interpolation of the coordinates of cross-section centers. Finally, in (Sagues et al., 2001), the points measured on the external workpiece surface are interpolated through a LS circle representing the cross-section.

However, the solutions currently present in literature are not fully applicable to this case study because of the complexity of the measurement system. Hence, an algorithm based on the error function approach, integrated with non-linear least-squares interpolations, has been developed.

# 4.3 Description of the process

Gears used in the aeronautic field must be manufactured with great accuracy. Such requirement is obtained through a multi-step manufacturing process, in which the principal operations are machining, heat treatment and grinding processes.

Heat treatment is performed to achieve the necessary surface hardness and through-toughness of the gear. During this process, some form distortions are introduced; thus, in order to produce a gear with the necessary accuracy, the finishing tool must be able to remove all the distortions. Grinding is a mandatory technological process for the finishing operations, since it allows to achieve smooth surfaces and tight tolerances: this operation consists in removing the material in the form of small chips, through the mechanical action of a grinding wheel. However, since grinding is a costly operation (with respect to other machining processes), it should be utilized under optimal conditions (Alagumurthi et al., 2007). The definition of a new reference system is necessary before the grinding operation. This operation must be performed accurately: even a small misalignment could dramatically affect the result of the grinding process, leading to *defective* workpieces or even to low quality items to be rejected. This results in increased resources utilization, since a further effort must be performed to balance the defects of the process and produce the right number of items with the right quality level. In turn, excessive *waiting* and *inventory* may accumulate due to poor quality of the produced parts. Hence, the centering of the workpiece into the grinding machine plays a key role for process performance.

To improve the performance of the grinding operation, a pre-processing task is made to identify the workpiece axis that minimizes the geometrical distortions. This task is performed by finishing the two countersinks of the gear, which are used to position the part into the grinding machine. At present, in the considered process, the centering operation for the countersinks finishing is a manual, time-consuming task based on the experience of the operator. Given the importance of this task, an innovative, automatic machine has been developed to perform the centering operation. It has been equipped with a monitoring and control system consisting in measurement sensors, a control unit, and devices to automatically correct the position of the gear.

In this case study, two kinds of geometrical tolerances are addressed for the spur gears: total axial run-out and concentricity (ISO 1101, 2012). The former is prescribed for the side surfaces of the gear, and consists in the distance among two parallel planes, orthogonal to workpiece reference axis, which contain one of the gear side surfaces. Concentricity tolerance is prescribed for the bearing seats (BS) and the gear: it consists of a cylinder centered on the workpiece reference axis, that contains the mean points of all the couples of opposite surface points; in the following, eccentricity toler-

#### 4.4 Monitoring and control system

ance will be used, which is equal to one half of concentricity (because it is referred to a radius rather than a diameter). The prescribed tolerances rely on the definition of an ideal reference axis, which is simulated by the machine axis. To define the simulated datum axis, an innovative locking system is employed: it consists in a double row of spheres inserted in the internal groove of the wheel hub to define the simulated datum axis.

The case study is synthesized in Figure 4.1 in agreement with the methodology defined in Chapter 3: the classes of wastes affecting this process are highlighted. The process is schematically described and a synthesis of the mathematical technique is presented.

## 4.4 Monitoring and control system

In the monitoring phase, the measurements performed by the machine allow to determine the actual form of the workpiece. The choice of the features to be measured is strictly tied to the tolerances prescribed by the manufacturer. A representation of the measurement system is shown in Figure 4.2; it is composed of four displacement transducers with touching probes: three of them are able to move along the radial direction, to measure the distances from the machine axis to the bearing seats surfaces and to the pitch circle of the gear. The forth transducer moves along a direction parallel to the machine axis (identified with z) to measure the run-out of the side surface of the gear. The resolution of the sensors is equal to 0.1 µm; the repeatability error of the measurement system is  $10 \,\mu$ m, the reproducibility error is  $30 \,\mu$ m: these data have been provided by the sensors supplier. All the measured quantities are referred to arbitrary zeros. The sensors measuring the bearing seats and the side surface acquire  $N_p = 4096$  points at a fixed angular step, while the gear sensor measures one point per each tooth gap. An example of measured points is shown in Figure 4.3: the acquired measurements were transformed into a common Cartesian reference system.

In the control phase, a mathematical algorithm determines the transformations that must be applied to the workpiece to correct its position and minimize the residual errors. This procedure allows to optimize the utilization of the raw material in the heat treated part, in order to assure the respect of the geometrical product specification through the next grinding operations.

# 4.4.1 The mathematical technique

Since the measures collected from the different sensors are referred to arbitrary zeros, the first step of the algorithm is the transformation of mea-

	Process	des	scripti	on	
	Suppliers Gears leaving the heat treatment Gears ID & part type Planning	I N		Reusable Operators Centering machine Current state	
C O	Batch size: 1 Interarrival times & variability	P U T		Disposable Energy	R
M P O	Part dimensions Tolerance types & values Working parameters			rorerrigerants & compressed air Countersinks Finishing tool Current state	S O
N E N T S	Performance Finished gears Cycle time, work in process, throughput Failures incidence	0 U T		Reusable Operators Centering machine Current state	R C E S
	Quality Residual geometrical error Conformity to tolerances Incidence of non–conformities	P U T		Disposable Worn out tool Resource consumption	
	Wastes			Data—analysis	
Overpr	oduction	D s	ata ource	Four displacement transducers	
Waitin Transp	ig 🗸	D D P	ata rocessing	Least–squares interpolation	
Extra	processing	F	eature eneration	Coordinates of the centers of the sections	
Invent Motior	ory 🗸	F	eature xtraction	Eccentricity and runout errors	
Defect	is 🗸	···· D	ecision naking	Part rotation to minimize the sum of residual errors	

**Fig. 4.1** The application of the methodology introduced in Chapter 3 to this case study. Green colors represent physical quantities; black colors is used to denote data.



Fig. 4.2 A representation of the measurement system integrated into centering the machine.



**Fig. 4.3** An example of measurements collected by the monitoring system, transformed into a common Cartesian reference system.

sured data into a common Cartesian reference system, taking into account the positions of the four touching probes. Then, the three radially measured sections are considered. To be as general as possible, the intersection of a cylinder with a plane not orthogonal with the cylinder axis is considered: hence, the measured sections should be represented through ellipses. Thus, a non-linear least squares (Myers, 1990) interpolation is made; the extracted features are the coordinates of the centers of each section, denoted with  $\hat{O}_i = (\hat{x}_i, \hat{y}_i, \hat{z}_i)^T, i = 1, 2, 3$ , since they contain an information exhaustive to evaluate the eccentricity error. In the following, only the points  $\hat{O}_i$  will be used. The deployment of this feature allows to discard the whole set of measured points, reducing the computational cost of the algorithm and the influence of outliers and data imperfections. Two rotation matrices, corresponding to the corrective transformations, are applied to the points  $\hat{O}_i$ . In the machine, the clockwise rotation about the *z* axis (denoted with  $\alpha$ ) first occurs; it is described by the matrix  $R_z$ :

$$R_z = \begin{bmatrix} \cos \alpha & \sin \alpha & 0 \\ -\sin \alpha & \cos \alpha & 0 \\ 0 & 0 & 1 \end{bmatrix}$$
(4.1)

Hence, the coordinates  $\tilde{O}_i = (\tilde{x}_i, \tilde{y}_i, \tilde{z}_i)^T$  of the points after the rotation  $\alpha$  are given by:

$$\tilde{O}_i = R_z \hat{O}_i = \begin{bmatrix} \hat{x}_i \cos \alpha + \hat{y}_i \sin \alpha \\ -\hat{x}_i \sin \alpha + \hat{y}_i \cos \alpha \\ \hat{z}_i \end{bmatrix}$$
(4.2)

Then, the counterclockwise rotation about the *y* axis (denoted with  $\gamma$ ) is applied; it is described by the matrix  $R_y$ :

$$R_{y} = \begin{bmatrix} \cos \gamma \ 0 - \sin \gamma \\ 0 \ 1 \ 0 \\ \sin \gamma \ 0 \ \cos \gamma \end{bmatrix}$$
(4.3)

The coordinates

$$O_i = (x_i, y_i, z_i)^T$$

of the centers after the two transformations are given by:

$$O_{i} = R_{y}\tilde{O}_{i} = \begin{bmatrix} \tilde{x}_{i}\cos\gamma - \tilde{z}_{i}\sin\gamma\\ \tilde{y}_{i}\\ \tilde{x}_{i}\sin\gamma + \tilde{z}_{i}\cos\gamma \end{bmatrix} = \begin{bmatrix} (\hat{x}_{i}\cos\alpha + \hat{y}_{i}\sin\alpha)\cos\gamma - \hat{z}_{i}\sin\gamma\\ -x_{i}\sin\alpha + y_{i}\cos\alpha\\ (\hat{x}_{i}\cos\alpha + \hat{y}_{i}\sin\alpha)\sin\gamma + \hat{z}_{i}\cos\gamma \end{bmatrix}$$
(4.4)

Thus, for each section, the residual eccentricity is given by the following expression, as a function of the two admissible transformations  $\alpha$  and  $\gamma$ :

$$e_i(\alpha,\gamma) = \sqrt{x_i^2 + y_i^2}$$

$$= \sqrt{\left(\left(\hat{x}_i \cos\alpha + \hat{y}_i \sin\alpha\right) \cos\gamma - \hat{z}_i \sin\gamma\right)^2 + \left(-\hat{x}_i \sin\alpha + \hat{y}_i \cos\alpha\right)^2}$$
(4.5)

The points on the side surface of the gear also undergo both the rotations  $\alpha$  and  $\gamma$ , according to Equations 4.2 and 4.4. Since the prescribed tolerance is an oscillation along the *z* axis, only the *z* coordinates of the points after the rotations are evaluated:

$$z_k = (\hat{x}_k \cos \alpha + \hat{y}_k \sin \alpha) \sin \gamma + \hat{z}_k \cos \gamma, \quad k = 1, \dots, N_p$$
(4.6)

The residual eccentricities and the oscillation of the points on the side surface of the gears are collected into an objective function, which is given by the sum of the four residual errors divided by the corresponding prescribed tolerances  $w_i$ , i = 1, 2, 3 for the three radially measured sections and w for the axially measured feature:

$$F(\alpha, \gamma) = \sum_{i=1}^{3} \frac{e_i}{w_i} + \frac{(\max_k z_k - \min_k z_k)}{w}$$
(4.7)

In Figure 4.4, a schematic representation of the transformations applied to the points is shown. The developed approach is similar to the Iterative Closest Point (ICP) method; however, this technique is not fully applica-

50



Fig. 4.4 A schematic representation of the transformations applied to the points. Picture is not in scale.

ble in this case study, because reference points are not given for the side surface: the manufacturer is only interested in minimizing the oscillation. Further, only two corrective transformations are allowed while the ICP acts on six degrees of freedom. Thus, the objective function proposed in the ICP method has been adjusted to take into account these constraints.

The algorithm has been implemented in a Matlab code; the values for  $\alpha$  and  $\gamma$  that minimize *F* are evaluated through the fminunc function, which uses a quasi-Newton method.

For the application in this machine, the rotation  $\gamma$  is transformed into a displacement  $\Delta$  exerted by a translator with a given *z* coordinate *H*; the value for  $\Delta$  is evaluated through the following equation:

$$\Delta = H \sin \gamma \tag{4.8}$$

Finally, the two transformations are mathematically applied to the centers of the ellipses and to the points on the side surface to check whether the final workpiece configuration is in compliance with the maximum admissible tolerances or not. The algorithm can be synthesized as follows:

- 1. Transform data into a common reference system
- 2. Interpolate the 3 measured sections through LS ellipses
  - a. Evaluate the coordinates of the centers
  - b. Evaluate the eccentricities as a function of  $\alpha$ ,  $\gamma$
- 3. Evaluate the residual oscillation of the side surface as a function of  $\alpha$ ,  $\gamma$
- 4. Build the objective function *F*
- 5. Evaluate  $\alpha$ ,  $\gamma$  that minimize *F*
- 6. Transform the rotation  $\gamma$  into a displacement  $\Delta$
- 7. Evaluate whether the final positioning complies with the prescribed tolerances or not

## 4.5 Results

In order to test the presented technique, the algorithm has been first applied by simulating the data acquired in the measurement process: 30 simulations have been run, in which a spur gear – equal to the ones produced by the manufacturer – without geometrical errors is measured. The only source of error affecting these data is a Gaussian measurement noise with zero mean and standard deviation equal to  $10 \,\mu\text{m}$ . For each simulation, a different initial workpiece position, randomly determined, was assumed. The residual positioning errors after the centering process are synthesized in Table 4.1. The oscillation of the side surface exhibits high values, compared to the eccentricities, because no interpolations are made on this feature; the largest part of the oscillation is due to the Gaussian noise: the 99.7% of the simulated points lies into an interval whose width is equal to 6 times the standard deviations of the residual errors lead to state that the developed algorithm exhibits a good reproducibility of the final configuration.

After assessing the performance of the algorithm on simulated data, real data have been deployed as input. A sample gear has been measured both through a CMM and the sensors system installed on the monitoring and control system of the manufacturing machine. The deployed CMM is a Dea Global Advantage; the diameter of the touching probe is 2 mm. To have more reliable results, the measurement on the manufacturing machine has been repeated 5 times: at the end of each session, the gear was removed from the machine and placed again on it.

First, to evaluate the quality of the measurement system installed on the manufacturing machine, the data acquired by the two systems have been compared. To do this, the measured data have been preprocessed in order to have equivalent workpiece configurations: for each dataset, the centers of the two bearing seats sections have been identified and moved onto the z axis through a set of translations and rotations; the same transformations have been applied to the whole set of measured points. Since the positions of the workpiece into the manufacturing machine and into the CMM are now equivalent, the differences in the data may only be attributed to the measurement system. The amplitudes of the oscillations measured on the two bearing seats (i.e. a roundness error), the residual eccentricities of the pitch circles and the amplitudes of the oscillations of the side surfaces have been compared. A comparison among acquired data is reported in Table 4.2 and in Figure 4.5: the amplitudes of the oscillations measured on the bearing seats and on the side surface, and the pitch circle eccentricity are slightly larger on the data acquired through the manufacturing machine. However, the results obtained through the two measurement systems are comparable. The small values for the standard deviation of the oscillations

#### 4.5 Results

-						
Cimulation	Corr	ections	Eccer	ntricities [µm	]	Side surface
Simulation	α [°]	$\Delta$ [mm]	Lower BS	Upper BS	Gear	oscillation [µm]
1	308.28	3.53	0.2	0.0	0.7	66.3
2	104.36	2.26	0.2	0.0	3.2	67.3
3	298.28	2.63	0.5	1.1	0.1	73.4
4	289.00	4.91	0.2	1.2	2.2	75.3
5	60.61	4.49	0.6	0.0	1.5	70.2
6	328.35	3.88	0.5	0.0	2.1	78.8
7	118.97	6.35	0.6	0.0	4.0	71.5
8	306.27	1.83	0.3	2.9	1.3	69.1
9	6.47	4.56	0.4	0.0	4.3	72.6
10	352.85	4.80	0.1	0.1	2.2	76.3
11	327.86	1.87	0.1	0.9	1.5	71.6
12	14.81	6.46	0.1	0.4	3.4	66.8
13	61.48	2.88	0.3	0.0	4.0	72.3
14	133.15	6.57	0.3	0.0	1.9	71.4
15	143.00	9.14	0.1	3.2	4.6	84.7
16	68.14	10.81	0.1	0.0	3.4	80.2
17	53.75	12.89	0.4	0.1	2.5	90.2
18	4.78	5.07	0.1	1.4	0.3	74.7
19	57.72	5.83	0.4	1.4	0.0	71.6
20	69.91	2.10	0.5	0.0	1.1	75.9
21	31.99	7.42	0.5	2.4	1.9	71.8
22	238.98	6.41	0.1	0.0	1.8	74.3
23	29.93	9.22	0.2	3.3	0.0	80.8
24	14.69	10.41	0.3	0.0	2.8	84.4
25	80.38	1.83	0.4	1.7	1.6	77.8
26	319.08	6.71	0.3	0.5	2.1	78.3
27	344.61	2.15	0.3	0.0	0.7	69.4
28	42.54	2.43	0.8	2.1	1.2	72.7
29	80.69	7.50	0.2	0.0	0.9	67.3
30	28.88	5.39	0.3	0.0	2.3	69.7
Mean			0.3	0.8	2.0	74.2
Std. deviation			0.2	11	13	57

**Table 4.1** Residual errors after the alignment of a simulated workpiece with 30 different initial positions.

measured on the manufacturing machine confirm the good reproducibility of the measurement system.

After having assessed the metrological quality of the measurement system, the alignment algorithm was applied to the datasets acquired both through the CMM and the manufacturing machine. To have comparable initial configurations, the reference system of the data measured on the CMM was transformed to simulate the locking system employed on the manufacturing machine. The values for the corrective transformations and the residual errors are shown in Table 4.3 and in Figure 4.6: in each plot, the curves of the data measured on the manufacturing machine exhibit similar behaviors and confirm a good reproducibility of the measurement system of the machine. However, systematic differences can be detected through the



**Fig. 4.5** Comparison between the measurements acquired through a CMM and the monitoring system: blue points represent the acquired data; black points correspond to the reference circles; red points are the centers of the gear.

4.6 Alternative positioning technique

Manager	Bearing sea	ts oscillation [µm]	Gear eccentricity	Side surface
Measurement	Lower BS	Upper BS	[µm]	oscillation [µm]
CMM	16.1	16.8	15.0	82.8
1	23.0	31.4	31.2	85.0
2	23.1	29.4	40.7	85.9
3	22.6	28.1	36.5	86.7
4	23.6	30.5	28.3	84.8
5	22.3	33.5	18.2	82.4
Mean	22.9	30.6	31.0	85.0
Std. deviation	0.5	2.0	8.6	1.6

**Table 4.2** Comparison between the measurements acquired through a CMM and the monitoring system of the manufacturing machine.

**Table 4.3** Residual errors after the alignment of a workpiece measured both through a CMM and the monitoring system of the manufacturing machine.

N i	Corre	ctions	Eccer	ntricities [µm]	]	Side surface
Measurement	α [°]	$\Delta  \mu m$	Lower BS	Upper BS	Gear	oscillation [µm]
Tolerances			50	50	50	120
CMM			2.6	8.4	12.2	75.4
1	232.01	76.8	9.6	2.3	36.3	75.8
2	225.86	88.6	10.9	2.8	45.3	74.7
3	226.16	81.8	9.8	3.8	40.9	74.4
4	239.34	77.9	10.4	2.6	36.3	73.1
5	252.93	65.7	8.4	0.7	26.8	73.1
Mean			9.8	2.4	37.1	74.2
Std. deviation			0.8	1.0	6.1	1.0

comparison among the results obtained from data acquired with the two measurement systems. Thus, in order to reduce the impact of such systematic differences, the measurement system should be further adjusted. Two additional gears have been measured on the manufacturing machine: even these results (shown in Table 4.4) confirm the good reproducibility of the measurement system, and its capability to provide an acceptable workpiece centering.

## 4.6 Alternative positioning technique

The introduction of the monitoring and control system presented in this Chapter allows the automation of a manual task, leading to improved accuracy and repeatability of the process. The developed mathematical technique is also able to predict whether, after the repositioning operation, the configuration of the workpiece satisfies the tolerances or not.

However, the locking system of the machine results in a gear positioning constraint which is not present in the manual task: when the centering is performed, the row of spheres is locked; since it is very close to the lower



Fig. 4.6 Residual errors after the alignment of the workpiece measured both through a CMM and the monitoring system.

#### 4.6 Alternative positioning technique

Magazinamaant	Corre	ections	Ecce	ntricities µm		Side surface
Measurement	α [°]	$\Delta$ [µm]	Lower BS	Upper BS	Gear	oscillation [µm]
Tolerances			50	50	50	120
Workpiece 1						
1	270.97	11.0	0.5	13.9	5.4	38.8
2	299.55	14.8	4.0	14.3	0.5	42.3
3	268.91	11.1	2.5	12.1	5.1	40.0
4	278.28	12.3	3.0	14.6	3.5	35.3
5	291.30	13.9	7.5	34.4	0.7	36.7
Mean			3.5	17.9	3.0	38.6
Std. deviation		-	2.3	8.3	2.1	2.5
Workpiece 2						
1	64.01	45.7	8.3	0.0	5.4	58.6
2	81.36	43.5	11.1	14.6	1.5	98.3
3	61.50	41.4	7.8	0.0	4.5	51.2
4	59.24	53.5	16.8	25.0	4.8	92.9
5	44.16	40.0	11.7	27.4	11.5	51.6
Mean		•	11.1	13.4	5.5	70.5
Std. deviation			3.2	11.8	3.3	20.7

**Table 4.4** Residual errors after the alignment of two workpieces measured through the monitoring system of the manufacturing machine.

bearing seat, the movement of this feature is very limited. Conversely, in the manual alignment, the operator has a higher number of degrees of freedom and is free to move this feature; thus, he can increase the positioning error on the lower bearing seat in order to compensate errors on other features. Hence, due to such constraint, it is possible that the algorithm for automatic positioning does not identify a repositioning within the prescribed tolerances, even if there would be at least one acceptable positioning reachable by increasing the error for the lower bearing seat. Thus, to avoid the rejection of acceptable parts, in case the automatic positioning does not lead to acceptable configurations, a second algorithm, based on the ICP algorithm (Besl and McKay, 1992), is run to simulate the manual operation. In case the result of the ICP technique is acceptable, the gear is manually placed into another machine for countersinks finishing, and centered through dial indicators.

Conversely, in case even the second algorithm does not find a solution able to satisfy the prescribed tolerances, form error may be arisen in previous manufacturing operations. To solve this issue, the value for the planarity error of the side surface is evaluated. If the value for this form error is high, a reworking operation is performed: the side surface is finished again to reduce the form error, and then the gear centering process is restarted. Otherwise, if the value of the planarity error is low, the issue can be due to a misalignment between the bearing seats, or to a non-orthogonality condition between the side surface and the axis of the gear. An overall flow chart of the operations performed by the monitoring and control system is shown in Figure 4.7. In Table 4.5 the results concerning 12 sample pieces are synthesized: the automatic centering algorithm is able to provide a result that satisfies the prescribed tolerances for 5 parts; acceptable results are obtained for three further parts through the manual positioning. Finally, the positioning of four gears is not acceptable neither with the automatic operation nor with the manual one. One of them is affected by a high value for the planarity error of the side surface; thus this feature must be finished again, to reduce the form error before countersinks grinding. The three remaining workpieces exhibit low values for the planarity error, and the non-acceptable centering can be due to form error of the bearing seats or a deformation of workpiece axis: because of this, these three part are rejected.

## 4.7 The role of MES

The whole strategy shown in Figure 4.7 can be synthesized into a unique software to be installed into the PLC of the manufacturing machine or into an external computer connected to the machine; this stand-alone solution could be used by the operator in charge of loading the part to be finished and running the operation. The output consists in a flag ("the alignment is acceptable" or "the alignment is not satisfactory"), and in the evaluation of the residual errors. The deployment of an automation system allows to improve the quality of the output: the automatic alignment results in the configuration with the minimum residual errors; conversely, the manual technique leads to a configuration with higher residual errors, although the tolerances are satisfied. The higher positioning reproducibility provided by the monitoring and control system also enables a reduction of variability in the quality of the finished parts.

The replacement of a manual task with an automatic machine also allows to reduce the variability of centering time. Furthermore, the flexibility of the process is enhanced: in case a new kind of gear is to be centered, no training for the operators is required; only an adjustment to the software is necessary, to enable the objective function in Equation 4.7 to take into account the proper features, and to set the correct values for the tolerances.

Nevertheless, this monitoring and control system is also able to provide data which can be transformed into information useful for further purposes. The integration of this technique with a MES allows to collect information concerning both the results of the centering process and the performance of the process (e.g. cycle time, work in process, downtimes).



Fig. 4.7 Synthesis of the operations performed by the monitoring and control system.

		14	utomati	c alignment			2	Aanual	alignment	
	Ecc	centricities [p	[ur	Side surface $e^{A}$ oscillation [µm]	Acceptable?	Eccent	ricities [µ	[IJ	Side surface oscillation [µm]	Acceptable?
Workpiece 1	Lower	BS Upper B	S Gear			Lower BS	Upper B5	5 Gear		
Tolerances	50	50	50	120						
1	9.7	70.8	17.7	43.8	×	13.3	31.1	57.6	46.4	×
7	21.8	35.2	34.2	76.8	>					
ς	24.6	51.3	34.3	30.2	×	16.6	28.5	50.0	49.1	\$
4	22.7	7 140.6	0.0	167.7	×	13.8	29.5	128.4	125.5	×
ŋ	38.6	40.2	20.4	46.2	>					
9	7.9	87.9	2.5	121.2	×	13.7	40.5	63.2	110.1	×
	16.4	22.0	0.0	174.9	×	4.8	20.3	22.9	128.2	×
∞	4.7	30.0	20.4	44.0	>					
6	5.7	22.4	26.2	62.6	>					
10	9.0	42.9	0.0	119.4	>					
11	3.8	22.8	109.3	104.4	×	1.2	21.4	25.5	120.0	>
12	14.9	18.2	57.8	160.9	×	2.9	24.0	30.6	117.0	>

Table 4.5 Results of the alignment strategy on 12 workpieces.

# 4.7.1 Integration between the monitoring and control system and a MES

As stated in Chapter 2, one of the roles of MES is to collect data from the shop-floor, analyze it and dispatch the resulting information to the departments that can benefit from this feedback.

Several commercial solutions are available to deal with information concerning the performance of the process; thus, here the attention is focused on product and process quality data. The integration of the shown monitoring and control system with a MES enables to analyze and use the collected data at different time-scales with different purposes.

On the short-medium term, MES allows to check whether the process is stable or not. Further, when instability symptoms appear, MES can predict when the process is going to be out of control and produce parts not matching the expected quality. Thus, setup or maintenance interventions can be planned in a preventive approach, also taking into account further constraints, such as the availability of operators or already planned downtime. This kind of prediction is helpful to avoid producing parts that will be rejected, thus reducing waste.

On the long term, the information collected into the MES can be further analyzed to extract historical trends, to synthesize criticalities and identify the sources of issues and wastes. The integration of a traceability system strongly supports this functionality: in this case study, each workpiece is identified by a unique ID. Information concerning each gear, such as the time at which the centering operation occurs and the expected results of the alignment, can be collected and stored into a database. This information can be useful to monitor the results of the centering process over time, and identify the reasons for possible decays or drifts; however, a careful analysis of these data is necessary, since issues identified on the centering machine can be due to inefficiencies in the upstream workstations. The results of this analysis can be shared with different departments of the company. For example, the business unit can benefit from this information to define new strategies, or to correct the previously defined ones; the design department can use this experience-driven knowledge to improve the design of a product or process. The feedback information provided by the MES supports the test and validation of new process or product releases. This, in turn, enables the implementation of kaizen practices for continuous improvement, such as the PDCA cycle.

## 4.7.2 Integration between MES and PLM

In the previous Chapters, the functionalities of PLM and MES and the benefits that result from their deployment have been introduced. The two information systems have different purposes: PLM contains information concerning the to-be product and the production process; conversely, as-built data are stored in the MES. The integration among these two systems allows to create a feedback information mechanism that can enhance the performance of the production process and the quality of the manufactured parts.

When a new product or production process is released, the PLM contains all the information that, according to the project, allows to meet the required specifications. Then the ramp-up phase is run, and a tight monitoring is necessary to detect any difference between the real products and the expected output. A careful analysis of the data collected in this phase is necessary: they can be rich of useful indications to optimize process and product design, allowing to improve the performance of the production process and the quality of the manufactured parts. The deployment of a MES is also useful after the ramp-up phase, when the steady state is reached: a continuous analysis of shop-floor data allows to monitor the behavior of the process and detect systematic trends, criticalities or deviations. For example, the MES can trace process variability: as machines get aged, the quality of the products can be lower and machined parts may result out of tolerances. The results of such analysis have to be used to identify strategies or practices for performance and quality improvement. These actions lead to redesign or revise some operations: such changes must be integrated into the PLM system, in order to store this experience-driven acquired knowledge, and make it available for the future production. A system for product traceability, based, for example, on RFID tags, would enhance this task: correlations between the state of the process and the quality of the products can be extracted, and the causes that led to a specific kind of issue can be detected. Furthermore, the knowledge collected by the PLM can be shared even among several different plants or with different suppliers; thus, the expertise acquired in one place can be standardized and made available elsewhere.

# 4.8 The support to lean manufacturing

Beside improving the efficacy of the information collected at the shop-floor and enabling to deploy it with several purposes and at different time-scales, the MES also enhances the implementation of lean manufacturing techniques to improve the performance of the process.

According to the scheme in Figure 4.1, the first source of waste to be reduced is *defects*. The monitoring and control system allows to improve the precision of the gear positioning into the machine and its reproducibility, resulting in better quality for the finished parts and in the prevention of issues that can affect workpiece quality and lead to their rejection. This, in

#### 4.9 Conclusions

turn, allows to reduce the activities necessary to improve the quality of the output, such as reworking operations resulting in time waste. The automation of a manual process and the resulting quality improvement also allow to face with *inventory* and *waiting* reduction. The reduced process variability enables to substantially decrease the lead time and the work in process: the measurement cycle and the repositioning of the gear take less than 90 seconds, and the algorithm computes the results in real-time; the previous manual procedure takes a much larger amount of time. With the deployment of this new technology, the lead time and the Work In Process (WIP) are expected to be reduced by 40%; the improved precision of the performed operations should lead to a 50% reduction of parts to be rejected or reworked.

The reduction of these waste sources also allows to improve the sustainability of the process. From the economical point of view, the cost of the processes to be executed after the heat-treatment should decrease by, approximately, 60%. From the environmental perspective, the first consequence of the presented system is a reduction in the amount of material necessary to have good-quality finished products; nonetheless, the reduction of extra processing also allows to reduce the energy demand of the process: this is not a negligible task for manufacturers, given the increasing cost of energy and the tighter constraints to be reached in energy and pollution footprint.

## 4.9 Conclusions

In this Chapter, an algorithm integrated into a manufacturing machine for the automatic centering of a spur gear after the heat treatment has been introduced. The technique is fully described in Section 4.4.1. The approaches currently available in literature are not fully satisfactory to deal with our case study, since they do not allow to optimize the position of a points cloud in real-time. Conversely, patented solutions, including algorithms for optimal workpiece alignment, are not sufficiently accurate to satisfy the tight tolerances usually prescribed in the field of aeronautics. A correct alignment is mandatory, since grinding is a costly operation and even a small misalignment may heavily affect the result of this operation.

In the considered case study, before the introduction of this innovative machine, the positioning of the workpiece was a manual task based on the experience of the operator. The replacement of this procedure with an automatic monitoring and control system allows to improve the precision and the reproducibility of the positioning of the gear into the machine: the enhanced workpiece positioning results in higher quality of the ground gears, with lower scraps and reduced reworking operations. Further, the reduction of variability sources allows to decrease wasted, leading to improved process performance. The importance of integrating the monitoring and control system into a MES has been shown, as well as the necessity to share the results of MES analysis with the PLM.

In this work, the attention was focused on geometrical issues affecting the gears; in future research, new functions can be integrated into the monitoring and control system. First, the analysis of the measured data can be extended to the verification of form tolerances, such as the flatness (defined in the norm ISO 12781:2011) of the side surface, the roundness (ISO 12181:2011) of the bearing seats and the cilindricity (ISO 12180:2011) of the whole spur gear. Further, the cooperation between MES and PLM can enhance a comparison between the as-is and the to-be states of the part, to identify focused issues.

Beside the functionality concerning the workpiece centering, additional tasks can be implemented to enrich the information provided by the MES. Sensors to monitor the countersinks finishing can be integrated into the machine; for example, the temperature of the tool, the current requested by the machine or the vibration produced by the tool can be monitored: these variables have been identified as significant to predict the quality of the finished part and to evaluate the remaining life of the tool itself (Abellan-Nebot and Romero Subirón, 2010).

**Note** This work has been supported by Regione Piemonte under the research project MANUM5, "Study, design, development and realization of a reconfigurable production system for aeronautical gears, holistic of methods, fixtures, tools, coolant, measurement and human interaction". The results of this research have been published in:

- G. Barbato, P. Chiabert, G. D'Antonio, M. De Maddis, F. Lombardi, S. Ruffa. *Method for automatic alignment recovery of a spur gear*, International Journal of Production Research (2015). doi:10.1080/00207543.2015.1064180
- G. D'Antonio, J. Sauza Bedolla, P. Chiabert and F. Lombardi. *PLM-MES integration to support collaborative design*. In: Proceedings of The 20th International Conference on Engineering Design (ICED15) 10: Design information and knowledge management, pp. 81–92. Milano, Italy, July 27–30, 2015

Furthermore, the following patent application has been deposited:

 F. Lombardi, G. Barbato, P. Chiabert, M. De Maddis, F. Ricci, S. Ruffa, G. D'Antonio, S. Milletari and R. Siccardi. *Metodo di recupero automatico dell'allineamento di un pezzo in lavorazione, in particolare una ruota dentata, e relativa macchina di lavorazione automatica*, application number TO2014A000066.

# Chapter 5 MES for monitoring of product geometry failure

Abstract The deployment of Additive Manufacturing (AM) processes had a rapid and broad increase in the last years, and the same trend is expected to hold in the near future. To better exploit the advantages of such technology, the use of appropriate information tools is mandatory. However, today there is a lack of software applications devoted to this innovative manufacturing process. In particular, at the state of the art, there is no application of MES in this technological fields. To overcome this issue, a laboratory case study has been developed to show the importance of MES deployment even in additive manufacturing. Furthermore, the importance of supporting design with shop-floor data is discussed: hence, a framework is presented for the integration between MES and Design for Additive Manufacturing (DFAM), a set of methods and tools helpful to design a product and its manufacturing process taking into account AM specificities from the early design stages. The case study is developed in the form of a proof of concept, in order to understand the advantages of such cooperation: the obtained results are promising, hence an online implementation is recommended.

## 5.1 Introduction

In the ASTM standard, Additive Manufacturing (AM) is defined as "the process of joining materials to make objects from 3D model data, usually layer upon layer, as opposed to subtractive manufacturing methodologies, such as traditional machining". Several synonyms are also defined for AM: additive fabrication, additive processes, additive techniques, additive layer manufacturing, layered manufacturing, rapid manufacturing and freeform fabrication. To perform an additive fabrication process, the model data of the object to be produced is decomposed into a number of 2D cross sections and a file is created: common standards for such file are STL, VRML or the recently introduced 3MF. Then, the file is sent to the AM machine, which adds material layer by layer to produce the physical object.

To fully exploit, support and promote this technology and its advantages, appropriate information tools are necessary. However, currently there is a lack of software applications devoted to this innovative manufacturing process. In particular, few work has been done in the deployment of real-time monitoring systems to improve the quality of the product and the stability of the process. However, the topic of machine monitoring has industrial relevance. EOS has deposited a patent (Perret and Philippi, 2014) for the automatic calibration of a machine through a scanner; the MIT deposited a patent (Perez et al., 2015) for an AM process control methodology including a camera to acquire images of the object being manufactured.

In traditional manufacturing processes, monitoring and control systems are integrated with Manufacturing Execution Systems (MES). However, in literature no applications of MES exist in the field of AM. Thus, the aim of this paper is to extend the state of the art providing a framework for the deployment of new information tools for AM; a case study for the monitoring of surfaces created by 3D printers is introduced to validate the framework. Beside quality monitoring, the information collected by such monitoring system can be deployed during the design phases for product or process adjustments; hence, the integration between MES and Design For Additive Manufacturing (DFAM) is also discussed. DFAM is a set of methods and tools helpful to design a product and its manufacturing process taking into account AM specificities from the early design stages (Laverne et al., 2015). The scientific contribution of this paper is to propose and validate, through a use-case, a framework for the integration of MES and DFAM.

The remainder of the paper is organized as follows: in Section 5.2 the state of the art on AM technologies and methodologies is provided. In Section 5.3 the developed case study is introduced; the monitoring and control system used to deal with the problem is described in Section 5.4. The results of the application are described in Section 5.5. The role of MES and a framework for its integration with DFAM are discussed in Section 5.6; their support

#### 5.2 State of the art

to lean manufacturing is described in Section 5.7. Finally, some conclusive remarks and hints for future work are provided.

# 5.2 State of the art

## 5.2.1 AM processes

Several AM processes are currently available: 7 process categories are defined in the ASTM standard; previously, a classification of the production technique had been given by Kruth (1991). The choice of the fabrication process is strictly tied to the deployed material: polymers, metals, ceramics and organic materials are among the main ones (Doubrovski et al., 2011). Material extrusion is one of the most deployed methodologies: a thermoplastic material is heated over its glass transition temperature and extruded through a nozzle in a controlled manner. The extruded material is used to print 2D sections successively, one on-top of another, until the object is complete. ABS and PLA are the most common thermoplastic polymers in material extrusion, because of their relatively low glass transition temperatures (Mellor et al., 2014; Rao et al., 2015). Recently, an innovative technique, named CLIP, has been developed and patented: a continuous liquid interface is used to build 3D objects; it is much faster than "traditional" additive techniques (Tumbleston et al., 2015). Metal AM techniques are mainly based on powder; the mostly used materials are steels, pure titanium and titanium alloys, aluminum casting alloys, and this list of alloys is continuously growing as new processes are developed (NIST, National Institute of Standards and Technology, 2013). Nevertheless, currently no process is able to create net shape parts, and a post-processing operation is necessary (for example, to remove supports or to finish the surface). This kind of operations can lead to some deformations and, sometimes, to destruction of the AM part. Consequently, post-processing operations can be a source of functional problems for the part: dimensional, shape, roughness errors, etc. Thus, a stand-alone AM implementation is not yet feasible, and the integration among production processes is necessary (Mellor et al., 2014).

AM processes allow an extraordinary design freedom: this advantage makes feasible shape complexity and geometry customization levels that are not reachable with traditional manufacturing technologies (Vayre et al., 2012). Furthermore, material waste is reduced, since structural parts without functionalities for the user can be unnecessary (NIST, National Institute of Standards and Technology, 2013). Time-to-market is shortened with respect to traditional processes, both because the design can be quicker and because additive fabrication totally occurs in a single place (while traditional processes can take place in different locations). This makes feasible 5 MES for monitoring of product geometry failure



**Fig. 5.1** Comparison among the timelines for industrial revolutions, traditional manufacturing and additive manufacturing technologies.



Fig. 5.2 Focus on the timeline for AM development.

a just-in-time production approach, as well as a reduction in transportation problems, cost and energy consumption (Vayre et al., 2012).

AM processes are quite new, compared to traditional manufacturing processes: in Figure 5.1, a comparison between the timelines for industrial revolutions and the development of traditional and additive manufacturing techniques is shown; however, AM development has been very quick and rich of milestones (see Figure 5.2). This class of process is still rapidly changing, and new applications arise as new materials become available. This rapid growth is also due to the support of the technological tools that led to the third and the fourth industrial revolutions. Currently, AM technology is mainly deployed in aerospace, automotive and biomedical devices manufacturing. The high customization level allows to profitably use freeform fabrication in personalized products and in the production of small lots (Atzeni and Salmi, 2012; Mellor et al., 2014). Beyond end-user products, an indirect usage of AM is also feasible, for example to develop and produce tools for conventional machines as well as for reverse engineering of components which are out of production or under maintenance (Vayre et al., 2012).

The AM market is significantly growing in every manufacturing sector: according to the last report by Wohlers Associates (2014), the global addi-

68

#### 5.2 State of the art



Fig. 5.3 Hype Cycle for AM technology Basiliere and Shanler (2014).

tive manufacturing market in 2013 was \$3.07 billion (corresponding to 2.8 billion, given a change rate equal to 1 dollar = 1.11 euro); the yearly increase, with respect to the 2013, is 35%.

## 5.2.2 AM methodologies

The hype cycle presented in Figure 5.3 (Basiliere and Shanler, 2014), envisages that in the next 5-10 years AM will be deployed in several different production fields such as bioprinting systems, oil and gas, and medical devices. Currently, research in AM is mainly focused on the development of new materials and fabrication techniques; conversely, little investigation is performed on the methods for designers. Yet, the design has a remarkable impact on the downstream phases, e.g. production, distribution, utilization and disposal. The DFAM methodologies are now a major issue to exploit in an appropriate way the potential of AM technologies for product development (Laverne et al., 2015). Furthermore, digital fabrication and on-demand production dramatically changed the manufacturing paradigms: a customer can look for a product in a digital catalogue, customize it and send the resulting file to a small firm to fabricate it (Doubrovski et al., 2011). AM allows to produce huge quantities as well as small volumes of a product, with little or no stock; however, this technology is not yet used for the production of large products lots, because of economic reasons (Atzeni and Salmi, 2012).

To manufacture high-quality products, the properties of the material must be well-known; these properties can strongly vary according to the production parameters, such as the orientation of the part in the 3D printer, the build speed and the tool path. Thus, the deployment of a consistent and structured design approach is mandatory. In traditional processes, Design for Manufacturing (DFM) practice is deployed to eliminate production issues, and minimize manufacturing, assembly and logistics costs (Gibson et al., 2010). However, additive processes have different constraints and DFM cannot be used as it is; it must be re-thought to take into account the unique capabilities of AM, in order to fully exploit the advantages of such technology and consider its limits from the early design stage (Ponche et al., 2012) in particular, new design tools are necessary to define and explore product shape and properties, new materials, new efficient manufacturing processes, and to assess lifecycle costs (Huang et al., 2015).

DFAM is a set of methods and tools helpful to design a product and its manufacturing process taking into account AM specificities from the early design stages: DFAM allows to determine an optimized process planning from the functional specifications (Ponche et al., 2014). Mançanares et al. (2015) developed a method to select the best manufacturing process for a part, in order to best satisfy the target. Rosen (2007) defines DFAM as the synthesis of shapes, sizes, geometric mesostructures, material compositions and microstructures to best utilize manufacturing process capabilities to achieve desired performance and other life-cycle objectives. He also defines the DFAM structure shown in Figure 5.4. Design is represented by the right-left flow: functional requirements are transformed into properties and an appropriate and realistic geometry; a process planning is performed to formulate a potential manufacturing process. On the left-right flow, the designed product and its fabrication are simulated to determine how well the original requirements are satisfied. Another structure for DFAM is formulated by Ponche et al. (2014) (Figure 5.5): their methodology is organized in three steps: determination of part orientation into the machine; topological optimization of the part; optimization of the manufacturing paths. This methodology allows to take into account the characteristics and constraints of the chosen AM process from the early stage of design.

Thus, different definitions for DFAM are given in literature. Nonetheless, this work is mainly focused on information systems to collect and analyze data from the additive machine: the spread of MES in the field of additive technology is discussed in the following Section.

# 5.2.3 MES for Additive Manufacturing

At the state of the art, there is no application of MES in the field of additive manufacturing. In literature some predictive models based on the values of machine parameters have been developed. Vijayaraghavan et al. (2014) formulated a model to predict the wear strength of a part, based on layer

#### 5.2 State of the art



Fig. 5.4 Design for DFAM methodology extracted from (Rosen, 2007).



Fig. 5.5 Design for DFAM methodology extracted from (Ponche et al., 2012).

thickness, orientation, air gap, raster angle and width. Byun and Lee (2006) developed a decision making strategy for part orientation based on surface quality, building time and part cost. Sood et al. (2012a,b) used air gap, raster angle and raster width to predict the wear rate and the compressive resistance of a part. Boschetto and Bottini (2014) developed a model to predict dimensional deviations of fabricated parts as a function of the process parameters. However, machine parameters are not sufficient to predict possible anomalies and failures; hence, the use of sensor-based monitoring systems is necessary. Rao et al. (2015) measured vibrations and temperatures to optimize process conditions, in order to obtain the best surface roughness and to real-time detect possible drifts. Bukkapatnam (2006) used a set of accelerometers to trace in real-time variations in process dynamics and to early detect possible anomalies. Dimensional and geometrical measurement and control of the manufactured parts also plays a key role. Faes et al. (2014) deployed an optical sensor to measure layer width and height and to control the geometrical error in the z direction (perpendicular to slicing direction, traditionally vertical).

Two kinds of technologies can be used to evaluate the quality of the produced geometry. Contact measurements through a probe are not currently feasible in AM, since the part is fixed in the manufacturing environment, and the room for a moving probe is very scarce; this technology can be used for offline measurements, after removing the part from the machine. For example, the object can be measured onto a CMM; this process leads to high quality measurements, but requires a huge amount of time, both for the measurement and for the creation of the measuring path. An alternative technology is given by non-contact measurements. Several different tools can be used to measure an object in a few seconds, according to the available room, the dimension of the object to be measured and the desired accuracy of the output. For example, optical measurements can be performed and different technologies can be chosen (such as time of flight or stereoscopic vision).

At the state of the art, the deployment of real-time monitoring systems in AM is restricted because of the lack of proper smart sensors. This is mainly due to two factors. The first one is the reduced access to the build chamber: in high-performance additive machines the printing environment is closed to keep constant the temperature, and cannot be opened. On the other side, the temperatures are too high for a common measuring device. The second factor is the need for intensive computing power: due to the very small time scale at which additive phenomena occur, fast and reliable in-situ measurements and analyses must be performed.

Given the importance of MES role in AM combined with a monitoring system, in the next Sections the core part of this research work is introduced: a theoretical framework is presented for the integration between MES and DFAM, and describe a case study for optical measurement of an additive manufactured part.

# 5.3 Description of the process

The test sample used in this study consists in a set of toys representing hollow ducks fabricated through a BFB 3D Touch machine, an additive machine based on the fused deposition modeling technique. Although a toy, the surface geometry of this part is not trivial at all, from the building process perspective. The dimensions of the ducks (as designed) are 80.6 mm (length), 45.6 mm (width) and 51.5 mm (height). The objects have been produced both in PLA and ABS. After the production of some parts, a non-detected issue in axes calibration arose, leading to surface *defects* of the final products. In particular, the part shown in Figure 5.6 exhibits a hole which width is 1.5 mm. The presence of defects is a source of waste by itself; nonetheless, it also implies increased utilization of the resources, since a higher number of operations must be performed to have the target number of parts matching the desired quality.


Fig. 5.6 The toy duck used as a case study, which exhibits a fabrication defect.

The case study is synthesized in Figure 4.1, according to the methodology defined in Chapter 3: the classes of wastes affecting this process are highlighted. The process is schematically described and a synthesis of the mathematical technique is presented.

## 5.4 Monitoring and control system

Due to the lack of sensors discussed in the previous Section, in this study offline laboratory measurements have been performed: the whole profile of the defective part has been measured after the fabrication. The surface has been scanned through a GOM Atos machine; the distance between adjacent measured points is 0.6 mm. The technique for measurement analysis has been implemented in Matlab, and consists in two parts: in the first one, the presence of possible discontinuities in the surface of the part is detected. In the second part, the scanned points grid is compared with the nominal cloud of points to evaluate the adherence to the expected output. In the following, a pseudo-code of the two algorithms is provided (also see Figure 5.8):

- 1. Test for part integrity
  - a. Identify the points without close neighbors: for each scanned point, evaluate whether other surface points were detected in a square centered in the point with edge size equal to 2 mm.
  - b. Identify the boundary points: for each scanned point, evaluate whether there exist some points that can be considered boundaries or not. For each direction, a  $0.5 \times 20$  mm rectangle is considered: if no points are detected into this area, the point is tagged as boundary point.
  - c. Identify the points without close neighbors which are not boundary points, i.e. the points found in step 1a which do not respect the criteria in item 1b.

The points identified at step 1c are tagged as risky points: the lack of close neighbors and the presence of other, far, points with a similar x or z coordinate could be due to a discontinuity in the surface of the object.

2. Alignment to the nominal geometry

	Process	; de	scripti	on
	Suppliers         Raw material         Material properties         Planning         Batch size         Interarrival times & variability         Design         STL File         Part dimensions and positioning         Working parameters         Performance         Finished part         Printing time, work in process, throughput         Failures incidence         Quality         Surface measurements         Adherence to the model			Reusable Additive machine Current state
C O M P O N E N T S			Disposable Energy Current state	
				Reusable Additive machine Current state E S
				Disposable Energy consumption
	Wastes			Data—analusis
Overpi	Overproduction Data sou		Data source	GOM Atos For optical (offline) measurements
Waiting Transport			Data processing	Filter the data related to the measured part
Extra processing		· · · · ·	Feature generation	Distances among adjacents points Deviations from the nominal geometry
Inventory			Feature extraction	Distances and deviations higher than a given threshold
Defec	Defects		Decision making	Stop the printing process (offline simulation)

Fig. 5.7 The application of the methodology defined in Chapter 3 to this case study.

#### 5.5 Results

- a. The CAD file for the nominal geometry is transformed into a cloud of points.
- b. The scanned points are roughly aligned to the nominal geometry: 3 translations are performed to align the two centers of mass, as well as rotations to align the axes directions.
- c. The ICP algorithm (Besl and McKay, 1992) is used to best align the measured points cloud to the nominal model, by minimizing the distances between the two surfaces. The repositioning performed at item 2b is used as initial condition for the ICP, and allows to reduce computational time.
- d. For each scanned point the distances from the nominal surface are measured.

The position of the nozzle during printing is not available; in future developments, the tools introduced in this work should be integrated into a virtual manufacturing environment to validate the nozzle path and the postprocessing operations.

## 5.5 Results

In Figure 5.9 some snapshots from the simulation for the integrity test algorithm are shown. Points on the surface of the duck are plotted in cyan; black points are the boundary points; red dots represent the risky points, i.e. the points which do not have near neighbors, but cannot be considered as boundary points (identified at step 1c). Thus, red points would not exist if the surface of the object does not exhibit discontinuities. In the example, 4612 points are scanned on the surface of the object; among them, 115 are considered risky.

In Figure 5.10 some results for the comparison between real and expected geometries are shown. Beside the hole, in some areas the distance between the fabricated surface and the expected one is greater than 2 mm. Even in this case, the algorithm is able to generate an alarm and the user can decide whether to interrupt the fabrication process or not.

The algorithm has been run on a common laptop with CPU frequency 1.7 GHz: the computational time necessary to analyze the whole dataset (approximately, 4600 points) with the two algorithms is lower than one minute; since the material deposition rate is much slower, the inspection algorithm can be used to real-time detect the presence of issues, given a suitable class of sensors.



Fig. 5.8 Pseudo-code of the algorithms for measurements analysis.



**Fig. 5.9** Results of the simulation for the integrity test algorithm. Points on duck surface of the duck are plotted in cyan; black points are the boundary points; red dots represent the risky points.



Fig. 5.10 Results of the simulation for the algorithm to compare the fabricated surface with the expected geometry.

# 5.6 The role of MES

The technique for surface monitoring can be used as a stand-alone application able to provide alarms in case an issue arises. However, beyond the monitoring and control of the fabrication process, the output of the algorithm can be used with further purposes.

If the monitoring and control system is used offline, like in this study, MES allows to compare the performance of the last production process with the previous ones and detect symptoms of possible criticalities. The information analyzed and stored by the MES represents an instrument to support automatic correction or compensation strategies, as well as to improve the awareness of operators decisions. The availability of data-sources able to perform online measurements would enhance this capability, allowing to adjust the process in real-time and improve the quality of the part undergoing the printing process.

Over longer time-scales, syntheses of the collected data can be performed to highlight the reasons for which criticalities occurred. This capability is very significant in the field of additive manufacturing: currently, the most important challenges for this technology are poor part accuracy and lack of process repeatability (Rao et al., 2015). These issues are due to the complex relationships among variables which are, in many cases, still unknown. Hence, the information provided by the MES allows acquire knowledge about the process and to improve its performance.

## 5.6.1 Integration between MES and DFAM

A broader deployment of MES results can be done if they are integrated with tools for design: DFAM has been introduced in Section 5.2.2.

MES can profitably be supported by the design: the cooperation between the two systems allows MES to continuously compare the "as-is" product and process states to the expected conditions and quickly detect mismatchings. In the presented case study, this task is performed by the algorithm that evaluates the adherence of the measured points cloud to the expected output.

MES can also profitably support design: the analyses performed over longer time-scales, based on real data, the information about criticalities and the in-field acquired knowledge allow to correct and improve the design of a product or its fabrication process (e.g.: change machine parameters, the positioning of the part on the machine tray, change material, choose a different machine, ...). Given this reciprocal interaction, the framework in Figure 5.11 can be defined.

This framework allows to extend the DFAM model proposed by Ponche et al. (2014) shown in Figure 5.5: it consists in three tasks and is able to opti-

#### 5.6 The role of MES



Fig. 5.11 Proposed framework for MES-DFAM integration.

mize the design of a product taking into account the capabilities and the constraints (such as the thermal distribution and the trajectories) of the process that will be used for its fabrication. MES support can be useful to improve the result of all the three tasks performed by the DFAM.

First, part orientation has a strong impact on the quality of the finished part. Decision support tools have been developed to identify the part orientation that results in the best roughness and accuracy of the produced object. However, such tools are based on predictions and simulations of process behavior. A feedback information from a set of sensors able to evaluate the quality of the physical part would be useful to validate the predictions and, in case of mismatching, to correct the orientation of the part into the machine. Also the simulation model can benefit from the feedback mechanism: it can be enriched or adjusted with the empirically acquired knowledge, resulting in a more accurate output; improved models can lead to better predictions and, in turn, to better support the DFAM. Second, the shop-floor information can also be used for further adjustments of the part geometry. In case the quality or the precision of specific features is not satisfactory, the shape of the object and the material distribution can be revised more quickly. The third task included in the DFAM methodology is the optimization of manufacturing paths and machine parameters. A feedback information from the shop-floor is useful because even minute uncontrolled variations can lead to strong differences in the quality of the fabricated part. Several variables can affect the production process, concerning the deposition and the material (melt pool geometry, temperature, deposition height); furthermore, the output quality is also correlated to the state of the deposition chamber, including temperature, humidity or oxygen concentration (Reutzel and Nassar, 2015). All these parameters interact with each other. It is not trivial to ensure the quality of a produced part by controlling only

5 MES for monitoring of product geometry failure

a few variables, but further adjustments to the machine parameters can be necessary according to the real operating conditions. Such adjustments can be taken according to the measurements performed by the sensors, and a MES-DFAM integration allows to quicker take decisions and actions to improve output quality.

This proposal for MES-DFAM integration also allows to extend the model proposed by Rosen (2007) in Figure 5.4: after planning a manufacturing process, simulations are performed to check whether the design process results in producing an object in compliance with the specifications. Such simulations can be supported by the MES feedback, that allows to continuously validate the process and its model and to better take into account process variability. In case an issue arises, alarms may be generated, or strategies for self-adaptation or self-compensation can be undertaken. Furthermore, functionalities to early detect possible decays of the process over a longer time scale can be implemented.

# 5.7 The support to lean manufacturing

The implementation of the techniques shown in this Chapter enables to reduce *Defects* sources of waste. The detection of issues performed in real-time would allow to stop the production of a part as soon as its quality is believed unsatisfactory: hence, the material waste due to finishing a part that will be rejected is avoided. Further, the deployment of a monitoring and control system supported by a MES allows to identify in advance criticalities, and to undertake corrections or compensations to avoid extra processing.

The deployment of a MES in the field of additive technology also allows to improve the economical sustainability of the process. Today, due to the unpredictability of the process, the unitary cost of a part produced through additive manufacturing is higher than producing it by the common technology. Nonetheless, AM is convenient to produce single parts or small lots, because it does not require the initial investments necessary to start the process in traditional manufacturing (e.g. specialized machines, moulds, ...). However, improved knowledge process results in improved predictability and output quality. This, in turn, could change the balance between traditional and additive manufacturing, and make AM economically convenient even for higher batch sizes.

# 5.8 Conclusions

In this Chapter, the importance of monitoring systems in an additive machine has been shown. The results obtained with offline tests are promising,

#### 5.8 Conclusions

and an online implementation would provide useful hints while the production process is still occurring. Nevertheless, the scanning system used in this work cannot be deployed in a production environment, since the ratio between costs and benefits would be excessively high. However, several different optical devices can be used to acquire a 3D cloud of points; there exist reliable, recently developed tools which cost is sufficiently low to allow the deployment of several devices to have a complete scan of the object undergoing fabrication (such as the Asus Xtion). Nonetheless, such devices cannot be integrated into the building chamber of an additive machine; a further miniaturization effort is necessary.

The advantages of a direct communication between MES and DFAM have also been shown: the efficacy of the knowledge extracted by the sensors system can be enhanced. The feedback mechanism allows to reduce the rampup phase of a new product, since machine parameters can be tuned in realtime. Similarly, information concerning process properties, instabilities or criticalities can be collected in a structured way. Hence, the acquired expertise can be used for further product developments, and improved quality of the output can be reached (e.g. lower surface roughness and stair-stepping effect, or improved stress resistance). This, in turn, allows to decrease material and energy waste for unsatisfactory productions, leading to cost reductions and improved sustainability of the process.

This study has been shown in the form of proof-of-concept. The approach is general enough to be deployed with any of the available additive technologies discussed in Section 5.2.1. In this work, the attention was restricted to a specific case study in the field of fused deposition modeling for a sample test; nevertheless, the deployment of a MES could be much more significant in high value-added productions, such as the production of aeronautical or biomechanical components, where the tolerances are very tight and expensive materials are deployed. MES-DFAM integration can also be helpful in testing new materials or alloys: the sensors-based system can collect information about the behavior of the process and the final quality of the product: the acquired data may validate the expected performance, or provide hints for further adjustments or improvements.

Note The results of this research have been published in:

- G. D'Antonio, F. Segonds, J. Sauza Bedolla, P. Chiabert, N. Anwer. A proposal of Manufacturing Execution System integration in Design for Additive Manufacturing. In: Proceedings of The IFIP 5.1 12th International Conference on Product Lifecycle Management (PLM15). Doha, Qatar, October 19-21, 2015
- G. D'Antonio, F. Segonds, F. Laverne, J. Sauza Bedolla, P. Chiabert. A framework for Manufacturing Execution System integration in an advanced Ad-

5 MES for monitoring of product geometry failure

*ditive Manufacturing process*. Submitted to the Journal of Intelligent Manufacturing.

82

# Chapter 6 MES for storing processes and warehouse management

Abstract The overall performance of a production system relies on the performance of the manufacturing operations as well as on the behavior of surrounding activities, such as logistics. To have a just-in-time production, both the raw materials and the finished products must be available at the right time and place, at the minimum possible cost. Several investments have been recently performed to improve warehouses efficiency, for example through the introduction of automation systems. Among the available technologies, systems based on the deployment of Autonomous Vehicle Storage and Retrieval System (AVS/RS) are very promising: they are based on the possibility to use lighter vehicles and to remove constraints among the movements along different axes. Despite the increasing diffusion of such systems, there is a lack in techniques for performance evaluation. Therefore, methods to evaluate and control the exploitation of an AVS/RS system have been studied. Due to the complexity of such systems, analytical techniques can be used for basic, predetermined cycles; a more complete overview can be obtained through the deployment of discrete events simulations, which enable to consider even complex scenarios with low efforts. The developed tools can be integrated into a MES to online evaluate and improve the performance of the system, and collect information for decision making; further, such tools can also support the design phase, to verify in advance the reaction of the system in front of multiple, composite scenarios.

## 6.1 Introduction

In the last decades, beside the performance improvement of manufacturing operations, great efforts have been done in enhancing the design and the management of warehouses. Similarly to manufacturing activities, one way to better deal with global competition is the deployment of automation solutions to successfully face with complex situations and quickly take decisions, as well as to replace repetitive, manual tasks. One innovation that increasingly spread in the last decades is the deployment of automated systems able to store and retrieve items in and from warehouses. The market of automated warehouse is increasing at a high pace, driven by the outstanding growth of e-commerce.

The most common Automated Storage and Retrieval Systems (AS/RS) consist in a set of stacker cranes moving along aisles. In order to move the items in the warehouse, the stacker cranes can perform three types of movements: longitudinal, on a rail along the aisle; vertical, up the column of the stacker crane; transverse, in order to store or retrieve a unit load (UL) into the rack. The cranes are able to move, at the same time, both on the longitudinal and the vertical directions, in order to minimize the traveled distance and the cycle time. Here, the term *cycle time* refers to the time necessary to store and/or retrieve unit loads starting and finishing at the same location, according to a round trip (Bozer and White, 1984). The main drawback of these systems is low flexibility: ULs are processed one by one, hence in case of high variability in the storage or retrieval activities, the reactivity of the system is limited.

Recently, different types of AS/RS able to uncouple the vertical and the longitudinal movements have been developed; in this work, the attention is focused on systems able to separate vertical, the longitudinal and the transverse movements. To perform these movements, the system adopts autonomous vehicles, hence it can be classified as an Autonomous Vehicle Storage and Retrieval System (AVS/RS), as defined by Malmborg (2002).

Two aspects are crucial to best exploit the capabilities of such systems: first, an accurate performance evaluation is necessary; second, appropriate tools must be used for operations management, execution and control. Nonetheless, up to date few work has been done for the evaluation of AVS/RS performance, as revised in Section 6.2. Furthermore, information tools to support warehouses are undergoing a reorganization: the arise of Warehouse Execution Systems is leading to a redefinition of the the tasks to be performed by each information tool and of the information to be exchanged.

Thus, the aim of this work is twofold. First, new analytical techniques are presented to estimate the cycle time of an AVS/RS. Second, a discrete-event simulation tool has been developed to test the impact of different manage-

### 6.2 State of the art

ment criteria and select the best one, according to the specific scenarios that can occur.

The remainder of the Chapter is organized as follows: in Section 6.2 the the state of the art for performance evaluation and information tools to support automated warehouses is reviewed. In Section 6.3 the case study at stake is presented. Then the performance evaluation tools are introduced. First, the analytical technique is shown in Section 6.4; an example of application is shown in Section 6.5. Second, the discrete event simulation tool is presented in Section 6.6 and applications to a case study are shown in Section 6.7. The role of the MES and the support to lean manufacturing are discussed in Sections 6.8 and 6.9. Final conclusions are presented in Section 6.10.

# 6.2 State of the art

# 6.2.1 Techniques for performance evaluation

AVS/RS have been introduced in the late 1990s in European facilities. The first scientific paper on this technology is dated 2002: Malmborg developed an analytical model to evaluate the utilization of the machines, the cycle time and the throughput of the system, based on the topology of the rack and the features of the vehicles. He focused on tier-to-tier configurations, i.e. racks in which vehicles are able to move through different levels using a lift, aiming at compare AS/RS and AVS/RS performances. He also developed (2003) analytical tools to estimate the proportion of dual command cycles (i.e. cycles in which both a storage and a retrieval task are performed) based on the demand of storage and retrieval tasks and the estimated cycle times for single and dual command cycles. Other approaches are based on queuing theory. Kuo et al. (2007) developed a model for estimating the cycle time and the utilization of machines in single command cycles. Zhang et al. (2009) presented a model to deal with non-Poissonian queues keeping analytical simplicity. Other queue approaches have been studied by Cai et al. (2014); Fukunari and Malmborg (2008); Roy et al. (2012). Ekren and Heragu (2009) developed a regression, simulation-based model to tie the average cycle time of the system to its features; in (Ekren et al., 2010), a design of experiment approach has been used to identify the factors affecting the warehouse performance: dwell point policy, scheduling rule, input/output locations and interleaving rule are significant factors.

These papers all consider single-depth rack. The only work that takes into account multi-depth racks is (Manzini et al., 2016): an analytical model is presented to evaluate the performance of a rack with arbitrary rack width.

All these works are based on the assumption that items are randomly stored into the rack, although the allocation criterion has been found to be significant in (Ekren et al., 2010). Furthermore, the developed techniques are mainly devoted for designers, in order to quickly identify the configuration of the rack that meets the end-user requests; no applications of models for performance monitoring during the deployment of the AVS/RS have been found in literature.

# 6.2.2 Information tools for warehouses

The main operations that a warehouse needs to plan and control are: (i) receiving items from a supplier; (ii) storing the items; (iii) receiving orders from customers; (iv) retrieving the requested items; (v) ship the orders to customers. These operations should be performed in the minimum time, at the minimum cost, with high reactivity and customer satisfaction.

To achieve these purposes, the support of information tools is essential. In the past years, two classes of tools have been developed. The first one is named **Warehouse Management System** (WMS): it provides the information necessary to manage warehouse resources and control the flow of products in a warehouse, from receiving to shipping (Forte Industries, 2014; Nynke et al., 2002). WMS are often integrated into ERP systems (Nynke et al., 2002). The second class of software is given by **Warehouse Control Systems** (WCS): they are in charge of the activities execution and the detailed equipment control on the warehouse floor (QC software, 2008).

Given these "traditional roles" for WMS and WCS, a gap exists between the long-term planning performed by the former system and the detailed, short-term control in charge of the latter. To fill this lack, a first approach has been the extension of the work areas of the two systems. Real-time systems for data acquisition have been integrated into WMS: they mainly consist in RFID traceability systems, enabling to facilitate data collection and storage, and to enhance the performance of the warehouse (Chow et al., 2006; Poon et al., 2009; Wang et al., 2010). On the other side, Amato et al. (2005) proposed an approach for short-term optimization, integrated with WCS, of handling sequences, in order to minimize the time to complete a little number of picking or storage missions, based on the current state of the system.

Nonetheless, the extension of tasks in charge for each system did not lead to the expected efficiency improvement. Therefore, in recent years, a further software layer has been developed: the **Warehouse Execution System** (WES). This system supports the communication between the planning and the control levels: it collects order data from software systems to provide a global view of the state of the equipment and optimize the movements of items based on what occurs in the various parts of the warehouse (Forte Industries, 2014). This enables more aware management decisions in terms of resources allocation and system layout to improve reactivity, for example, in front of seasonal peaks of activity: historical data can be analyzed to

#### 6.3 Description of the process

extract order profiles, customers preferences, profitable strategies to better deal with the demand. Functionalities for automated dynamic adjustments of the workflow can also be integrated (Intelligrated, 2013).

Hence, an analogy between the information tools devoted to manufacturing and to warehouses can be made. WMS matches ERP functionalities: it keeps a global view of the process and the stored inventory, and drops down the orders to a lower level. WES is in charge of the purposes typical of a MES: it provides performance metrics for operational and strategic planning; it also supports the validation of the current management strategy and provides hints for improvements based on the evaluated analytics. Finally, the detail level required for WCS is comparable to the one implemented in shop-floor systems.

# 6.3 Description of the process

As stated in the Introduction, AS/RS exhibit low flexibility and their performance is low in case of high variability for storage and retrieval requests. Conversely, in order to approach a just-in-time flow of items, the capability to deal with great variability keeping high system efficiency is mandatory: this is the strength of AVS/RS.

The case study chosen for this research is a system able to separate the movements along the vertical, the longitudinal and the transverse directions by using a set of autonomous vehicles. The rack is made of an arbitrary number of levels; each of them has a cross aisle that goes from one side to the other, to provide access to the channels. Three different kinds of machines are used:

- the lift is in charge of the vertical movement: one or more units are placed on the border of the rack;
- the shuttle performs the movement through the aisle;
- the satellite is the storage/retrieval machine: it moves through the channels, to deposit or pick an item at/from the target position.

Hence, the whole system consists of several vehicles (shuttles and satellites), autonomous and integrated with each other: the shuttle moves back and forth through the aisle transporting the satellite in front of the target channel. The satellite, in turn, transports the UL: it leaves the shuttle and enters the channel, to perform the storage or retrieval task. The shuttles change the operating level through the lift. Furthermore, there exist one or more bays, which are the interfaces of the system with the external world: the bays are the places where the satellites pick the parts to be deposited in the rack and leave the retrieved items. In this study, the bays are supposed to be on the same side of the rack. A graphical representation of the rack and the AVS/RS system is provided in Figure 6.1.



Fig. 6.1 A representation of the rack and the AVS/RS system.

The sequence of the operations to store a UL is the following:

- 1. The UL is carried in the input bay;
- 2. The UL is loaded by a satellite; the satellite joins the shuttle, and they get on the lift;
- 3. The lift moves to the target level, and the shuttle leaves the lift;
- 4. The shuttle moves through the aisle and stops in front of the target channel;
- 5. The satellite leaves the shuttle and enters the channel;
- 6. The satellite moves along the channel towards the last pallet stored;
- 7. The satellite unloads the pallet at the last empty location, according to a LIFO (Last In First Out) policy;
- 8. The satellite moves back through the channel and joins the shuttle.

The retrieval task is performed symmetrically. The capability of separating movements enables to perform, at the same time, different storage or retrieval tasks. For example, after operation 3 the lift is free to serve another shuttle that needs to move along the vertical direction. The system is able to reconfigure itself to face complex, variable scenarios.

However, the traditional methodologies to evaluate AS/RS performance do not take into account AVS/RS peculiarities, and the techniques shown in Section 6.2 mainly take into account single-depth systems. Conversely, in this case study, the length of the channel can be arbitrary. Further, criteria for items allocation are not taken into account into the analyses. Nonetheless, they are strategic factors for warehouse performance. Inappropriate management of the system can lead to avoidable motion of items and of resources. This, in turn, can reduce the reactivity of the warehouse, since requests for parts to enter or leave the warehouse need extended time to be satisfied, leading to *waiting* and *inventory*. For these reasons, in the following Sections a new analytical tool for performance evaluation is presented. A set of relationships to evaluate cycle times will be presented. 6.4 Analytical evaluation of warehouse performance

Since the term *cycle* implies that the initial and the final configuration of the system must be the same, in case of storage tasks, after operation 8 the above list is also performed backwards to carry each machine at the initial position. In literature, evaluations are mainly performed for *single-command* and *dual-command* cycles: these two expressions refer to the number of items transported during the cycle. In the former case, one storage or one retrieval task are performed; in the latter, both the tasks are executed. Nevertheless, in the system at stake the lift is able to transport up to two items at the same time. Hence, a broader combination of cycles is feasible: several levels can be visited during the same cycle, as well as more than two items can be involved in the movements on the same level. Hence, the model that will be introduced in the next Section concerns with general *multi-command* cycles; they will be classified as *homogeneous* (in case only one level is visited) or *heterogeneous* (in case more levels are visited).

The schematic in Figure 6.2 synthesizes how the methodology defined in Chapter 3 has been used to deal with this case study.

# 6.4 Analytical evaluation of warehouse performance

In this Section, an analytical technique, based on a probabilistic approach for the evaluation of cycle time is presented: it allows to calculate an average performance of the system and to compare it with some benchmarking situations. The evaluation is based on:

- the topology of the rack;
- the performances of the machines;
- the number of items involved and their task (storage or retrieval);
- the criteria for items allocation, in terms of statistical distributions.

## Rack topology

The rack is supposed to be symmetric in each direction: all the aisles have the same number of channels, and each channel has the same number of storage positions. The (x,y,z) coordinates for each storage position in the rack must be known. For sake of simplicity, the lift is supposed to be in the position (0,0,0); this assumption does not reduce the generality of the model. The following parameters are defined:

- $N_x$ : the number of channels for each aisle
- $N_{\nu}$ : the number of levels
- *N<sub>z</sub>*: the number of storage positions for each channel.

The number of lifts, shuttles and satellites are provided, as well as the capability of the shuttles to travel without a satellite. These parameters are

	Process	des	cripti	on
CO	Suppliers Storage & Retrieval orders Item type and lot Planning Batch size Interarrival times & variability		Reusable Vehicles Current state: position, condition, battery level Disposable Enerau	
M P O	Design Rack topology Vehicles properties Storage and vehicles management criteria			Current state S O U
N E N T S	Performance Stored & Retrieved items Cycle time, queue content, throughput Machines utilization Idle times Failures incidence Quality Correct positioning (with respect to the provided criteria)		Reusable Vehicles Current state	
				Disposable Energy consumption
	Wastes			Data-analysis
Overpi	roduction	D S	ata ource	ERP. Design. PLC
Waitir Transp	ig 🗸	Di Pi	ata rocessing	Analytical models Discrete Event Simulation
Extra processing		F 9	eature eneration	Cycle time Detailed log of resources activities
Inventory		Fi e	eature xtraction	Comparison with benchmarking conditions Throughput: Resources utilization
Defect	ts	D m	ecision naking	Change management criteria Change system composition

Fig. 6.2 The schematic of the methodology used to develop this work.

6.4 Analytical evaluation of warehouse performance

necessary to determine which model, among the ones shown below, is to be deployed.

## **Machines performance**

The acceleration and the steady-state speed of the machines are provided, both in the loaded and in the unloaded conditions. Given this class of input and the former one, the time necessary to reach each rack position from the bay can be evaluated:

- $t(x_i)$ ,  $i = 1, ..., N_x$ : the time necessary to move from the lift to the *i*-th channel
- t(y<sub>j</sub>), j = 1,..., N<sub>y</sub>: the time necessary to move from the bay to the *j*-th level
- $t(z_k)$ ,  $k = 1, ..., N_z$ : the time necessary to move from the channel entry to the *k*-th position in the channel

## Number of items and tasks

. The cycle to be examined is described through the following parameters:

- *I*: the number of items involved in the cycle;
- *T*: the number of transitions from a storage to a retrieval task on the same level. It denotes the number of times in which the shuttle moves between two channels without passing through the lift;
- *S*: the number of times in which parallel activities of the satellite and the shuttle take place;
- *P*: the number of different levels visited in the cycle.

The value of *S* is tied to *I* and *T* through the following relationship:

$$S = I - (1+T),$$

since for the first item and for transitions from storage to retrieval a shuttlesatellite uncoupling would not make sense.

## Criteria for items allocation

For each level, channel, and position into the channel, the probability of interaction with the transport machines is evaluated through the following probability distributions:

- $\mathbf{a} = \{a_i\} = \{P(x = x_i)\}, i = 1, ..., N_x$  describes the movements along the *x*-axis;
- $\mathbf{b} = \{b_j\} = \{P(y = y_j)\}, j = 1, ..., N_y$  describes the movements along the *y*-axis;

6 MES for storing processes and warehouse management

-  $\mathbf{c} = \{c_k\} = \{P(z = z_k)\}, k = 1, ..., N_z$  describes the movements along the *z*-axis.

### Variables

Given the time necessary to reach each rack position and the probability distributions defined above, the following average times can be determined:

$$x_{M} = E[t(x)] = \sum_{i=1}^{N_{x}} a_{i}t(x_{i})$$

$$y_{M} = E[t(y)] = \sum_{j=1}^{N_{y}} b_{j}t(y_{j})$$

$$z_{M} = E[t(z)] = \sum_{k=1}^{N_{z}} c_{j}t(z_{k})$$
(6.1)

Hence,  $x_M$  is the average time spent to move along the aisle from the lift;  $y_M$  is the average time spent moving in the vertical direction from the bay;  $z_M$  is average time necessary to move along the channel from its entrance.

Further, a shuttle may also move between two different channels on the same level (e.g. when it is switching from a storage to a retrieval operation): the duration of this travel is denoted by  $\delta x$ , and is evaluated through the probability distribution **a**. Similarly, the lift may need to move between different levels; the duration of this travel is denoted by  $\delta y$ .

$$\delta x = E[t(\delta x)] = \sum_{i=1}^{N_x} a_i |t(x_i) - x_M|$$

$$\delta y = E[t(\delta y)] = \sum_{j=1}^{N_y} b_j |t(y_j) - y_M|$$
(6.2)

This quantity is not evaluated for the z direction, since there is no physical meaning for a satellite to interact with two positions in the same channel.

An additional variable must be evaluated to describe the case in which a shuttle can move even without a satellite: while the satellite is moving into the channel, the shuttle can go to the lift to pick the next UL to be stored and then move back to the channel to join the satellite. Alternatively, this situation can occur when two consecutive retrievals have to be performed: while the shuttle is leading the first UL to the lift, the satellite moves in the channel to pick the second item. Hence, the time that the two machines spend being uncoupled is given by the maximum duration among the two activities; it can be described through the following relation:

92

6.4 Analytical evaluation of warehouse performance

$$S_{xz} = \sum_{i=1}^{N_x} \sum_{k=1}^{N_z} a_i c_k \max\{x_i, z_k\}$$
(6.3)

Two classes of data sources are necessary to provide the input for data analysis. Design tools provide data concerning the topology of the rack and the performances of the machines. The data source for allocation criteria is the PLC of the system: it contains detailed data about the movements of the machines and the positions in the rack that have been deployed.

Given the variables defined above, the analytical models for the cycle time can be formulated. Different models, mutually exclusive, are formulated for systems consisting in one or more shuttles: this classification is necessary because in the former case the lift is idle while the shuttle is working along the aisle; conversely, in the latter case, the lift may serve one shuttle while another one is working along an aisle. The system chosen as case study can comprehend an arbitrary number of machines; nevertheless, the most common configuration consists in one lift and one or two shuttles. Thus, the models shown below are focused on these configurations. Furthermore, a distinction is made between homogeneous and heterogeneous cycles.

## 6.4.1 Models for systems with one shuttle

The models described in this Section can be used to evaluate the performance of systems consisting in one lift, one shuttle and one satellite.

First, models for **homogeneous cycles** are considered. In case the uncoupling of the satellite from the shuttle is not feasible (or it is supposed to be not convenient), the cycle time is given by:

$$CT = 2y_m + [2Iz_m + (2I - 2T)x_m + T\delta x].$$
(6.4)

The cycles that such a system is able to perform are: (i) 1 storage; (ii) 1 retrieval; (ii) 1 storage and 1 retrieval. In Appendix A, the full description of these cycles and a comparison with this analytical model is provided.

The cycle time for systems in which uncouplings of the satellite from the shuttle can occur is given by:

$$CT = 2y_m + |2(I-S)z_m + 2x_m + (I-1)\delta x + S \cdot S_{xz}|.$$
(6.5)

Two examples are provided in Appendix A to explain the model: (i) 2 storages (or 2 retrievals); (ii) 2 storages and 1 retrieval (or vice-versa). In case a lower number of items is involved, uncouplings do not take place and the model in Equation 6.4 can be used; the case in which 2 storages and 2 retrievals are performed is already described by the cycle in which

1 storage and 1 retrieval occur (except for the lift time). Cycles involving a higher number of items are not feasible, since the lift – usually – is not able to move more than 2 items at the same time.

In the models above, square brackets have been used to divide lift times from shuttle and satellite times. This distinction is helpful to introduce the models for **heterogeneous cycles**: they are a generalization of homogeneous cycles. The lift time is increased by the quantity  $\delta y$  multiplied by the number of times a switch between levels occurs. The subscript *p* is introduced to denote the events occurring at the level *p* (since different activities can take place at different levels). Thus, the heterogeneous model for systems without shuttle–satellite uncouplings is:

$$CT = 2y_m + (P-1)\delta y + \sum_{p=1}^{P} \left[ 2I_p z_m + \left( 2I_p - 2T_p \right) x_m + T_p \delta x \right].$$
(6.6)

The heterogeneous model for systems in which shuttle–satellite uncouplings occur is:

$$CT = 2y_m + (P-1)\delta y + \sum_{p=1}^{P} \left[ 2(I_p - S_p)z_m + 2x_m + (I_p - 1)\delta x + S_p \cdot S_{xz} \right]$$
(6.7)

## 6.4.2 Models for systems with two shuttles

The developed approach can be extended to systems consisting in more than a single shuttle. To fully exploit the system, while one shuttle is working on one level, the lift is free to serve the other shuttle. Hence, parallel activities occur and the working pace of the system is determined by the bottleneck machine.

The underlying assumption of this model is that both the shuttles are supposed to perform the same kind of activity on the level. Nonetheless, the rack positions involved in the cycle are different. At the beginning and at the end of the cycle, the lift is supposed to be in the bay; the shuttles are supposed to be at different levels.

Before calculating the cycle time of the system, the bottleneck must be identified. Thus, the following quantities must be evaluated:

- *T*<sub>*level*</sub>: the time spent by the shuttle and the satellite to perform their task on a level; it is given by the terms within square brackets in Equations 6.4 or 6.5;
- $T_{lift}$ : it is the time needed by the lift to serve the other shuttle, i.e. picking it to the level, going together the bay and then back to the level, and finally reach the level of the first shuttle. It is given by  $T_{lift} = 2y + 2\delta y$ .

6.5 Results of the analytical model

In case  $T_{level} < T_{lift}$ , the lift acts as a bottleneck; the cycle time is given by

$$CT = 10y + 3\delta y \tag{6.8}$$

Conversely, in case  $T_{level} > T_{lift}$ , the shuttle is the bottleneck and the cycle time is:

$$CT = T_{level} + 8y + \delta y \tag{6.9}$$

A graphical representation for these relationships is provided in Appendix A.

A synthetic chart to summarize the presented models and to drive the choice of the model that fits with the system to be evaluated is provided in Table 6.1.

**Table 6.1** Synthesis of the models for analytical evaluation of automated warehouses performance.

Nr. of shuttles	Type of cycle	Uncontraints	Model
	Homogonoous	No	Eq. 6.4
1	Homogeneous	Shuttle – Satellite	Eq. 6.5
1	Hotorogonoous	No	Eq. 6.6
	Theterogeneous	Shuttle – Satellite	Eq. 6.7
2	Lift is the bottleneck	Lift – Shuttle	Eq. 6.8
Ζ	Shuttle is the bottleneck	Lift – Shuttle	Eq. 6.9

# 6.5 Results of the analytical model

The analytical tool for performance evaluation provides an estimation of the cycle time for predetermined cycles based on the real allocation criteria. The purpose of these tools is to evaluate whether the potential of the system is correctly exploited or not: the estimated cycle time can be compared with a set of benchmarking conditions to evaluate whether the overall performance of the system can be enhanced by changing the allocation criteria. Namely, the benchmarking conditions are:

- Best case: the machines always interact with the channel closest to the bay. It represents the best case of resources management (from the perspective of the cycle time) and leads the minimum possible cycle time;
- Worst case: the machines always interact with the channel farthest from the bay. It is the case of worst resources management and leads the maximum possible cycle time;
- Practical worst case: it is an intermediate case between the two previous ones; it represents the case of maximum randomness, thus is evaluated by supposing a totally random allocation criteria (i.e. the probabilities are given by uniform distributions).

The definitions for the three cases are adapted from (Hopp and Spearman, 2011). The former two cases define the interval of all the possible values that can be obtained by the analytical models; however, the last value is the most interesting, since it allows to state whether the management criteria is better than a random choice or not. Values higher than the practical worst case are due to issues in the allocation criteria: items frequently exchanged are stored too far from the bay, leading to time waste; an intervention should be undertaken to change the allocation criteria and improve the overall performance of the system.

As example, a warehouse with  $N_x = 15$ ,  $N_y = 4$  and  $N_z = 5$  can be considered; the distance between adjacent levels, channels and storage positions is equal to 2 m. The storage and retrieval system consists of one lift and two shuttles; the shuttles are not capable to uncouple their movement from the satellites. The performances of the machines are listed in Table 6.2: the acceleration varies according to the presence of a UL above the machine, while the steady-state speed does not change.

Table 6.2 Performances of the machines involved in the case study

Speed [m/s]	Acceleration [m/s <sup>2</sup> ]			
opeca [m/ 5]	Loaded	Unloaded		
2.00	0.50	0.60		
0.23	0.30	0.40		
1.20	0.50	0.70		
	Speed [m/s] 2.00 0.23 1.20	Speed [m/s]         Accelera Loaded           2.00         0.50           0.23         0.30           1.20         0.50		

At the beginning of the observation, the fill rate of the warehouse is approximately the 50%. During the observation, 300 items are stored and 284 are retrieved. The empirical distributions of the rack positions for storage and retrieval activities during this period have been evaluated; they are synthesized in Table 6.3.

Such empirical distributions are first used to evaluate  $T_{level}$  and  $T_{lift}$ , and decide which is the model appropriate to describe the system. Since  $T_{level}$  = 54.1 s and  $T_{lift}$  = 34.2 s, the model in Equation 6.9 has to be used.

The prevalent class of cycles observed consists in one storage and one retrieval on the same level. According to the model, the average cycle time necessary to perform this kind of cycles is 152.4 s. This value has been compared with the three benchmarking values. The results are shown in Table 6.4: since the cycle time provided by the model is higher than the practical worst case, the capability of the rack is not well exploited because the system tends to frequently use rack positions far from the bay. For example, positions far from the bay are used along the *x* direction: the 11% of the interactions involve the farthest channel, while there are almost no interactions with channels in the middle of the aisle. This result is obtained because items frequently exchanged are put far from the lift, thus leading to time waste: this means that the allocation criteria should be reviewed.

6.6 Deployment of Discrete Event Simulation

<i>x</i> direction		<i>y</i> directio	n	z direction		
Coordinate [m]	Weight	Coordinate [m]	Weight	Coordinate [m]	Weight	
2.00	0.03	2.00	0.82	2.00	0.21	
4.00	0.23	4.00	0.09	4.00	0.12	
6.00	0.15	6.00	0.02	6.00	0.24	
8.00	0.08	8.00	0.07	8.00	0.23	
10.00	0.10			10.00	0.20	
12.00	0.01					
14.00	0.01					
16.00	0.01					
18.00	0.20					
20.00	0.00					
22.00	0.02					
24.00	0.05					
26.00	0.00					
28.00	0.00					
30.00	0.11					

**Table 6.3** Statistical distributions for the interaction between satellites and rack positions,evaluated by observing 300 storages and 284 retrievals.

 Table 6.4 Results of the performance benchmarking performed through the analytical model.

CT Value [s]
125.73
141.81
152.41
361.44

The analytical tool presented in this Section represents a significant progress with respect to the state of the art: the approaches in literature mainly focus on single or double deep racks; the only work dealing with deeper racks is (Manzini et al., 2016), but this evaluation is not based on the criteria for UL allocation. The tool can be easily implemented (even on a common spread-sheet) and used.

Nonetheless, the analytical model provides average indicators, without information about the punctual behavior of the system; furthermore, the analysis has been extended to the allocation criteria, but it does not take into account the criteria used to manage the machines. Hence, a more detailed representation can be necessary to have a complete picture of the system. It can be obtained through a virtual representation; for this reason, the simulation tool shown in the next Section has been developed.

# 6.6 Deployment of Discrete Event Simulation

As stated, the alternative approach to evaluate the performance of the system is the deployment of a numerical simulation. Due to the physical characteristics of the system, a Discrete Event Simulation (DES) approach was chosen. In the main commercial DES software (e.g. FlexSim, Tecnomatix Plant Simulation, Arena) a net classification of the objects in the model is performed to distinguish servers and items. However, in this case study, there exist objects that behave both like servers and items (e.g. the shuttle is transported by the lift to move along the *y* direction, and transports the satellite along the *x* direction).

Thus, in order to be as free as possible and to skip such implementation issues, the simulation was coded from the scratch in a Matlab environment. The key steps for a **storage task** are synthesized here:

- A source introduces ULs into the system, according to a probability distribution or to a delivery table. Each UL is given a unique ID, an ItemType and an ItemLot code;
- A storage position in the rack is chosen. Different criteria can be chosen, as described below;
- The machines necessary for the storage task are identified.
  - If the resources are not free, the task is delayed;
  - If the selected machines are not already in the bay, their travel to the bay is planned with priority;
- The UL is picked from the bay and brought to the target position through the following steps:
  - The satellite, the shuttle and the lift are in the bay;
  - The satellite leaves the shuttle to pick the UL;
  - The satellite goes back above the shuttle;
  - The lift transports the shuttle to the target level;
  - The shuttle with the satellite and the UL above leaves the lift, moves through the aisle and stops in front of the target channel;
  - The satellite leaves the shuttle, moves through the channel and deposits the UL at the target position.
  - The satellite moves back through the channel to join the shuttle.

The **retrieval tasks** are managed in a similar way. Retrievals are created according to probability distributions: an ItemType and an ItemLot are requested; then the UL to be retrieved is identified through the selection criteria described below.

### Criteria to select the storage position

The rack consists in a set of UL positions which may have different properties; for example, different levels may exhibit different sizes in order to host different kinds of ULs. The following, mutually alternative, criteria have been implemented to select the target channel (TC). The provided labels are used to quickly identify the criteria in Table 6.6.

- 6.6 Deployment of Discrete Event Simulation
- Select the channel already containing the same ItemType and ItemLot, at the level closest to the bay, closest to the lift; labeled ClosestLevelLot.
- 2. Select the channel already containing the same ItemType and ItemLot, closest to the lift, at the level closest to the bay; labeled ClosestChannelLot
- 3. TC is on the level as close as possible to the bay: the TC is the channel containing the same ItemType and ItemLot, with an available position, closest to the lift; if no channel are found, the empty channel closest to the lift is chosen; otherwise, the selection is performed on the second level closest to the bay, and so on; labeled ClosestLevel.

The first two criteria differ in the search hierarchy. In the first one, available positions are searched through all the channels on the first level; if no one contains similar ULs, the search is extended to the second level, and so on. In the second criterion, the search is started by considering, for each level, the channel closest to the lift; if no channels fit the request, the second channel of each level is considered, and so on.

## Criteria to select the item to be retrieved

When the request for an item is generated an ItemType and an ItemLot are generated; The following, mutually alternative, criteria have been implemented to select the target UL (TUL):

- Select the reachable UL with the requested ItemType and ItemLot at the level closest to the bay, in the channel closest to the lift; labeled ClosestLevel.
- Select the reachable UL with the requested ItemType and ItemLot at the channel closest to the lift, at the level closest to the bay; labeled ClosestChannel.
- Select the oldest, reachable UL, without regard for the position in the rack; labeled Oldest.

### Criteria for machines management

In case the system consists in more than a shuttle, the one to be used for the storage task is chosen according to this priority:

- Choose the shuttle already going towards the bay (if there is one);
- Choose the shuttle already at the target level or moving towards it (if there is one), in order to avoid having more than one shuttle per aisle;
- Choose the idle shuttle closest to the bay.

To take into account for lift-shuttle uncouplings, while a shuttle is running along an aisle, the lift is set in an idle state; hence, it can be used if another shuttle needs to move through the levels. The set of features provided as output by the simulation model is much broader than the analytical model: it consists in a report of all the activities performed by the machines, in a chronology of the events experiences by each UL, and in a picture of the current state of the rack. These data can be further processed to be transformed into more useful features, such as average values and variabilities for the transport times, the waiting times, and the utilization of the resources. The major advantage of the simulation is the possibility to explore the behavior of the system even in composite scenarios.

## 6.7 Results of the simulation model

The simulation tool presented in the previous Section can be used to compare the performances of different management criteria in dealing with a given scenario, and to select the one leading to the best results.

To proof this capability, the results of a set of simulations are shown in this Section. The criteria for the selection of UL position shown above have been compared by simulating two scenarios. The former represents a steady-state situation: storage and retrieval activities are almost balanced. The latter is a stressful condition: 50 items to be stored are put into the bay at the same time, and must be put into the rack as soon as possible to make free the queue.

The topology of the rack is the same introduced in Section 6.5. The parameters describing the AVS/RS are shown in Table 6.5; the shuttles are not capable to uncouple their movement from the satellites. The performances of the machines are listed in Table 6.2.

Table 6.5 Synthesis of the parameters describing the AVS/RS used as case study.

Quantity	Value
N <sub>x</sub>	15
$N_{y}$	4
Ňz	5
Spacing [m]	2
Lifts	1
Shuttles	2
Satellites	2

At the beginning of the simulation, the lift is in the bay; the two shuttles are idle at the first level (y = 2) and at the third level (y = 6). The fill rate of the rack is 50%: the stored units belong to 8 different lots. The queue of ULs waiting to enter the rack is treated according to a FIFO (First In First Out) policy.

100

6.7 Results of the simulation model

Cimulation Nr	Rack criteria				
Simulation INF.	Storage	Retrieval			
1	Closest Level Lot	Closest Level			
2	Closest Channel Lot	Closest Channel			
3	Closest Level	Closest Level			
4	Closest Level Lot	Oldest			
5	Closest Channel Lot	Oldest			
6	Closest Level	Oldest			

**Table 6.6** Criteria used in the simulations to identify the best UL position for storage and retrieval in the rack.

The ULs to be stored are randomly generated: the time lapsing between the creation of two consecutive items is provided by a normal distribution with average value equal to 150 s and standard deviation equal to 10 s. The ItemLot corresponding to each UL is generated according a uniform distribution. To keep the system balanced, the ULs to be retrieved are requested, on average, once on every 150 s, but the variability is higher: the standard deviation is equal to 20 s. The ItemLots are still chosen according to a uniform distribution.

A graphical representation for the content of the rack at the beginning of the simulation is provided in Appendix A.

The storage and retrieval criteria used in each simulation are listed in Table 6.6. A synthesis of the obtained results is shown in Table 6.7. For storage activities, queue times are the times in which the entering UL is waiting for a machine in the bay: the minimum time is 2 s, corresponding to the time necessary for the satellite to leave the shuttle and reach the UL. Travel times are measured from the instant in which the item is loaded from the bay and the instant in which it is stored in the rack. Travel and queue times are defined in a specular way for retrieval tasks. These data are not comparable with the cycle times obtained through the analytical model: here, after the storage of a UL, the shuttle remains idle in its position and does not go back to the bay to close the cycle.

The results listed in Table 6.7 show that allocation criteria that less use the lift provide a better performance in terms of average queue and travel times, and related variability. However, it must be highlighted that this result strongly depends on the ratio between shuttle and lift steady-state speeds, which is approximately 8:1. The criterion Closest Channel Lot (simulation 2) – which is the one that most uses the lift – leads to the highest queue times and the highest time variability. The criterion Closest Level shows better reactivity, since the queue and the travel times exhibit the minimum values among the first three criteria combinations. Nonetheless, in these cases the retrieved UL is chosen only according to its position. In case the rack is handling perishable products, the retrieved UL should be the one that has been in the rack for the longest time, among the ones matching the requested ItemLot. Hence, simulations 4-6 have been run to evaluate

		Simulation Nr.					
		1	2	3	4	5	6
	Lift	28.9%	40.2%	14.7%	32.5%	39.8%	30.2%
N 1 1	Shuttle 1	28.4%	31.5%	26.3%	32.8%	34.9%	37.0%
Machines	Shuttle 2	20.0%	21.9%	6.7%	22.9%	23.9%	17.3%
utilization	Satellite 1	42.3%	44.9%	48.2%	46.2%	48.0%	54.2%
	Satellite 2	29.3%	31.6%	10.9%	31.9%	33.1%	24.9%
	Average	20.6	24.0	11.8	28.5	26.4	26.8
Queue time in	Std. Dev.	25.0	26.9	15.4	32.2	30.6	30.6
the bay [s]	Max	110.8	105.5	70.2	135.5	121.3	127.2
	Min	2.0	2.0	2.0	2.0	2.0	2.0
	Average	36.4	39.3	30.5	37.9	40.8	36.1
Travel time to the	Std. Dev.	9.0	10.9	6.2	10.2	11.3	9.4
rack [s]	Max	63.1	89.2	47.4	81.3	72.3	66.0
	Min	17.4	17.4	17.4	17.4	17.4	17.4
Potrioval augus	Average	33.7	37.3	20.4	38.8	41.2	38.9
time in the reals	Std. Dev.	21.0	21.6	13.4	22.1	22.4	23.0
	Max	108.5	111.9	81.7	130.3	137.6	216.8
[8]	Min	3.5	3.5	3.5	3.5	3.5	3.5
	Average	34.7	38.8	29.6	37.4	39.2	34.1
Travel time to the	Std. Dev.	8.1	11.2	5.8	10.3	10.2	9.1
bay [s]	Max	63.1	105.5	43.4	79.6	60.1	69.0
	Min	17.4	17.4	17.4	17.4	17.4	17.4

Table 6.7 The results of the simulations performed, according to the criteria in Table 6.6.

the performance of the Oldest retrieval criterion. The utilization of the machines, of course increases, due to the longer travels necessary to retrieve ULs. This, in turn, affects queues and travel times. The storage time also increases, since the rack positions made free by retrievals are no more the closest to the bay. Given the performances of the machines and the topology of the rack, the best storing criterion is Closest Level: it results in the lowest machines utilization to perform the given task. This, in turn, leads to lowest time spent traveling and to a lower energy need.

In the second tested scenario, 50 items are put into the bay at the same time and have to be stored as quick as possible. This case mimics the situation in which an external supplier deposits an amount of material. The ItemLot values are generated according to a uniform distribution; the initial state of the rack is the same described for the previous case.

The results of these simulations are shown in Table 6.8. The storage criteria are the same described for simulations 1-3. In this case, lift travels should be reduced as much as possible. The configuration of simulation 2 leads to the highest amount of time necessary to perform the whole storage operation; the other two criteria require comparable amounts of time, but the utilization of the machines exhibit huge differences. Hence, the criteria to be deployed can also be chosen according to the current energy level of machines batteries.

#### 6.8 The role of MES

			Simulation	Nr.
		7	8	9
	Lift	83.6%	89.4%	50.3%
NC 1.	Shuttle 1	46.9%	44.4%	60.7%
Machines	Shuttle 2	55.4%	51.5%	23.9%
utilization	Satellite 1	61.7%	55.6%	84.9%
	Satellite 2	68.2%	62.7%	29.5%
	Average	35.4	39.0	31.6
Travel time to the	Std. Dev.	9.2	9.8	6.0
rack [s]	Max	58.1	58.1	45.0
	Min	17.4	17.4	17.4
	Total time	2701.6	3296.0	2723.5

**Table 6.8** Results of the performed simulations in case 50 ULs need to be stored simultaneously.

# 6.8 The role of MES

As stated in Chapter 2, performance monitoring is one of the core tasks for a MES. One way to monitor the behavior of the warehouse is the extraction of data from the PLC of the AVS/RS: data concerning the movements performed by each vehicle, their utilization and the time necessary for the performed operations can be collected and analyzed. Nevertheless, the mere extraction of average cycle-times can be an information not satisfactory. Shopfloor data allow a MES to perform deeper analyses. The analytical models shown in Section 6.4 can be integrated into the MES to compare the real behavior of the system with a set of benchmarking situations: this allows to evaluate whether the performance of the system can be improved or not, and how much effort should be done.

A more complete instrument is represented by the simulation tool. As shown, it provides more detailed information about each activity of the system. Average times, measures for variability and queues can be extracted through post-processing analyses. Furthermore, punctual information about the state of the rack can be extracted. The integration of this simulation tool with a MES establishes an extraordinary tool for activities planning and for the prevention of criticalities. The management level provides the MES with a bill of items to be stored or retrieved. A simulation for such a bill can be run to test the criteria for allocating items to rack positions and identify in advance possible issues. For example, in Section 6.7 the results provided by different allocation criteria in front of the same scenario have been compared. Since the run-time of this algorithm is a few minutes, it can be used to select in real-time the criterion that best fits with the current necessities to plan the storage and retrieval activity. Hence, the criteria to manage the AVS/RS system can vary according to the particular condition of the rack and to the specific situation in which the storage and retrieval task has to be performed.

The deployment of the simulation tool also supports the prevention of criticalities. The final content of the rack can be virtually observed; in case it is not satisfactory, an intervention is possible before starting the storage and retrieval tasks. For example, some ULs can be reallocated to reduce the operational time, or to make free some strategic positions for the new items to be stored. Furthermore, the integration of simulation into MES also allows to better deal with variability: in case a huge quantity of items must be rapidly put into the rack, the MES can manage in advance the content of the rack by making free the positions closest to the bay, in order to quickly conclude the storing activities and successfully deal with activities peaks. Similarly, in case broad quantities of items need to be retrieved, the MES can move such items as close as possible to the bay during the idle times, in order to improve the reactivity of the system.

Over longer time-scales, the MES can collect data concerning the average staytime into the rack for each item type. Hence, it can manage the allocation criteria by placing the most exchanged item types closest to the bay, in order to minimize the storage and retrieval time, thus reducing transport.

## 6.9 The support to lean manufacturing

The two tools that have been developed in this work allow to enrich the functionalities of a MES in the field of warehouses. This also allows to better deal with the lean manufacturing approach. The first source of waste that can be faced and reduced through these tools is *transport*: a tight monitoring and control on the allocation activity is certainly strategic to decrease useless transportation of the items. Transport reduction implies an average reduction of the cycle time, leading to improved reactivity of the system. An improved control of the allocation criteria also enables *motion* reduction: storage strategies appropriate to the current conditions of the system allow to decrease the useless motion of vehicles. Motion reduction is particularly significant: the energy necessary to move shuttles and satellites is provided by batteries; when a machine gets out of energy, it is recovered for a given time. Hence, motion reduction fosters increased availability for the machines.

Furthermore, the flexibility in redesigning functional criteria enabled by the MES can also be extended to the hierarchy of the operations to be performed. The system chosen as case study performs activities in the order they are introduced into the system. For example, consider the case in which two storing activities are planned and a retrieval operation is required while the first one is running. The system will finish the first storage activity; the machines will go unloaded to the bay; the second storing activity will be performed and, finally, the retrieval operation will be examined. The flexibility provided by the MES can support a redefinition of the hierarchies, allowing to perform – if convenient – the retrieval activity between the two

#### 6.10 Conclusions

storage operations; this improved planning of activities also contributes in reducing non value-added motion.

The improved flexibility and capability in dealing with variability provided by the MES also allow to reduce *waiting* and *inventory*: queues of items to be stored and temporarily stocked into the bay can be reduced, as well as the time interval necessary to lead an item in the bay after its request.

## 6.10 Conclusions

In this Chapter, heterogeneous techniques to evaluate the performance of an automated warehouse based on AVS/RS technology have been provided. The approaches currently available in literature are not satisfactory because mainly consider single- or double-depth racks. Conversely, the system studied in this Chapter consists in multi-depth racks. Furthermore, usually the allocation criteria is not considered to evaluate the performance of the system; nevertheless, it plays a key role in stating whether the warehouse is well exploited or not.

The importance of integrating the developed techniques into a MES (or a WES) has been discussed, as well as its role in supporting the achievement of lean manufacturing. Beside online monitoring and control, the presented techniques can also be used to support the design phase of a new system or to evaluate interventions to improve the performance of an existing AVS/RS. The analytical technique provides a preliminary estimation of the expected performance: practitioners can use it to evaluate whether the configuration of the designed system (i.e. the number of machines and their performances) is able to satisfy the needs of the customer in dealing with standard cycles. On the other hand, the DES tool is based on a virtual copy of the designed system: hence, the AVS/RS can be validated before the physical realization of the warehouse, by testing its reaction in front of the most different scenarios.

**Note** This work has been supported by Regione Piemonte under the research feasibility study MOMA, "Analisi delle prestazioni dei sistemi di MOvimentazione per Magazzini Automatici", in cooperation with the company Eurofork.

# Chapter 7 MES for the management of automated vehicles traffic

Abstract Beside storage, another activity surrounding the manufacturing operations is the transport of items through the workstations in the shopfloor. Usually, in mass-production, workstations are placed in a serial layout, and items move through consecutive workstations by conveyors. This is an efficient solution, but the resulting flexibility of the production line is very low and even a small failure in the transportation means may result in blocking the whole line. Recently, the spread of Automated Guided Vehicles (AGVs) increased: this solution allows to improve the flexibility of the line. However, switching from a traditional conveyor to a fleet of AGVs, keeping the serial layout for the workstations, is not economically convenient; a cellular layout would enable to better exploit the advantages provided by the new transportation system. Hence, a novel methodology to identify the best cellular layout for a manufacturing process has been studied. A mathematical model has been developed for re-dimensioning the workstations and to identify their best position in the plant, in order to minimize the path traveled by each item. Then, a technique based on Discrete Events Simulation has been used to evaluate the behavior and the performance of the transport system, in order to monitor and control the behavior of the whole manufacturing process in the new layout, and predict possible criticalities. This tool can be used to online manage the process as well as to a-priori test different scenarios and support the decisions taken by the designers.

# 7.1 Introduction

In the case studies previously discussed, the methodology presented in Chapter 3 has been deployed to deal with already existing processes: waste classes affecting the process were identified; the exhaustive process description allowed to identify the possible sources of such wastes; then, techniques to be integrated into MES were developed to reduce wastes and improve process performance. The aim of this Chapter is slightly different: here, the purpose of the research is to provide a support to the innovation process by working hand-by-hand with the design, in order to reduce expected wastes since the very early design stages.

A case study has been provided by a system integrator settled in the Turin area. It consists in a line for the assembly of engine cylinder heads: this line is usually designed in a serial layout, and traditional conveyors are used to perform the transport of items between subsequent workstations. This is an efficient solution, but the resulting flexibility of the production line is very low and even a small failure, either in a machine or in the transportation means, may result in blocking the whole process. Hence, the company is considering the opportunity to switch from the traditional transport system to automated vehicles. Two ways to exploit this change exist. The first one is a mere replacement of the conveyors with a set of Automated Guided Vehicles (AGVs); however, due to the higher cost of the fleet of vehicles with respect to the conveyors, this strategy is not economically sustainable. The second way is a complete change in the design paradigm, enabling to fully exploit the potential benefits deriving from the deployment of flexible transportation means.

The efforts performed in this work mainly involve two areas. The first one is the layout design of the manufacturing line: the freedom provided by the AGVs allows to remove transport constraints and opens the possibility to switch from a serial workstation positioning to a cellular layout. The flexibility of the process can be dramatically improved; nonetheless, a careful analysis must be performed to minimize the impact of sources of waste. The second work area is the definition of the tasks to be performed by the automated vehicles: they can be used for the mere transport of items through workstations or, alternatively, they can take in charge an item since it is introduced into the line and accompany it through the whole process.

The remainder of the Chapter is organized as follows. In Section 7.2 the state of the art in the field of cellular manufacturing design is depicted. In Section 7.3 the process selected as case study is introduced. In Section 7.4 an innovative technique to optimize the position of workstations in a cellular layout is presented; the results are shown in Section 7.5. Then, the issue of AGVs tasks is dealt: in Section 7.6 a simulation tool is presented to compare different scenarios; results are provided in Section 7.7. Beside process design, the developed tools can have further purposes: in Section 7.8, the
### 7.2 State of the art

possible deployment of such tools to manage the process after its implementation is discussed; their support to lean manufacturing is presented in Section 7.9.

# 7.2 State of the art

For long time, in traditional layout configuration the issue of handling material has not been considered, since it was a manual task; alternatively, standard automation solutions were deployed. With the development of new manufacturing systems and the spread of AGVs, the flow path layout gained importance for the efficacy of the whole manufacturing process.

The first aim of this work is the optimization of a cellular layout: cellular manufacturing is a consequence of group technology, in which a manufacturing system is partitioned into several independent systems with enhanced flexibility. According to Dimopoulos and Zalzala (2000), this problem consists in three sub-problems: (i) the definition of the manufacturing cells, i.e. the machines and the operations that constitute a cell must be identified; (ii) the layout of the cells in the plant; (iii) the layout of the machines within the cells. The techniques developed in this work address the second task: the first one is currently performed by the industrial partner, according to technological constraints. The third one, at the moment, is not dealt, since cells are made of identical machines and there is not a transport of items within cells; further, the cells considered in this work only consist of one or two machines.

Different approaches are available in literature. Given the complexity of the problem, often the three tasks are not performed at the same time, thus multi-step techniques have been developed. A first attempt has been made by Gupta et al. (1996): a genetic algorithm was proposed to address the machine cell-part grouping problem. Three different objective functions were proposed with different purposes: (i) minimize the total inter- and intra-cell moves; (ii) minimize cell load variation; (iii) minimize simultaneously both the former functions. Wu et al. (2007) developed a hierarchical genetic algorithm (HGA) to simultaneously form manufacturing cells and determine the group layout of a Cellular Manufacturing System (CMS). Chang et al. (2013) developed a two stage model: first, they solve cell formation and cell layout; then, machine layout in each cell is evaluated. A linearized, constrained objective function is minimized. A similar approach has been used by Chan et al. (2008). Javadi et al. (2013) enriched the intra-cell optimization by taking into account unequal dimension of machines, machines orientation and the exact material handling costs. Kia et al. (2014) aimed at minimizing the total cost of material handling between and within cells by taking into account machine relocation, the purchase of new machines, machine overhead and processing in a multi-floor layout. Hu et al. (2006) focused on the inter-cell layout and material handling system: they aimed to minimize the inter-cells path through genetic algorithms to reduce the cost of material handling. Tavakkoli-Moghaddam et al. (2007) developed a model to simultaneously minimize the total costs of inter- and intra-cell movements, also taking into account a stochastic demand. Hafezalkotob et al. (2015) used the fuzzy goal programming to jointly minimize four independent objective functions concerning: (i) the inter-cellular traveled distance; (ii) the economic investment on machines; (iii) the workload balance of cells; (iv) the capacity of the cells.

Other approaches have been inspired by physical or natural systems. Ariafar and Ismail (2009) aimed at arranging facilities in a cellular system to minimize handling material cost. They used a simulated annealing technique, which is based on physical annealing: it is a heat treatment process that gradually cools a physical system to reach the state of minimum potential energy. Jolai et al. (2012) developed an electromagnetism-like model in which each point is considered as a charged particle. Charged particles interact and influence with each other through attraction and repulsion; electromagnetism-like algorithms are designed to optimize non-linear, realvalued problems. Soto et al. (2015) used an artificial fish swarm algorithm, belonging to a class of techniques able to solve complex optimization problems. The aim of their work is the minimization of movement and material exchange between cells, to optimize time and costs.

# 7.3 Description of the process

The process considered in this study consists in a line devoted to the assembly of engine cylinder heads. This line is made of the 19 operations listed in Table 7.1. In the Table, a short description of these operations is provided, together with the corresponding process and setup times. Automatic, semi-automatic and manual operations are performed; each of them is supposed to take place into a different workstation. The transport of items through the process is supported by conveyors joining the workstations in a serial layout. When switching to the AGV fleet, the design must be optimized to minimize *transport* and *motion*. The schematic in Figure 7.1 synthesizes how the methodology defined in Chapter 3 has been used to develop the research work in this case study.

### 7.4 Technique to optimize a cellular layout

As stated in Section 7.1, the aim of this Chapter is to develop tools able to optimize the position of the workstations into a cellular layout. Currently, the approach to design a serial layout is mainly based on the experience of

	Process	des	scripti	on
	Suppliers Engine blocks Item type and lot Planning Batch size = 1			Reusable Workstations Operators AGVs Current state of the vehicles: position, condition, battery level
C O M P O N E N T S	Interarrival times & variability Design Shop–Floor topology Vehicles properties Bill o Fprocess	r U T	Lu	Disposable Energy prorefrigerants & compressed air Tools Current state
	Performance Finished items Cycle time, queue content, throughput Machines utilization Idle times Failures incidence	O U T	Reusable Workstations Operators AGVs Current state of the vehicles: position, condition, battery level	
	Quality Correct assembly of components	P U T		Disposable Worn out tools Energy consumption
	Wastes	Τ		Data—analysis
Overpi	roduction	D S	ata ource	Bill of process Bill of orders
Waiting Transport		 D P	ata rocessing	Layout: Agent–based model Performance: Discrete Event Simulation
Extra	processing	F g	eature eneration	Layout: Workstations positions Performance: Detailed resources activities
Invent Motio	ory	F	eature xtraction	Layout: AGVs path length Performance: Performance indicators
Defects		··· D	ecision naking	Layout: Approve/Change the new layout Performance: Change management criteric

Fig. 7.1 The schematic of the methodology used to develop this work.

ID Cluster		Description	Avg. process	Avg. setup
	Cluster	Description	time [s]	time [s]
1	Load	Load cylinder head to pallet	10	8
2	Sealant and lubrica-	Lubricate valve guide bores or	15	8
	tion	valves		
3	Insertion	Install intake and exhaust	25	4
		valves		
4	Leak test	Valve blow-by leak test	25	8
5	Rollover	Turnover 180 degrees	10	8
6	Load	Load camshafts to pallet	15	8
7	Load	Load camshaft caps and bolts	20	8
		to pallet		
8	Insertion	Assemble valve stem seal	25	4
9	Press	Press valve stem seals	15	8
10	Insertion	Assemble valve springs, valve	25	4
		spring retain		
11	Press	Key-up	15	8
12	Sealant and lubrica-	Apply sealant	25	4
	tion			
13	Load	Assemble camshafts, camshaft	10	4
		caps, bolts and pre-torque		
14	Tightening	Torque camshaft cap bolts	20	8
15	Measure	Torque to turn	25	8
16	Press	Press camshaft seal ring	15	8
17	Tightening	Torque, intake, exhaust and/or	25	8
		injector studs		
18	Marking	Cylinder head label	15	8
19	Load	Unload cylinder head assem-	10	8
		bly		

 Table 7.1 Bill of operations of the current process for cylinder head assembly.

the designer: approaches such as Knowledge Based Engineering (KBE) are at study, aiming to identify tools capable to formalize this knowledge and make it available for all the designers working into a company. Nonetheless, KBE can provide hints concerning good practices, particular requests or habits of a particular customer. When switching from a serial to a cellular layout, optimization tools are necessary to design a process involving the lowest number of components and able to provide the best performance.

In this Section, an algorithm to evaluate the best workstations position in the shop-floor is presented. It is based on the analogy with a mechanical system consisting in a set of bodies connected with each other through springs. Each spring has given stiffness and equilibrium length. The equilibrium configuration of this system results in the lowest residual energy. The mechanical system is transposed to the manufacturing layout as follows. The workstations act as the connected bodies; the springs represent the travels to be performed according to the bill of process: the stiffness of the springs is proportional to the number of travels performed through each

112

#### 7.4 Technique to optimize a cellular layout

couple of cells. In this way, each working cell is considered as an agent that interacts with other agents, which position is due to a balance of forces.

Hence, a lattice-free agent-based model to simulate the interactions among cells and determine the equilibrium configuration has been formulated. The morphology of the cells is not explicitly introduced: in the model, agents are described as circular objects: this simplification allows to model interactions as acting only on the radial direction, rather than the *x* and *y* coordinates. Each cell is given two characteristic radii:  $R^{in}$  is the radius of the area physically occupied by the workstation;  $R^{out}$  depicts the area necessary to move around the workstation (for example, to permit AGVs transit or to supply components to the machines).

In Figure 7.2, an example of the interacting forces is shown: the black area is the circle with radius  $R^{in}$ ; the lilac area is the circle with radius  $R^{out}$ . Two kinds of forces act between the cells *i* and *j*: the attraction force tends to lead the length of the spring towards the equilibrium length (i.e. the distance at which the two external circles are tangential with each other); the repulsion force avoids overlapping between adjacent stations. Such interactions involve the centers of the two cells. The following variables can be introduced:

$$\mathbf{d}_{ij} = \mathbf{x}_i - \mathbf{x}_j$$

$$\ell_{ij}^{eq} = R_i^{out} + R_j^{out}$$
(7.1)

The two forces are modeled through a unique spring with variable stiffness: when the length of the spring is higher than the equilibrium length, the stiffness is equal to the number of travels performed through the stations i and j to produce one unit, denoted by  $T_{ij}$ ; hence, the highest is the number of travels, the highest is  $K_{ij}$ , the lowest is the final distance between the two cells. The overlapping between cells is discouraged through a quadratic relationship, as shown in Equation 7.2.

$$K_{ij} = T_{ij}(|\mathbf{d}_{ij}| > \ell_{ij}^{eq}) + 10 T_{ij}^{2}(|\mathbf{d}_{ij}| \le \ell_{ij}^{eq})$$

$$\mathbf{F}_{ij} = K_{ij}\left(|\mathbf{d}_{ij}| - \ell_{ij}^{eq}\right) \cdot \frac{\mathbf{d}_{ij}}{|\mathbf{d}_{ij}|}$$
(7.2)

The warehouse is modeled as a punctual entity, i.e. items to be manufactured and finished items enter and leave the line from the same point. It is described as an additional cell interacting with the other agents in the model according to the relationships above. Further, repulsion forces are added to avoid cells going outside the plant area:

$$\mathbf{d}_{i,wall} = \mathbf{x}_{i} - \mathbf{x}_{wall}$$

$$\mathbf{F}_{i,wall} = 10 \left( \left| \mathbf{d}_{i,wall} \right| - R_{i}^{out} \right) \frac{\mathbf{d}_{i,wall}}{\left| \mathbf{d}_{i,wall} \right|} \cdot \left( \left| \mathbf{d}_{i,wall} \right| \le R_{i}^{out} \right)$$

$$(7.3)$$



Fig. 7.2 Example of the interaction forces involving the cell *i*.

A balance of forces is performed for each of the *N* agents in the model. The inertialess assumption can be made, since the cells can be freely placed and moved into the shop-floor. At the equilibrium configuration, for each cell  $\sum \mathbf{F} = 0$ . Nevertheless, the balance of forces described above allows to evaluate the distances among cells, but not their position into the shop-floor. Hence, an artificial movement due to drag forces is introduced:

$$\mathbf{F}_{i}^{drag} = -\nu \dot{\mathbf{x}}_{i} \tag{7.4}$$

The parameter  $\nu$  can be set equal to 1. In this way, the balance of forces is given by the following equation:

$$\dot{\mathbf{x}}_{i} = \sum_{\substack{j=1\\j\neq i}}^{N} \mathbf{F}_{ij} + \mathbf{F}_{i,wall}, \quad i = 1, \dots, N$$
(7.5)

This expression must be solved for all the agents that can be freely moved along the plant area; conversely, a given position can be set for the agents that cannot be moved (e.g. the warehouse, or cells with localization constraints). 7.5 Results of the layout optimization

Cell ID	Cluster	Operations	Avg. process and setup time [s]	Inlet flows	Nr. of machines	Utilization
1	Load A	1; 19	18.00	2	1	0.80
2	Sealant and Lu-	2; 12	26.00	2	2	0.58
	brication					
3	Insertion	3; 8; 10	29.00	3	2	0.97
4	Leak test	4	33.00	1	1	0.73
5	Rollover	5	18.00	1	1	0.40
6	Load B	6; 7; 13	21.67	3	2	0.72
7	Press	9; 11; 16	23.00	3	2	0.77
8	Tightening	14; 17	30.50	2	2	0.68
9	Measure	15	33.00	1	1	0.73
10	Marking	18	23.00	1	1	0.51

 Table 7.2 Description of the working clusters composing the cellular layout to be optimized.

# 7.5 Results of the layout optimization

The algorithm shown in the previous Section has been tested to design in a cellular layout a line able to perform the bill of operations presented in Table 7.1.

The ultimate goal of the cellular layout is to have cells composed of flexible machines able to perform operations similar with each other, rather than having heavily focused machines. Hence, the listed operations have been clustered in order to identify cells performing similar tasks: the column "cluster" has been used to group similar operations. The only exception is given by the *Load* cluster: the first and the last operations must be performed close to the warehouse, due to the weight of the part to be loaded; conversely, in operations 6, 7, and 13 lighter parts are managed. Further, since load operations are spread along the sequence, a unique cell placed close to the warehouse would generate a huge quantity of traffic and crossings between vehicles. For this reason, two *Load* cells are considered. In total, 10 working cells have been identified.

The number of machines composing each cell has been evaluated, keeping lower than one the overall utilization of the workstations. To perform this analysis, the current takt time equal to 45 seconds has been kept. The results of this analysis are synthesized in Table 7.2: the 10 cells consist in one or two machines. In the traditional layout, 19 machines are necessary to perform the whole sequence. The first result of the layout reorganization is a reduction of this number: 15 machines are sufficient to keep the process stable.

Given this clusterization, the values for the matrix T, containing the number of travels between each couple of cells can be evaluated. The direction of the travel is not taken into account: the scope of this step is to identify the movement aisles and the intensity of the traffic. The matrix T for this bill of

processes is provided below; the first row and the first column are referred to the warehouse. For sake of readability, non-zero values are highlighted with bold fonts.

				V	Vork	stat	ion						
	WH	1	2	3	4	5	6	7	8	9	10		
	0	1	0	0	0	0	0	0	0	0	0	WH	
	1	0	1	0	0	0	0	0	0	0	1	1	
	0	1	0	1	0	0	1	1	0	0	0	2	
	0	0	1	0	1	0	1	3	0	0	0	3	Ц
т_	0	0	0	1	0	1	0	0	0	0	0	4	atic
1 =	0	0	0	0	1	0	1	0	0	0	0	5	kst
	0	0	1	1	0	1	0	0	1	0	0	6	/or]
	0	0	1	3	0	0	0	0	1	1	0	7	S
	0	0	0	0	0	0	1	1	0	1	1	8	
	0	0	0	0	0	0	0	1	1	0	0	9	
	0	1	0	0	0	0	0	0	1	0	0	10	

Then, the dimension of each cell must be evaluated. Each machine has length equal to 4 m and width equal to 1.8 m. The distance between machines in the same workstation is 2.2 m. Hence, the radii of the area occupied by the machines are equal to 2.7 m for single-machine cells and 3.8 m for double-machine cells. The annulus surrounding the cell to support the movement of vehicles is 5 m thick.

The plant is supposed to be square; the length of the edge is 100 m. For sake of simplicity, the coordinates of the point representing the warehouse are (0,0): this point is not moved during the optimization of the layout; conversely, all the working cells can be freely placed in the available area, without further constraints.

The initial position of the agents is evaluated adding cells one-by-one in the area. First, the cell with the highest number of interactions with the warehouse is introduced (in this case, cell number 1). The other cells are introduced one by one in the model: at each step, the cell with the highest number of interactions with the agents already placed in the area is introduced in an iterative way.

Finally, the model can be run. An explicit numerical method has been used to solve the set of Equations 7.5. The model is iterated until the steady state is reached, i.e. the equilibrium configuration is achieved. The equilibrium configuration for the bill of processes used as case study is shown in Figure 7.3. Black circles represent the area occupied by the machines; lilac circles depict the annuli to support vehicles movement. The blue lines represent the traffic directions; the ticker is the line, the more is the traffic on the aisle. Critical crossings involve the aisles 2-6 and 7-3, and the aisles 8-10 and 7-9. In Table 7.3 the final positions for the cells are provided. The algorithm also evaluates the distance to be run by an AGV to transport an item through the process: the estimation is 226 m.



Fig. 7.3 The equilibrium configuration of the workstations for the sequence of operation chosen as case study.

 Table 7.3 Equilibrium positions for the working cells.

```
Cell ID x [m] y [m]
        2.74 1.02
  1
       10.81 -5.62
  2
  3
       22.52 -8.12
       34.06 -9.76
  4
  5
       39.84 -1.12
       28.38 2.33
  6
  7
       16.69 4.77
  8
       24.01 14.37
  9
       14.50 17.85
  10
       8.49 10.04
```

# 7.6 Discrete Event Simulation for shop-floor traffic management

The first step of this project was the definition of the manufacturing cells and the identification of their position in order to minimize the *transport* of each item through the process. The second research direction, presented in this Section, is the comparison among different scenarios to evaluate the best way to exploit the AGVs. Two paradigms are compared. In the first, the vehicles are only used to transport items through the workstations; in the second, the vehicles assist an item through the whole process, since it enters the process until it leaves the line, staying idle when the item undergoes manufacturing operations. The choice among these two scenarios, performed during the design stage, can affect the flexibility of the line and its capability in dealing with takt times different with respect to the initial target. A Discrete Event Simulation approach has been developed to mimic the behavior of the resources involved in the manufacturing process. The simulation has been implemented in FlexSim. However, this is not a restriction: the simulation can be run in any DES environment (such as Tecnomatix Plant Simulation, Arena, AutoMod,...). The input of the simulation model are:

- 1. The position of the working cells, i.e. the output of the model presented in Section 7.4.
- 2. The number of machines composing each cell.
- 3. The setup and processing times for each operation. Each machine has to perform different operations (corresponding to different setup and processing times), and the correct one must be performed at the proper stage. Hence, a numeric label is defined for each item; the initial value is equal to one, and it is increased by one unit after performing each operation; for example, an item with label equal to 10 has to experience the operation number 10.
- 4. The number of AGVs and their performances.
- 5. The task sequence for each AGV (i.e. the sequence of movements to be performed).
- 6. The inter-arrival time and variability of the parts entering the process.

A snapshot of the simulation environment is provided in Figure 7.4. The number of machines composing each workstation is graphically represented by the size of the station. The task executers (representing the AGVs) are initially placed in correspondence of the source of items; the dispatcher is in charge of allocating tasks within the vehicles. Different output can be extracted: in the following, performance indicators concerning the cycle time and the work in process are evaluated, the utilization of the machines and the activities performed by the AGVs.



Fig. 7.4 A screenshot of the FlexSim model implemented to evaluate the performance of the AGVs fleet.

118

Nr. of AGVs	Nr. of items	Takt time [s]	Efficiency
10	1599	62.54	0.72
11	1723	58.04	0.78
12	1822	54.88	0.82
13	1894	52.80	0.85
14	1933	51.73	0.87
15	1983	50.43	0.89
16	2042	48.97	0.92
17	2104	47.53	0.95
18	2169	46.10	0.98
19	2204	45.37	0.99
20	2205	45.35	0.99
21	2205	45.35	0.99
22	2204	45.37	0.99
23	2204	45.37	0.99
24	2204	45.37	0.99
25	2204	45.37	0.99
26	2204	45.37	0.99

**Table 7.4** Synthesis of the performance for a system in which AGVs are free to assist any item.

# 7.7 Results of the simulation

In the first simulated scenario, the vehicles are supposed to be used only to transport items through the workstations. The processing and setup times provided in input to the model are generated according to a uniform distribution, with the mean values in Table 7.1 and interval width equal to 2 seconds. The AGVs have steady state speed equal to 1.2 m/s and acceleration equal to  $0.4 \text{ m/s}^2$ . The interarrival time is deterministic, equal to 45 s, corresponding to the desired takt time. Simulations have been performed with different sizes for the AGVs fleet. In Table 7.4, a synthesis of the performance measures is provided: the optimal number of vehicles active in the shop-floor is 19. A quantity of AGVs lower than 19 results in efficiency (i.e. the ratio between the effective and the desired cycle time) loss, thus in higher values for the takt time and a lower demand can be satisfied. Conversely, a number of AGVs greater than this value does not lead to efficiency benefits and, as shown in Figure 7.5, idle times (hence, wastes) increase.

It must be underlined that in these simulations the energy level of the batteries has not yet been considered; hence, the fleet size discussed in the following is referred to the number of vehicles able to perform transport.

Given this active fleet size, on average, the time spent by each vehicle is composed as follows:

- the 66% is spent in idle condition, i.e. the AGVs is waiting for an item that needs to be transported;
- the 29% is spent transporting items;

 the remaining 5% is due to empty transport, i.e. transport through different positions performed without items above the vehicle.

The average cycle time is 811 s; the time due to processing and setup is equal to 465 s, hence the 42% of the cycle time is spent in transport and queue. The path traveled by each item is 255 m; a 10% difference results from the layout model, due to the increased detail level provided by DES in describing movements (e.g. AGVs rotations, alignment of machines,...).



Fig. 7.5 Distribution of average time spent by each AGV in different activities for the first scenario.

In the second simulated scenario, the vehicles have been supposed to assist the same item since it enters the line until it is finished and leaves the process, in a kind of kanban system. Beside the deployment modality, this scenario implies a different AGV technology. In this case, when the item enters the machine, it is not unloaded from the AGV. Thus, on one side, the AGV must sustain the working operation; on the other side, it must have a sufficiently accurate system for positioning into the workstation. A synthesis of performance measures is provided in Table 7.5. The minimum number of AGVs to reach an efficiency equal to 98% - a value considered satisfactory for the system integrator - is equal to 18. The distribution of AGV times is shown in Figure 7.6. The ratio between different classes does not change as the number of vehicles varies, since times are determined by the events experienced by the assisted item. For a fleet size equal to 18, the distribution of times for each task is similar to the previous kind of system: 30% is spent in transport loaded; 5% is spent in empty transport; 65% is waiting. However, a further distinction can be made for the last class. In case of higher fleet sizes, the quantity of AGVs exceeding the optimal value is not used. In this kind of system, the 57% of the time is spent in processing, hence the AGV stands in each workstation for a given amount of time known a-priori.

120

### 7.7 Results of the simulation

Nr. of AGV	Nr. of items	Takt time [s]	Efficiency
10	1298	77.04	0.58
11	1432	69.83	0.64
12	1554	64.35	0.70
13	1647	60.72	0.74
14	1759	56.85	0.79
15	1866	53.59	0.84
16	1971	50.74	0.89
17	2075	48.19	0.93
18	2174	46.00	0.98
19	2259	45.00	1.00
20	2261	45.00	1.00
21	2261	45.00	1.00
22	2261	45.00	1.00
23	2261	45.00	1.00
24	2261	45.00	1.00
25	2261	45.00	1.00
26	2261	45.00	1.00

**Table 7.5** Synthesis of the performance for a system in which AGVs are constrained to assist a given item.



Fig. 7.6 Distribution of average time spent by each AGV in different activities for the second scenario.

The average result, for this fleet size, is similar to the previous system; however, the punctual behavior is different. In the previous system, the vehicle was free to assist different items, hence the time spent in each work-station was not fixed. Here, the waiting time is set (small fluctuations may occur due to process variability); from the AGV perspective, this is a *wait-ing* waste, since it cannot perform any other activity. A way to better spend this time would be the integration of a charging station into (or close to) the workstations, in order to recharge the batteries, providing the vehicle with the energy to perform a few of the following operations. This would

avoid long recovery times requiring the AGV to stay out of the process. This capability can be a useful advantage for the manufacturer.

Nonetheless, the first system seems to be more flexible in case of variable demand. To avoid *overproduction*, the takt time of the line should be adapted to the demand. The AGVs deployment paradigm should be chosen also depending on the resulting flexibility of the system and the required versatility. According to the plot in Figure 7.7, in case of lower demand, the first scenario requires a lower number of active vehicles to achieve a given takt time.

Conversely, lower takt time values can be necessary to satisfy demand peaks. The plot in Figure 7.8 has been obtained by supposing a 10% increase of the demand; thus, the takt time is set equal to 40.5*s*. In both the cases, the minimum number of AGVs to reach a 98% performance efficacy is equal to 21, hence both the scenarios lead to satisfactory performance. In case higher efficacy is necessary, the second deployment scenario is preferable.



Fig. 7.7 Comparison between the takt times reachable with the two simulated scenarios.

# 7.8 The role of MES

The simulation tool presented and used in the previous Sections has been deployed to support the definition of a manufacturing process: it allowed to evaluate different scenarios and drive the technological choices of the design department. Nonetheless, the same tool can be used with further aims. First, MES is in charge of planning the production process. Thus the simulation tool can be used to validate a given bill of orders: the bill can be provided in input to the model, which encloses a virtual copy of the shop-



Fig. 7.8 Comparison between the takt times reachable with the two simulated scenarios.

floor. The simulation provides an evaluation of process performance and highlights possible criticalities, such as traffic density or crossings. This output can be used as a feedback for the planning functionality: the production plan can be adjusted before it is introduced, in order to minimize issues due to critical situations. For example, the inter-release time of the items can be manipulated, or signals can be sent to the shuttles to change their speed or run through different paths. This capability may lead to few advantages in case of single-product lines, like the one studied in this Chapter; nonetheless, it can be much more interesting in case different kinds of products are manufactured on the same line. The contemporary coexistence of different bills of process can be a source of issues: in this case, beside the inter-release time, a further degree of freedom to be used for planning purposes is the order at which items are released.

From the bottom-up data-flow perspective, the MES can be enriched with functionalities to take into account, beside the state of the workstations, the condition of the vehicles performing the transport. Two aspects are critical for their performance.

The first one is energy: an accurate evaluation of the current battery level and the amount of energy necessary to perform the following cycle is necessary, to avoid vehicles failing in the middle of a transport. Nonetheless, the energy issue cannot be considered only from the single vehicle perspective. A more general view is necessary: a minimum amount of vehicles must be available to prevent process performance degradation. To avoid getting below this threshold, some vehicles may be required to recharge earlier than due. Hence, the MES managing the shop-floor must also be responsible for the recharge cycle optimization, keeping the average energy level of the vehicles above a threshold. The second critical aspect is traffic. Crossings should be avoided as much as possible, since they are a possible risk, beside a source of time waste. First, controls must be implemented in the critical points, i.e. the crossings between corridors. For example, areas that can be occupied at most by one vehicle can be identified. Further, tight precedence rules must be defined to solve criticalities, such as the "keep the right" criterion, or a grid of nodes can be used to define a set of paths alternative with each other. The capability of the MES in changing the inter-release times and the speed of the vehicles can support the solution to such criticalities.

### 7.9 The support to lean manufacturing

The case study presented in this Chapter is a complex process, made of 19 operations, designed to assembly automotive engines, hence high quality parts must be manufactured with tight tolerances. Different sources of waste can affect the performance of the process. Nonetheless, the system integrator is already able to provide a good control system for the workstations, leading to a negligible quantity of defective items. At the current state, the machines exhibit very limited process time variability, thus waiting and inventory through the line are minimized. The main source of waste consists in transport and motion: the study presented in this Chapter aims at minimizing the efforts spent in these two *muda*.

*Transport* mainly depends on the layout of the line. Reductions in transport cannot be performed during the production, and are hardly difficult to be made onto an already existing line: they require a complete reorganization of the layout, thus huge investments. Hence, the layout of the manufacturing line must be optimized a-priori, in the design stage. For this reason, the algorithm shown in Section 7.4 has been developed. The technique has been used to design a single-product line, but is general enough to deal with multi-product lines. In the latter case, the matrix of travels is given by the sum of the matrices corresponding to each product, weighted by the planned production mix.

Conversely, *motion* is tied to the management of the vehicles. In the performed simulations, when an item needs to be transported the closest AGV is chosen, in order to reduce motion. Nonetheless, even in this case a global overview is necessary. The "item" perspective allows to minimize the motion related to the specific workpiece; nonetheless, a solution optimal for a single item can lead to worse performance for other parts. Hence, the MES has to perform a global optimization, considering the whole process and taking into account the short-term performance as well as the medium and long term ones.

### 7.10 Conclusions

In this Chapter, different techniques to support the (re-)design of a manufacturing line have been provided. The case study consists in a line for automotive engine assembly. The currently deployed conveyor system provides rigid constraints: it enables items to perform only a given path through the workstations, and its speed is tuned to achieve the target takt time. The system cannot efficiently deal with neither higher nor lower demand; in the latter case, the only solution is to switch off the line for a given period.

To improve the flexibility of the line, the deployment of a fleet of AGVs in place of the conveyors is at study: nevertheless, this is not a mere change in the transportation system, but a radical innovation in the production paradigm.

The first research effort has been performed in the evaluation of the optimal cellular layout for this manufacturing process. A topological simplification has been made since circular cells have been considered. In further research, the real shape and size of the cells should be considered. Further, the supply of components to the workstations has not been addressed and should be introduced in further work: it can be modeled as a set of further springs linking the cells with the warehouse. Another simplification was made for the warehouse: it was considered as a punctual entity; a more reliable representation can be obtained by modeling it as a set of multiple points, each of them responsible for supplying a subset of cells.

The second research theme was the implementation of a discrete event simulation model to evaluate the events taking place in the shop-floor. Further functionalities should be implemented in the model. First of all, a library of functions to take into account the energy level of the vehicles batteries is necessary, to best plan strategies and dispatch – even in real-time – the shop-floor activities. Further, more complex rules should be implemented to manage precedences and avoid collisions.

**Note** The work presented in this Chapter has been developed in cooperation with dr. Andrea Ascheri, Apprentice Ph.D. Candidate in Production Systems and Industrial Design in COMAU.

# Chapter 8 Conclusions

Abstract In the previous Chapters, four case studies have been presented to support the research hypotheses and provide evidence that the work methodology is general enough to be used in different manufacturing fields. Nonetheless, the research cannot be considered as concluded. In this Chapter, hints for future developments arisen during this work are presented, and an overall picture for the next research towards the factory of the future are discussed.

### 8.1 Introduction

The aim of this research work was the definition of a methodology to enable the development of MES-oriented tools and support the path towards lean manufacturing. A methodology was defined before starting the applicative work. Four case studies were presented to validate the methodology and provide evidence that it is general enough to be used in different manufacturing fields, including machining, warehousing, and logistics. This, in turn, enabled to investigate whether MES is a tool useful to support the implementation of lean practices as well as to evaluate in which fields it is useful or not. In the following, a summary of the case studies developed in the previous Chapters is provided. In Section 8.3, conclusive remarks are discussed. Nonetheless, the present research cannot be considered as concluded: several hints for future developments arose during this work; they are discussed in Section 8.4; further suggestions for future vision are provided in Section 8.5.

## 8.2 Summary of the work

### Traditional manufacturing: automatic workpiece positioning

In Chapter 4, a monitoring and control system developed to automatically perform the positioning of a spur gear into a finishing machine has been presented. In this process, aeronautic components are produced, hence tight tolerances are requested. The quality of the manufactured parts must be very high; consequences of poor parts may be dramatic.

In the process used as case study, the workpiece was manually centered into the machine. This led to a high rate of *defects*, since the operators were not able to perform an optimal positioning. The replacement of this procedure with an automatic monitoring and control system allowed to improve the precision and the reproducibility of gear positioning into the machine, leading to higher overall quality and reduced reworking operations. MES allows to analyze data over the medium-long term, to ensure process stability and to identify further sources of issues and wastes. Moreover, the automation of this task also enables to reduce *waiting* and *inventory* resulting from the variability of manual operations. The benefits of cooperation between MES and PLM have also been discussed. Their integration allows to dispatch shop-floor information towards design tools: this feedback information mechanism can support continuous improvement practices, enhancing the performance of the production process and the quality of the manufactured parts.

### 8.2 Summary of the work

### Additive manufacturing: surface quality monitoring

In Chapter 5, a technique to monitor the surface quality of a part produced through the additive technology has been shown. The process at stake exhibited *defects*, hence a mathematical technique has been developed to provide an alarm as soon as a criticality arises. This study was presented in the form of proof of concept since, at the state of the art, there is no availability for smart sensors capable to be integrated into the building chamber and monitor the process without interfering with the machine. Nonetheless, laboratory tests and simulations have been performed and the capabilities of both detecting surface defects and evaluating adherence to the expected output have been successfully validated. Also in this case, the integration of monitoring techniques into MES enables the early identification of issues and criticalities. Moreover, MES provides support to DFAM: the performed analyses can result in new knowledge for further improvement of both the design of a product and its production process.

### Warehousing: management of resources and items

In Chapter 6, techniques to evaluate the performance of warehouses equipped with AVS/RS have been developed. The analytical tools existing at the state of the art exhibit a non-negligible drawback: they do not consider the criterion used to store and retrieve items from the rack, although this is a significant factor affecting the performance of the warehouse. Hence, a mathematical technique, based on probabilistic reasonings, has been developed to enrich the analysis by taking into account this factor. This model provides information about an average behavior of the system. However, more detailed information may be necessary, both from the points of view of the stored unit or the resources involved in the system. Therefore, a simulation tool able to evaluate the reaction of the system in front of a multitude of scenarios has also been developed: it is capable of testing different management criteria and to choose the one resulting in the best performance.

The deployment of these tools allows to reduce the impact of useless *transport* of items and avoidable *motion* of the vehicles; their integration into a MES allows to perform short, medium and long term optimizations, based on the past experience as well as on the tasks planned to be performed in the future. This, in turn, enables improved reactivity of the system leading to lower *waiting* times for items needing to enter or leave the warehouse and lower *inventory*.

#### Logistics: transport of items through workstations

In Chapter 7, the case study of a manufacturing process to be redesigned was presented. It was provided by a system integrator that designs and produces manufacturing lines. The industrial partner is willing to introduce a radical innovation into its lines: it aims at eliminating traditional conveyors and perform the transport of items through the line by AGVs. Nonetheless, this switch is not trivial, and deep analyses must be performed; hence, tools have been developed for the best integration of AGVs into the manufacturing process and minimize *motion* and *transport*.

First, the introduction of AGVs in the process removes the constraints that determine the current layout shapes, and enables infinitely many possibilities for new solutions. Thus, an algorithm to identify the best cellular workstations layout has been developed. Second, there are several ways to assign tasks to the vehicles and exploit the resulting benefits. Hence, simulation tools have been used to test different deployment scenarios, compare different technologies and evaluate their impact. These tools have been designed to support the design phase; nonetheless, the simulation model is a flexible tool and its deployment area can be extended to further purposes. It can be integrated into a MES and used even after the physical realization of the AGVs into the manufacturing line. The monitoring of vehicles activities is the input for the simulation, which can be used to evaluate future scenarios based on the current picture. This enables improved reactivity of the whole system and better exploitation of the flexibility provided by the vehicles.

# 8.3 Conclusive remarks

The results obtained in the case studies described above enable to state the following conclusions.

1. **The approach is validated**. The results obtained from the four case studies provide evidence that the deployment of information-based techniques enables to develop adaptive strategies for waste reduction and achievement of lean manufacturing.

In the last few years, we assisted to an outstanding proliferation of relatively inexpensive and very smart sensors. This phenomenon strongly impacted also our daily life: smart devices are spreading at a high pace. The diffusion of sensors in manufacturing is also taking place in manufacturing, both within companies and throughout the supply chain. This technological evolution must be supported by appropriate applications, able to deal with the generated quantity of data: the key challenge is to understand and exploit their value.

### 8.3 Conclusive remarks

MES is the platform in charge of collecting and analyze such data, and dispatch the right information at the right system, at the right time. Due to this primary role, the implementation of smart, responsive MES is mandatory; the only way to reach this result is the integration of powerful analytics. The analyses performed by the MES can foster the definition of strategies for process improvement leading to enhanced overall performance and reduced wastes; hence, MES is a powerful tool in supporting the path towards lean manufacturing.

2. The approach is independent of the case study. The three-steps methodology presented in Chapter 3 was an effective tool to successfully deal with the case studies studied in this work.

The definitions for the classes of waste and for the kind of parts cooperating into a process are general enough to be used in different manufacturing technologies, as well as in services supporting the manufacturing operations. Despite this generality, they are accurate enough to provide a detailed description and deeply understand the process to be studied. The same conclusion can be stated for the list of steps supporting the design of an effective data analysis technique.

The schematic presented in Figure 3.1 was an effective tool to develop the research work. It allows to point out the problem and identify the solution strategy by answering questions in the logical order Why - What - How, hence starting from an overall perspective and driving the user to a higher level of detail. The compactness of the schematic is another advantage: it allows to keep a global view of the problem in a single sheet, thus enabling the practitioner to immediately identify inconsistencies or information lacks.

- 3. The approach is independent of the solving techniques. The five steps defined to perform data-analysis have a general significance; nonetheless, their implementation must be focused on the specific case study and on the sources of waste affecting it and the desired output. In the presented case studies, different kind of data have been analyzed and different analysis techniques have been used, such as statistical analyses, probabilistic reasoning, simulations.
- 4. The data-oriented approach goes beyond tight departments. The definition of MES relies on its vertical integration with the business level and the shop floor. One of the aims of this work was the smart integration of MES with further information tools, such as design tools. Frameworks for the integration of MES with PLM and DFAM have been presented; the cooperation among such systems can enhance the integration of product and process engineering. Given this integration, data collected by the same source can be used with different purposes, with different scales of application. Nonetheless, the techniques implemented into the MES must be smart enough to extract information with the right level of detail necessary for each interlocutor.

### 8.4 Future work

#### Additive manufacturing: surface quality monitoring

At the state of the art, additive technology is experiencing a disruptive growth and is providing new paradigms in manufacturing and unprecedented design freedom. Nonetheless, this technology is not yet totally under control: there exist phenomena occurring during the process, at low time and space scales, that are not fully understood. Hence, systems for process monitoring and control would be appreciated to improve the control on the process and, in turn, its performance. The quality of the produced parts would also be enhanced. Nonetheless, a technological lack is slowing the development of these systems. A new generation of smart sensors able to detect such phenomena without interfering with the building process is required. The development of monitoring and control systems devoted to additive technologies would be much higher for applications in high value added market niches, such as aeronautics, biomedical devices or high performance products. Thus, one research direction for the next future is the development of new technologies and framework for better monitoring and control these systems.

#### Warehousing: management of resources and items

Two lacks have been identified in dealing with this work. The first one concerns the absence of proper standards and methods for performance evaluation. Hence, techniques able to define standard methods to evaluate the performance of a warehouse equipped with a AVS/RS system must be identified. The second lack is the absence of systems to real-time control the events taking place in the warehouse. At the state of the art, criteria for managing resources are implemented into the software governing the system. Nonetheless, this is a rigid approach that may provide non-optimal solutions in case the operating conditions are different from the ones hypothesized by the designer. Hence, flexible tools should be implemented, able to compare in real-time the performance resulting by the application of different criteria and to choose the best one. Future work should be aimed in solving these issues.

### Logistics: transport of items through workstations

Even in this case, the need for tools supporting the flexible implementation of management criteria is necessary. The case study chosen to proof the concept is in the field of automotive powertrain assembly, thus a quite rigid manufacturing process. The flexibility provided by automated vehicles per-

#### 8.5 Vision for the Factory of the Future

forming items transport through the shop floor is enhanced in case multiple products are manufactured on the same line. Given a target output (e.g. a quantity of demanded items per each kind of product), there exist a huge number of combinations to schedule the production, and the one minimizing wastes and risks (such as traffic and crossings) should be chosen. This is a scheduling problem typically faced by the MES; nonetheless, actually there are no techniques to perform this task considering traffic issues. Similarly, another factor that can play a significant role in traffic management is variability. Usually, variability is a source of waste; nonetheless, hastening or delaying the release of an item into the process may be useful to avoid crossings with other vehicles.

# 8.5 Vision for the Factory of the Future

The development of information technology drove the third industrial revolution, and the pace at which IT solutions are pervading manufacturing is still increasing. The next industrial revolution, based on Cyber Physical Systems (CPS), cannot disregard IT tools: they must support manufacturers to improve: (i) product quality and robustness of production processes; (ii) speed and time with regard to innovations, lead times and start-up of production plants; (iii) competitive production costs. Further, the deployed IT tools must be adaptive to a wide variety of product variants, be able to act online and provide visibility of a compound of plants or organizations, rather than a single location (Rabbani et al., 2013). Beside this, the information tools for supporting the development of Factory of the Future must be able to comply with the following keywords.

### Distributed

This term can be used to characterize future factories on different scales.

From a high-level perspective, a company may have different shop-floors distributed on different geographical areas. This kind of distribution already occurred with mass production, in order to reduce the impact of finished items transport. Nonetheless, future factories will be much more distributed than today. The paradigm of *mini-factories* is emerging (Cluster Fabbrica Intelligente, 2015): the production of custom components is postponed until the "last mile" and carried out near to or at the place of delivery. This innovation requires new organizational models based on: (i) hubs for the production of standardized components; (ii) decentralized mini-factories, equipped with state-of-the-art machinery to support the aesthetic and functional personalization of the product quickly and cheaply, in order to guarantee companies the opportunity to differentiate their product by adding value to it.

The use of advanced, highly reconfigurable technologies able to adapt to the specific context is strategic for the implementation of this model. The deployment of additive technologies, which is already growing at a high pace, can support this paradigm. Additive machines are characterized by limited impact: they do not emit smoke or gas, and produce a low quantity of scraps (typically, only the support material). Further, noise is reduced and the room necessary to host a machine is usually limited. These advantages enable unprecedented freedom in machines allocation, and the deployment of several, smaller shop-floors distributed over the target markets can become economically convenient. This distribution would enhance the competitiveness and the reactivity of a company: transport time and cost are reduced, as well as the lead time for the customer.

In this perspective, an advanced concept has been developed by Amazon (2015): a solution was patented for installing additive machines on a truck to perform, at the same time, product manufacturing and delivery. A customer may order an item through the electronic market; then the delivery truck to be used is chosen by optimizing a set of criteria, such as the time necessary for part production and the time to reach the customer from its actual position. The truck receives the manufacturing instructions from a central library in which all the information concerning each item on sale is contained.

Another promising concept concerning the market of replacement parts has been patented by Boeing (2015). Actually, when a replacement part is desired, a customer requires it to the aircraft manufacturer, which should maintain an inventory to satisfy the demand in short times. Nonetheless, the delivery can take an excessive amount of time for the customer, which may keep an inventory of parts on hand to avoid waiting for delivery from the manufacturer. However, this storage can require a huge amount of resources. The deployment of additive technology can strongly support the effectiveness of the market of replacement parts. The patented concept concerns a system to uncouple the management of the inventory and the storage of replacement parts: the manufacturer maintains a library containing all the information necessary to produce each part. When a customer needs a part, he requires the corresponding file to the manufacturer and performs, by himself, the production of the part through an additive machine.

From a low-level perspective, even workstations into the shop-floor can be distributed. Recent technologies, such as the AGVs, already allow to replace rigid transport systems like conveyors. This change enables a huge variety of innovations in term of layout design, as shown in Chapter 7. However, a rigid classification for the times that an item spends in the shopfloor still exists: it is made of processing, transport and waiting. The last two classes do not add value to the product, hence must be minimized; a techno-

#### 8.5 Vision for the Factory of the Future

logical effort is necessary to transform them into productive times. One way to explore this research direction is to transform the transportation means (e.g. the AGVs) into equipped workbenches able to move and to perform at least some elementary operations at the same time; as a consequence, workstations would be used only to perform complex tasks. Such innovation can lead to a further distribution of the process: manufacturing operations would take place throughout the whole shop-floor – even in places where there are no machines – rather than in pre-determined points.

To exploit this direction, technological efforts are necessary to physically develop innovative solutions. Nonetheless, their exploitation goes hand in hand with the contribution of optimization techniques for operations management and execution. On one side, the distribution of working areas, both at the high and the low level, enables to redesign and reduce transport of physical entities. On the other side, the transport of data and information increases, and the availability of well-structured, high-performance information tools can boost the innovation in this direction; their importance increases with the complexity of the system, in order to efficiently perform the requested tasks. In this regard, MES as a platform for data transformation and sharing has a strategic role.

### Virtualized

The upcoming forth industrial revolution is triggered by the Internet, which makes feasible disruptive scenarios in communication. It concerns the deployment of CPS, that allow the integration of computation with physical processes. Embedded computers and networks monitor and control the physical processes, usually with feedback loops where physical processes affect computations and viceversa (Lee, 2008). The interaction between the controller and the physical process is performed through a set of sensors and controllers. CPS can regard machines as well as storage systems or production facilities.

In order to best exploit CPS, two complementary and parallel approaches must be undertaken: cyberizing the physical and physicalizing the cyber (Reiner, 2014). The first step to undertake this research direction is the creation of a virtual copy of all the elements of the physical factory. Then, simulation models must be implemented to transform the static models towards dynamic systems; the full virtual factory should represent both the logical schemas and the behavior of real manufacturing systems.

For long time, the virtual representation has been mainly restricted to design purposes: Virtual Models (VM) for the factory were created before the implementation of the real factory to better explore different design options, evaluate their performance and virtually commission the automation systems. This approach enables to reduce the time-to-production. Nonetheless, virtual models can also be maintained and deployed throughout the whole lifetime of the production: they are an effective connection with the shop-floor and allow to exploit CPS. The integration of VM into CPS enables the fusion of virtual and physical worlds: virtual models can be fed with real data and simulate future scenarios without the need to input data manually (Westkämper and Jendoub, 2003). Complex and composite scenarios can be tested, even in real time, to evaluate their impact on the production system and take corrective actions to adapt the process and achieve the best performance. On the other hand, the evolution of the real factory must be stored in the virtual models to maintain them effective.

To fully exploit CPS, appropriate simulation tools are necessary. Today, the simulation environments for production processes are mainly used as stand-alone methods, using simplified representations of the process based on estimated data. Such simulations are mainly used to support design and for offline (re-)configuration of the production system. The functionalities of simulation tools must be extended to achieve an integrated platform able to support the optimal management of the production process throughout the entire life cycle; simulations must support closed-circuit online optimization and must be connected with CPS and the related automation for process reconfiguration (EFFRA, 2013).

Beside company resources, products, processed materials, systems, must be treated in the same way. Physical products should be transformed in uniquely identifiable information carriers, which may be whenever located to know their history, current status and alternative routes to achieve their target state.

Factory virtualization relies on data exchange between the real and the virtual realization of the factory. An effective MES implementation is strategic to efficiently perform data collection and analysis, and the appropriate information dispatching. MES is in charge of a fundamental role in the connection between the virtual and the real realizations of a process. By performing data collection and analysis, it enables the information exchange that promotes "cyberizing the physical and physicalizing the cyber".

#### Integrated

The term *integrated* can be analyzed from different perspectives too.

First, integration among information tools is necessary for the configuration of production systems. The design and the management of a production system involve a huge variety of skills, tools and responsibilities, and increasingly require the support of software tools implementing design and evaluation methods. In many cases, these software tools work separately, thus the efficiency of information exchange is reduced and the risk for errors or incomplete information is increased. Further, the implementation of

### 8.5 Vision for the Factory of the Future

integrated methodologies for management or configuration is not feasible. Hence, the availability of a platform supporting interoperability between various tools would be significantly appreciated. The first step towards this direction is, necessarily, a shared representation of the production system, which is continuously updated and consolidated both in the design and management phases to ensure overall coherence of the results.

A deeper integration must also be pursued within departments and plants belonging to the same company, spread in different geographical areas. The interaction between information tools and the development of virtual models enables operators and technicians physically far with each other to successfully cooperate, for example, to jointly design a product, or act on a given process to check its health state and take decisions aiming at problem solving and performance improvement. Also, the interaction between machines and humans can benefit by mobile devices and interaction devices: production and enterprise specific information can be transmitted regardless of their geographical location and adapted to the context and the specific skills or responsibilities.

The virtualization of factory environments also promotes a tighter integration among all the entities taking part in the value chain of a product: each of them can be enabled to access information concerning any stage of the path from the raw material to the finished product, and adapt its own process to the operating condition of the other companies. Further, in collaborative networks, suppliers and OEMs will be also able to sell their products as services; for example, remote service management helps to improve equipment uptime, reduce costs for servicing (e.g. travel costs), increase service efficiency (e.g. first-visit-fix-rates) and accelerate innovation processes (e.g. through remote update of device software). Further, customers may be involved in the value chain of a product: information can be extracted after sale; this information can be used to develop customized solutions for future products. The increased integration also supports the development of innovative, inclusive business models (Cluster Fabbrica Intelligente, 2015; EFFRA, 2013; IEC, 2015).

A successful integration, for each of the described levels, requires effective and reliable exchange of data and information. Thus, MES has a primary role in supporting integration.

### Mindful

One of the basic concepts for the definition of a smart factory is the capability in measuring data concerning the current state of the process, in realtime sensing instability conditions and in reacting instantaneously, guided by automated systems. This kind of decision making mimics human reactions (Chui et al., 2010). Innovative IT tools must support increased autonomy of the machinery in the usage phase, in terms of both maintenance and optimization of the process parameters; this enhances profitable production of mass customized and highly personalized products, as well as faster reactions to shifts of market demands.

There are two ways to undertake this path. The most deployed strategy is the identification of local optima, i.e. solutions that are optimal over short time terms or for a specific machine. An alternative, more complex strategy is the identification of global optima, i.e. the evaluation of solutions able to solve an issue or to improve the overall process performance on a longer time horizon.

The next generation information tools must be provided with "mindfulness" capabilities, and must be able to evaluate the impact of their actions. Intelligent systems based on mechanisms for condition prediction, estimation of remaining useful life, and self-learning capabilities will lead to increased reliability, availability and safety in the entire production system. For example, maintenance will be increasingly planned before the failure occurs, and when its impact is at a minimum. Therefore, predictive data analytics techniques should be developed and integrated into the MES to aggregate and process the massive amount of data captured on the shopfloor: complex event processing techniques for cause-effect and trend analysis are required to provide the decision-makers with a holistic overview of the process. The way in which information is spread is even more important: the right amount of information must be transmitted to the right person/department at the right time.

### 8.5.1 The role of the human

Smart software provide manufacturing companies with the potential to reshape their industries, increasing the productivity of engineers and plant operators, and generating more efficient manufacturing processes. These applications require high skilled workers. Thus, the diffusion of IT based applications in manufacturing environments will generate more knowledgeintensive jobs and will require more knowledge workers to perform them. Three main aspects need to be addressed to define the role of humans in future factories (EFFRA, 2013):

- how people will work and learn;
- how people will interact with technology;
- how people can add value to manufacturing.

In this work, the attention was focused on the transformation of Data into Information. The next step to be addressed, according to the DIKW paradigm (Ackoff, 1989), is Knowledge: information must be appropriately treated, understood and utilized by workers, at all levels, in the manufacturing processes throughout the whole value chain. To comply with this capa-

### 8.5 Vision for the Factory of the Future

bility, efficient learning processes are required, based on the adequate use of information models and storing mechanisms. IT solutions can also be useful for workforce training, to support the transfer of best practices to workers at the shop floor (for example, aiming to improve efficiency, productivity and reliability, or to prevent job risks). Such models and tools should support knowledge creation and learning at all levels (strategic, tactic, operation) for the entire product and factory life cycle. Another deployment area of IT is the creation of multimedia technical documentation, to support exchanges between OEMs and companies, concerning learning, deployment and recovery practices. Skilled experts are necessary both for developing the software, to implement the applications at manufacturing sites, and to deploy them in a production environment. Hence, on the one hand the level of competences required to work in manufacturing is increasing, as well as the role of human brain. On the other side, education will play a strategic role: the spread of automation will decrease the impact of labor cost on the final product price and on the overall competitiveness of a company; hence, manufacturers will be more interested in setting up units where an adequate level of education is available. This aspect can be crucial for companies reshoring.

IT systems can introduce new relations between human and the factory resources. In the past, such relations were static: a manufacturing schedule was created according to a business plan, and the necessary workforce was assembled. Workers were used to plan their life according to the manufacturing schedule, sacrificing their personal schedules and sometimes their health. Company productivity was limited by the degree to which workers could unite their minds with the factory (IEC, 2015). Future human-factory relations will become more dynamic through the use of advanced IT tools: robots can be used to improve the ergonomics in production, and automated machines can replace manpower in performing intensive, repetitive, low value added tasks; this enables workers to focus on knowledge-intensive activities. A broader interaction between humans and machines will be required to achieve manufacturing objectives. Today, static approaches are mainly deployed: on the shop-floor, some tasks are in charge of human operators; other ones are performed by automated machines or robots. Conversely, more dynamical models must be developed, together with techniques for a safe physical interaction between humans and machines. Dynamically changing systems can also adapt to temporary changes or limitations due, for example, to inexperience or inappropriate skills of an operator. The human-machine interaction may occur physically, as well as virtually: the increasing deployment of virtual models allows operators to interact with machines even from remote positions, and take decisions and actions to adjust the process.

Therefore, the way in which human can add value to manufacturing is not any more limited to providing mechanical energy to the process. The image in Figure 8.1 provides a good synthesis about the importance of human in manufacturing: his knowledge and intelligence will play a key role in the development of intelligent processes and factories (Cluster Fabbrica Intelligente, 2015; EFFRA, 2013).



Fig. 8.1 The role of human in future factories. Picture taken from The Economist.

# Appendix A Automated warehouses: results of the simulations

In the first part of this Appendix, full lists of the operations performed during some example cycles are provided to show the generality of the analytical models presented in Section 6.4. Then, graphical representations are provided for the content of the rack at the end of the simulations described in Section 6.7. Different criteria for UL storage and retrieval have been compared: they are synthesized in the table below. Further, the reaction to two scenarios have been simulated: the first one represent a steady-state situation, in which the number of stored and retrieved ULs is similar; the second one mimics a stressful situation, in which 50 items have to be stored as soon as possible.

In the Figures, different colors represent different ItemLots; further, the ID of the item is provided. IDs lower than 300 are assigned to ULs already in the rack at the beginning of the simulation; ULs with an IDs greater than 300 are stored during the simulation

Cimulation Nr.	Rack cri	Stored III a	Retrieved ULs	
Simulation INF.	Storage	e Retrieval		
1	Closest Level Lot	Closest Level	300	284
2	Closest Channel Lot	Closest Channel	300	284
3	Closest Level	Closest Level	300	284
4	Closest Level Lot	Oldest	300	284
5	Closest Channel Lot	Oldest	300	284
6	Closest Level	Oldest	300	284
7	Closest Level Lot	None	50	_
8	Closest Channel Lot	None	50	_
9	Closest Level	None	50	_

**Table A.1** The criteria used in the simulations to identify the best UL position for storage and retrieval in the rack.

Movement	Duration
Lift: from bay to avg. floor	Ум
Shuttle: from lift to avg. channel	$x_M$
Satellite: travel through the channel	$z_M$
Satellite: travel through the channel	$z_M$
Shuttle: from avg. channel to lift	$x_M$
Lift: from avg. floor to bay	${\mathcal Y}_M$
Total time spent:	$2y_M + 2z_M + 2x_M$

**Table A.2** Time necessary to perform a cycle for 1 storage (or 1 retrieval) with a system comprising 1 lift, 1 shuttle and 1 satellite.

Parameters	Value
Ι	1
T	0
Result from model 6.4	$2y_M + 2z_M + 2x_M$

**Table A.3** Time necessary to perform a cycle for 1 storage and 1 retrieval with a system comprising 1 lift, 1 shuttle and 1 satellite.

Movement	Duration
Lift: from bay to avg. floor	Ум
Shuttle: from lift to avg. channel	$x_M$
Satellite: travel through the channel	$z_M$
Satellite: travel through the channel	$z_M$
Shuttle: move through the aisle	$\delta x$
Satellite: travel through the channel	$z_M$
Satellite: travel through the channel	$z_M$
Shuttle: from avg. channel to lift	$x_M$
Lift: from avg. floor to bay	Ум
Total time spent:	$\frac{2y_M + 4z_M + 2x_M + \delta x}{2}$
-	

Parameters	Value
I	2
Т	1
Result from model 6.4	$2y_M + 4z_M + 2x_M + \delta x$

**Table A.4** Time necessary to perform a cycle for 2 storages (or 2 retrievals) with a system comprising 1 lift, 1 shuttle and 1 satellite, with the capability of uncoupling the movements of the shuttle and the satellite.

Movement Dura	tion
Lift: from bay to avg. floor $y_M$	
Shuttle: from lift to avg. channel $x_M$	
Shuttle and satellite work in parallel $S_{xz}$	
Shuttle: move through the aisle $\delta x$	
Satellite: travel through the channel $z_M$	
Satellite: travel through the channel $z_M$	
Shuttle: from avg. channel to lift $x_M$	
Lift: from avg. floor to bay $y_M$	
Total time spent: $2y_M$ -	$+2z_M+2x_M+\delta x+S_{xz}$

	Parameters Value
	I 2
	T O
	S 1
Result from	n model 6.5 $2y_M + 2z_M + 2x_M + \delta x + S_{xz}$

**Table A.5** Time necessary to perform a cycle for 2 storages and 2 retrievals with a system comprising 1 lift, 1 shuttle and 1 satellite, with the capability of uncoupling the movements of the shuttle and the satellite.

Movement Duration
Lift: from bay to avg. floor $y_M$
Shuttle: from lift to avg. channel $x_M$
Shuttle and satellite work in parallel $S_{xz}$
Shuttle: move through the aisle $\delta x$
Satellite: travel through the channel $z_M$
Satellite: travel through the channel $z_M$
Shuttle: move through the aisle $\delta x$
Satellite: travel through the channel $z_M$
Satellite: travel through the channel $z_M$
Shuttle: from avg. channel to lift $x_M$
Lift: from avg. floor to bay $y_M$
Total time spent: $2y_M + 4z_M + 2x_M + 2\delta x + S_{xz}$

Parameters Value
 I 3
T 1
S 1
 Result from model 6.5 $2y_M + 4z_M + 2x_M + 2\delta x + S_{xz}$



**Fig. A.1** Graphical representation for a system comprising one lift and two shuttles, in case the lift is the bottleneck (Eq. 6.8).



**Fig. A.2** Graphical representation for a system comprising one lift and two shuttles, in case the shuttle is the bottleneck (Eq. 6.8).


Fig. A.3 Content of the rack at the beginning of the simulation.



Fig. A.4 Content of the rack at the end of simulation nr. 1. Different colors denote different item lots.

A Automated warehouses: results of the simulations



Fig. A.5 Content of the rack at the end of simulation nr. 2. Different colors denote different item lots.



Fig. A.6 Content of the rack at the end of simulation nr. 3. Different colors denote different item lots.









Fig. A.7 Content of the rack at the end of simulation nr. 4. Different colors denote different item lots.



Fig. A.8 Content of the rack at the end of simulation nr. 5. Different colors denote different item lots.







Fig. A.9 Content of the rack at the end of simulation nr. 6. Different colors denote different item lots.







Fig. A.10 Content of the rack at the end of simulation nr. 7. Different colors denote different item lots.



Fig. A.11 Content of the rack at the end of simulation nr. 8. Different colors denote different item lots.



Fig. A.12 Content of the rack at the end of simulation nr. 9. Different colors denote different item lots.

16 x [m]

14

18 20 22 24

30

26 28

10 12

6 8

2

4

# Appendix B List of abbreviations

Acronym	Description
AGV	Automated Guided Vehicles
AM	Additive Manufacturing
AS/RS	Automated Storage and Retrieval Systems
AVS/RS	Autonomous Vehicle Storage and Retrieval System
BS	Bearing Seat
CAGR	Compound Annual Growth Rate
CIM	Computer Integrated Manufacturing
C-MES	Collaborative Manufacturing Execution System
CMM	Coordinate Measuring Machine
CMS	Cellular Manufacturing System
CPS	Cyber Physical Systems
DES	Discrete Event Simulation
DFM	Design For Manufacturing
DFAM	Design For Additive Manufacturing
DIKW	Data - Information - Knowledge - Wisdom
ERP	Enterprise Resource Planning
FIFO	First In First Out
ICP	Iterative Closest Point
IT	Information Technology
KBE	Knowledge Based Engineering
LIFO	Last In First Out
LS	Least-Squares
MDA	Machine Data Acquisition
MES	Manufacturing Execution System
MRP	Material Requirements Planning
MRP II	Manufacturing Resources Planning
PDCA	Plan – Do – Check – Act
PDM	Product Data Management
PLC	Programmable Logic Controller
PLM	Product Lifecycle Management

### B List of abbreviations

RFID	Radio-Frequency IDentification
SME	Small-Medium Enterprise
SMED	Single Minute Exchange of Dies
SPC	Statistical Process Control
SQC	Statistical Quality Control
TC	Target Channel
UL	Unit Load
VM	Virtual Models
VSM	Value Stream Mapping
WIP	Work In Process
WCS	Warehouse Control System
WES	Warehouse Execution System

WMS Warehouse Management System

# List of Figures

1.1	A synthesis of the functionalities integrated into an ERP	16
2.1 2.2 2.3	The functional levels defined in the standard ISA 95 Time trend of the deposited patents in the field of MES The structure for C-MES model	26 28 32
3.1	The schematic of the methodology used to develop this work. Black color represents information; green color is used for physical quantities	41
4.1 4.2	Methodology for the gear alignment case study The measurement system integrated into centering the	48
12	Example of macurements collected by the monitoring system	49
4.3 4.4	Schematic representation of the transformations applied to	49
1.1	the points	51
4.5	Comparison between the measurements acquired through a CMM and the monitoring system	54
4.6	Residual errors after the alignment of the workpiece	
	measured both through a CMM and the monitoring system	56
4.7	Synthesis of the operations performed by the monitoring and control system.	59
5.1	Timelines for industrial revolutions, traditional	
0.11	manufacturing and additive manufacturing technologies	68
5.2	Focus on the timeline for AM development	68
5.3	Hype Cycle for AM technology.	69
5.4	Design for DFAM methodology (Rosen, 2007)	71
5.5	Design for DFAM methodology (Ponche et al., 2012)	71
5.6	The toy duck used as a case study.	73
5.7	Methodology for the surface defects identification.	74
5.8	Pseudo-code of the algorithms for measurements analysis	76

5.9 5.10	Results of the simulation for the integrity test algorithm 77 Results of the simulation for real and expected geometry
	comparison
5.11	Proposed framework for MES-DFAM integration
6.1	A representation of the rack and the AVS/RS system
0.2	evaluation
71	Methodology for driving shop-floor redesign 111
72	Interaction forces involving each working cell 114
7.3	Equilibrium configuration of the workstations for the
	srudied BOP
7.4	FlexSim model implemented to evaluate the AGVs
	performance
7.5	Distribution of average time spent by each AGV in different
	activities for the first scenario 120
7.6	Distribution of average time spent by each AGV in different
	activities for the second scenario
7.7	Comparison between the takt times reachable with the two
70	simulated scenarios
7.8	Comparison between the takt times reachable with the two
	simulated scenarios 123
8.1	The role of human in future factories
A.1	Graphical representation for a system comprising one lift
	and two shuttles, in case the lift is the bottleneck (Eq. 6.8) 144
A.2	Graphical representation for a system comprising one lift
	and two shuttles, in case the shuttle is the bottleneck (Eq. 6.8) 144
A.3	Content of the rack at the beginning of the simulation
A.4	Content of the rack at the end of simulation nr. 1. Different
	colors denote different item lots
A.5	Content of the rack at the end of simulation nr. 2. Different
	colors denote different item lots
A.6	Content of the rack at the end of simulation nr. 3. Different
	colors denote different item lots
A.7	Content of the rack at the end of simulation nr. 4. Different
	colors denote different item lots
A.8	Content of the rack at the end of simulation nr. 5. Different
	colors denote different item lots
A.9	Content of the rack at the end of simulation nr. 6. Different
	colors denote different item lots
A.10	Content of the rack at the end of simulation nr. 7. Different
	colors denote different item lots

158

### LIST OF FIGURES

A.11 Content of the rack at the end of simulation nr. 8. Different	
colors denote different item lots	153
A.12 Content of the rack at the end of simulation nr. 9. Different	
colors denote different item lots	154

# List of Tables

2.1	List of the top patents assignees in the field of MES	29
2.2	Patents portfolio of the top MES vendors	29
4.1	Residual errors after the alignment of a simulated workpiece with 30 different initial positions.	53
4.2	Comparison between the measurements acquired through a	
12	CMM and the machine monitoring system.	55
4.5	both through a CMM and the machine monitoring system	55
4.4	Residual errors after the alignment of two workpieces	
	measured through the monitoring system of the	
	manufacturing machine	57
4.5	Results of the alignment strategy on 12 workpieces	60
6.1	Synthesis of the models for analytical evaluation of	
	automated warehouses performance	95
6.2	Performances of the machines involved in the case study	96
6.3	Statistical empirical distributions for the interaction between	
	satellites and rack positions	97
6.4	Results of the performance benchmarking performed	
	through the analytical model	97
6.5	Synthesis of the parameters describing the AVS/RS used as	
	case study.	100
6.6	Criteria used in the simulations.	101
6.7	The results of the simulations performed, according to the	
	criteria in Table 6.6	102
6.8	Results of the performed simulations in case 50 ULs need to be stored simultaneously	103
7.1	Bill of operations of the current process for cylinder head assembly.	112

7.2	Description of the working clusters composing the cellular
	layout to be optimized115
7.3	Equilibrium positions for the working cells
7.4	Synthesis of the performance for a system in which AGVs
	are free to assist any item
7.5	Synthesis of the performance for a system in which AGVs
	are constrained to assist a given item
A.1	The criteria used in the simulations to identify the best UL
	position for storage and retrieval in the rack
A.2	Time necessary to perform a cycle for 1 storage (or 1 retrieval)
	with a system comprising 1 lift, 1 shuttle and 1 satellite 142
A.3	Time necessary to perform a cycle for 1 storage and 1
	retrieval with a system comprising 1 lift, 1 shuttle and 1
	satellite
A.4	Time necessary to perform a cycle for 2 storages (or 2
	retrievals) with a system comprising 1 lift, 1 shuttle and 1
	satellite, with the capability of uncoupling the movements of
	the shuttle and the satellite
A.5	Time necessary to perform a cycle for 2 storages and 2
	retrievals with a system comprising 1 lift, 1 shuttle and 1
	satellite, with the capability of uncoupling the movements of
	the shuttle and the satellite

162

## References

- J. Abellan-Nebot and F. Romero Subirón. A review of machining monitoring systems based on artificial intelligence process models. *The International Journal of Advanced Manufacturing Technology*, 47(1-4):237–257, 2010. doi:10.1007/s00170-009-2191-8.
- M. Abramovici. Future Trends in Product Lifecycle Management (PLM), chapter 64, pages 665–674. Springer Berlin Heidelberg, 2007. ISBN 978-3540698197.
- R. Ackoff. From data to wisdom. *Journal of Applied Systems Analysis*, 16:3–9, 1989.
- S. Akerley, D. Kartsen, and R. Reed. Aligning average axis of hollow workpiece cavity with rotary machine axis. Patent number US5274566, 1992.
- N. Alagumurthi, K. Palaniradjaa, and V. Soundararajan. Optimization of grinding process through design of experiment (DOE) – A comparative study. *Materials and Manufacturing Processes*, 21(1):19–21, 2007. doi:10.1080/AMP-200060605.
- F. Amato, F. Basile, C. Carbone, and P. Chiacchio. An approach to control automated warehouse systems. *Control Engineering Practice*, 13(10):1223– 1241, 2005. doi:10.1016/j.conengprac.2004.10.017.
- W. Anotaipaiboon, S. S. Makhanov, and E. L. J. Bohez. Optimal setup for five-axis machining. *International Journal of Machine Tools and Manufacture*, 46(9):964–977, 2006. doi:10.1016/j.ijmachtools.2005.07.046.
- L. Apsley, C. Bodell, J. Danton, S. Hayden, S. Kapila, E. Lessard, and R. Uhl. Providing services related to item delivery via 3D manufacturing on demand. Patent number US20150052024, 2015.
- S. Ariafar and N. Ismail. An improved algorithm for layout design in cellular manufacturing systems. *Journal of Manufacturing Systems*, 28(4):132– 139, 2009. doi:10.1016/j.jmsy.2010.06.003.
- ASTM F2792-12a. Standard technology for additive manufacturing technologies, 2012.
- ATOS. Product lifecycle management. Technical report, 2010.

- E. Atzeni and A. Salmi. Economics of additive manufacturing for end-usable metal parts. *The International Journal of Advanced Manufacturing Technology*, 62(9-12):1147–1155, 2012. doi:10.1007/s00170-011-3878-1.
- G. Barbato, P. Chiabert, G. D'Antonio, M. De Maddis, F. Lombardi, and S. Ruffa. Method for automatic alignment recovery of a spur gear. *International Journal of Production Research*, 2015. doi:10.1080/00207543.2015.1064180.
- P. Basiliere and M. Shanler. Hype Cycle for 3D printing. Technical report, Gartner, 2014. URL www.gartner.com/doc/2803426. Last accessed 2016, March 14th.
- P. J. Besl and N. D. McKay. Method for registration of 3-D shapes. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 14(2):239–256, 1992. doi:10.1109/34.121791.
- P. Blanc, I. Demongodin, and P. Castagna. A holonic approach for manufacturing execution system design: An industrial application. *Engineering Applications of Artificial Intelligence*, 21(3):315–330, 2008. doi:10.1016/j.engappai.2008.01.007.
- P. Bocij, A. Greasley, and S. Hickie. Business Information Systems: Technology, Development and Management for the E-Business. Financial Times Management, 4th edition, 2008. ISBN 978-0273716624.
- A. Boschetto and L. Bottini. Accuracy prediction in fused deposition modeling. *The International Journal of Advanced Manufacturing Technology*, 73(5-8): 913–928, 2014. doi:10.1007/s00170-014-5886-4.
- H. Boström, S. F. Andler, M. Brohede, R. Johansson, A. Karlsson, J. Van Laere, L. Niklasson, M. Nilsson, A. Persson, and T. Ziemke. On the definition of information fusion as a field of research. Technical report, IKI Technical Reports, 2007. URL http://www.diva-portal. org/smash/record.jsf?pid=diva2%3A2391&dswid=-262. Last accessed 2016, March 14th.
- Y. A. Bozer and J. A. White. Travel-time models for automated storage/retrieval systems. *IIE Transactions*, 16(4):329–338, 1984. doi:10.1080/07408178408975252.
- B. Bukkapatnam, Satish Clark. Dynamic modeling and monitoring of contour crafting – An extrusion-based layered manufacturing process. *Journal of Manufacturing Science and Engineering*, 129(1):135–142, 2006. doi:10.1115/1.2375137.
- H. S. Byun and K. H. Lee. Determination of the optimal build direction for different rapid prototyping processes using multi-criterion decision making. *Robotics and Computer-Integrated Manufacturing*, 22(1):69–80, 2006. doi:10.1016/j.rcim.2005.03.001.
- X. Cai, S. S. Heragu, and Y. Liu. Modeling and evaluating the AVS/RS with tier-to-tier vehicles using a semi-open queueing network. *IIE Transactions*, 46(9):905–927, 2014. doi:10.1080/0740817x.2013.849832.
- F. T. S. Chan, K. W. Lau, L. Y. Chan, and V. H. Y. Lo. Cell formation problem with consideration of both intracellular and intercellular move-

ments. *International Journal of Production Research*, 46(10):2589–2620, 2008. doi:10.1080/00207540500478843.

- C.-C. Chang, T.-H. Wu, and C.-W. Wu. An efficient approach to determine cell formation, cell layout and intracellular machine sequence in cellular manufacturing systems. *Computers & Industrial Engineering*, 66(2):438–450, 2013. doi:10.1016/j.cie.2013.07.009.
- J. F. Chatelain and C. Fortin. A balancing technique for optimal blank part machining. *Precision Engineering*, 25(1):13–23, 2001. doi:10.1016/S0141-6359(00)00050-7.
- X. Chen, Z. Xie, and L. Zheng. RFID-based manufacturing execution system for intelligent operations. In 16th International Conference on Industrial Engineering and Engineering Management, Beijing, China, 21-23 October 2009.
- H. K. H. Chow, K. L. Choy, W. B. Lee, and K. C. Lau. Design of a RFID casebased resource management system for warehouse operations. *Expert Systems with Applications*, 30(4):561–576, 2006. doi:10.1016/j.eswa.2005.07.023.
- M. Chui, M. Löffler, and R. Roberts. The internet of things. Technical report, 2010. URL http://www.mckinsey.com/insights/high\_ tech\_telecoms\_internet/the\_internet\_of\_things. Last accessed 2016, March 14th.
- CIMdata. Product lifecycle management. Empowering the future of business. Technical report, 2002.
- CIMdata. How being applied PLM is to support todays Technical report, 2011. dynamic enterprises. URL http://www.aras.com/plm-software/?n= aras-plm-npi-apqp-fmea-cmii-results-cimdata. Last accessed 2016, March 14th.
- Cluster Fabbrica Intelligente. Research and innovation roadmap. Technical report, 2015. URL http://www. fabbricaintelligente.it/wp-content/uploads/ Booklet-Fabbrica-Intelligente-2015-PAGINE-SINGOLE. pdf. Last accessed 2016, March 14th.
- J. Cottyn, H. Van Landeghem, K. Stockman, and S. Derammelaere. A method to align a manufacturing execution system with lean objectives. *International Journal of Production Research*, 49(14):4397–4413, 2011a. doi:10.1080/00207543.2010.548409.
- J. Cottyn, H. Van Landeghem, K. Stockman, and S. Derammelaere. The role of a manufacturing execution system during a lean improvement project. In 13th International conference on Modern Information Technology in the Innovation Processes of Industrial Enterprises (MITIP 2011), pages 317–326, Trondheim, Norway, 22-24 June 2011b.
- G. Curtis and D. Cobham. *Business information systems: analysis, design and practice*. Pearson Education, Harlow, UK, 2008. ISBN 978-0273713821.
- Q. Dai, R. Zhong, G. Q. Huang, T. Qu, T. Zhang, and T. Y. Luo. Radio frequency identification-enabled real-time manufacturing execution system: a case study in an automotive part manufacturer. *Interna*-

*tional Journal of Computer Integrated Manufacturing*, 25(1):51–65, 2012. doi:10.1080/0951192x.2011.562546.

- G. D'Antonio, J. Sauza Bedolla, P. Chiabert, and F. Lombardi. PLM-MES integration to support collaborative design. In 20th International Conference on Engineering Design, ICED15, Milano, 27-30 July 2015a.
- G. D'Antonio, F. Segonds, F. Laverne, J. Sauza Bedolla, and P. Chiabert. A framework for manufacturing execution system integration in an advanced additive manufacturing process. *International Journal of Product Lifecycle Management*, 2015b. (submitted).
- G. D'Antonio, F. Segonds, J. Sauza Bedolla, P. Chiabert, and N. Anwer. A proposal of manufacturing execution system integration in design for additive manufacturing. In *The IFIP 5.1 12th International Conference on Product Lifecycle Management (PLM15)*, Doha, Qatar, 19-21 October 2015c.
- L. De Bernardini. Manufacturing trends and the MES connection. Technical report, 2015. URL http://www.automationworld.com/ manufacturing-trends-and-mes-connection. Last accessed 2016, March 14th.
- A. P. Dillon and S. Shingo. A Revolution in Manufacturing: The SMED System. Taylor & Francis, 1985. ISBN 978-0915299034.
- C. Dimopoulos and A. M. S. Zalzala. Recent developments in evolutionary computation for manufacturing optimization: problems, solutions, and comparisons. *IEEE Transactions on Evolutionary Computation*, 4(2):93–113, 2000. doi:10.1109/4235.850651.
- DKSH. Digitization and the market expansion services industry: driving omni-channel growth. Fourth global market expansion services report. Technical report, 2014. URL http://www.rolandberger. asia/media/pdf/Roland\_Berger\_DKSH20140905.pdf. Last accessed 2016, March 14th.
- Z. Doubrovski, J. C. Verlinden, and J. M. P. Geraedts. Optimal design for additive manufacturing: Opportunities and challenges. In 16th Design for Manufacturing and the Life Cycle Conference, volume 9, pages 635–646, Washington, DC, USA, August, 2831 2011. doi:10.1115/DETC2011-48131.
- EFFRA. Factories of the future. Multi-annual roadmap for the contractual PPP under Horizon 2020. Technical report, 2013. URL http://www.effra.eu/attachments/article/129/ Factories%20of%20the%20Future%202020%20Roadmap.pdf. Last accessed 2016, March 14th.
- B. Y. Ekren and S. S. Heragu. Simulation based regression analysis for rack configuration of autonomous vehicle storage and retrieval system. In *Proceedings of the 2009 Winter Simulation Conference (WSC)*, pages 2405–2413, 13-16 December 2009. doi:10.1109/wsc.2009.5429733.
- B. Y. Ekren, S. S. Heragu, A. Krishnamurthy, and C. J. Malmborg. Simulation based experimental design to identify factors affecting performance of AVS/RS. *Computers & Industrial Engineering*, 58(1):175–185, 2010. doi:10.1016/j.cie.2009.10.004.

#### REFERENCES

- H. A. ElMaraghy, A. Barari, and G. K. Knopf. Integrated inspection and machining for maximum conformance to design tolerances. *CIRP Annals - Manufacturing Technology*, 53(1):411–416, 2004. doi:10.1016/S0007-8506(07)60728-8.
- European Commission Business Innovation Observatory. Smart factories: smart process applications. Case study 25. Technical report, 2014. URL http://ec.europa.eu/growth/industry/innovation/ business-innovation-observatory/files/case-studies/ 25-smf-smart-process-applications\_en.pdf. Last accessed

2016, March 14th.

- M. Faes, W. Abbeloos, F. Vogeler, H. Valkenaers, K. Coppens, T. Goedem, and E. Ferraris. Process monitoring of extrusion based 3D printing via laser scanning. In 6th International Conference on Polymers and Moulds Innovations (PMI), volume 6, pages 363–367, Guimaraes, Portugal, 10-12 September 2014. doi:10.13140/2.1.5175.0081.
- Forte Industries. Big picture. The evolution of the traditional WMS/WCS roles. Technical report, 2014. URL http://www.forte-industries.com/media/39404/FORTE\_Big%20Picture\_IAa.pdf. Last accessed 2016, March 14th.
- Freedom Technologies. Understanding manufacturing execution systems
  (MES). Technical report, 2012. URL http://www.freedomcorp.com/
  solutions/qad/White%20Papers/MES%20White%20Paper.pdf.
  Last accessed 2016, March 14th.
- Y. Fu and P. Jiang. RFID based e-quality tracking in service-oriented manufacturing execution system. *Chinese Journal of Mechanical Engineering*, 25 (5):974–981, 2012. doi:10.3901/cjme.2012.05.974.
- M. Fukunari and C. J. Malmborg. An efficient cycle time model for autonomous vehicle storage and retrieval systems. *International Journal of Production Research*, 46(12):3167–3184, 2008. doi:10.1080/00207540601118454.
- M. Garetti, M. Macchi, and S. Terzi. *Product lifecycle management: State-of-the-art, trends and challenges*. Taylor & Francis, London, 2007. ISBN 978-0415429122.
- I. Gibson, D. W. Rosen, and B. Stucker. *Additive Manufacturing Technologies*. Springer US, New York, 2010. ISBN 978-1441911193.
- T. Groff and T. Jones. *Introduction to Knowledge Management*. Routledge, 1st edition, 2011. ISBN 978-0750677288.
- E. Gunpinar and S. Han. Interfacing heterogeneous PDM systems using the PLM services. *Advanced Engineering Informatics*, 22(3):307–316, 2008. doi:10.1016/j.aei.2007.08.009.
- Y. Gupta, M. Gupta, A. Kumar, and C. Sundaram. A genetic algorithm-based approach to cell composition and layout design problems. *International Journal of Production Research*, 34(2):447–482, 1996. doi:10.1080/00207549608904913.

- A. Hafezalkotob, M. A. Tehranizadeh, F. Sarani Rad, and M. K. Sayadi. An improved DPSO algorithm for cell formation problem. *Journal of Industrial* and Systems Engineering, 8(2):30–53, 2015.
- P. Helo, M. Suorsa, Y. Hao, and P. Anussornnitisarn. Toward a cloud-based manufacturing execution system for distributed manufacturing. *Comput*ers in Industry, 65(4):646–656, 2014. doi:10.1016/j.compind.2014.01.015.
- W. J. Hopp and M. L. Spearman. Factory Physics. Waveland Press, 2011.
- T. Houy. ICT and lean management: Will they ever get along? *Communications & Strategies*, 59:53–75, 2005.
- G. H. Hu, Y. P. Chen, Z. D. Zhou, and H. C. Fang. A genetic algorithm for the inter-cell layout and material handling system design. *The International Journal of Advanced Manufacturing Technology*, 34(11):1153–1163, 2006. doi:10.1007/s00170-006-0694-0.
- Y. Huang, M. C. Leu, J. Mazumder, and A. Donmez. Additive manufacturing: Current state, future potential, gaps and needs, and recommendations. *Journal of Manufacturing Science and Engineering*, 137(1):014001:1–10, 2015. doi:10.1115/1.4028725.
- IEC. Factory of the Future. Technical report, 2015. URL http://www.iec. ch/whitepaper/pdf/iecWP-futurefactory-LR-en.pdf. Last accessed 2016, March 14th.
- IEC 62264. Enterprise-control system integration, 2013.
- Intelligrated. Seven order fulfillment strategies to fast track ecommerce business growth. Technical report, 2013. URL http://www.intelligrated.com/sites/default/files/ pdfs/INT\_eCommerce\_WhitePaper.pdf. Last accessed 2016, March 14th.
- ISA 95. Enterprise Control System Integration. Part 1: Models and Terminology, 2000.
- ISO 1101. Geometrical product specifications (GPS) Geometrical tolerancing – Tolerances of form, orientation, location and run-out, 2012.
- ISO 12180-1. Geometrical product specifications (GPS) Cilindricity. Part 1: Vocabulary and parameters of cylindrical form, 2011.
- ISO 12180-2. Geometrical product specifications (GPS) Cilindricity. Part 2: Specification operators, 2011.
- ISO 12181-1. Geometrical product specifications (GPS) Roundness. Part 1: Vocabulary and parameters of roundness, 2011.
- ISO 12181-2. Geometrical product specifications (GPS) Roundness. Part 2: Specification operators, 2011.
- ISO 12781-1. Geometrical product specifications (GPS) Flatness. Part 1: Vocabulary and parameters of flatness, 2011.
- ISO 12781-2. Geometrical product specifications (GPS) Flatness. Part 2: Specification operators, 2011.
- ISO 15531-1. Industrial automation systems and integration Industrial manufacturing management data Part 1: General overview, 2004.

#### REFERENCES

- B. Javadi, F. Jolai, J. Slomp, M. Rabbani, and R. Tavakkoli-Moghaddam. An integrated approach for the cell formation and layout design in cellular manufacturing systems. *International Journal of Production Research*, 51(20): 6017–6044, 2013. doi:10.1080/00207543.2013.791755.
- F. Jolai, R. Tavakkoli-Moghaddam, A. Golmohammadi, and B. Javadi. An electromagnetism-like algorithm for cell formation and layout problem. *Expert Systems with Applications*, 39(2):2172–2182, 2012. doi:10.1016/j.eswa.2011.07.030.
- R. Kia, F. Khaksar-Haghani, N. Javadian, and R. Tavakkoli-Moghaddam. Solving a multi-floor layout design model of a dynamic cellular manufacturing system by an efficient genetic algorithm. *Journal of Manufacturing Systems*, 33(1):218–232, 2014. doi:10.1016/j.jmsy.2013.12.005.
- J. Kletti. Manufacturing Execution System MES. Springer-Verlag, Berlin, 2007. ISBN 978-3540497431.
- R. Koreis. Three Dimensional Printing of Parts. Patent number US20150064299, 2015.
- J. P. Kruth. Material incress manufacturing by rapid prototyping techniques. *CIRP Annals - Manufacturing Technology*, 40(2):603–614, 1991. doi:10.1016/S0007-8506(07)61136-6.
- T. Kunugi and Y. Sasaki. Method of correcting misalignment of a workpiece on numerically controlled machine tool. Patent number US4942611, 1990.
- P.-H. Kuo, A. Krishnamurthy, and C. J. Malmborg. Design models for unit load storage and retrieval systems using autonomous vehicle technology and resource conserving storage and dwell point policies. *Applied Mathematical Modelling*, 31(10):2332–2346, 2007. doi:10.1016/j.apm.2006.09.011.
- F. Laverne, F. Segonds, N. Anwer, and M. Le Coq. Assembly based methods to support product innovation in design for additive manufacturing: An exploratory case study. *Journal of Mechanical Design*, 137:121701:1–8, 2015. doi:10.1115/1.4031589.
- E. A. Lee. Cyber physical systems: Design challenges. In 11th IEEE International Symposium on Object Oriented Real-Time Distributed Computing (ISORC), pages 363–369, 5-7 May 2008. doi:10.1109/isorc.2008.25.
- Z. Li, J. Gou, and Y. Chu. Geometric algorithms for workpiece localization. *Robotics and Automation, IEEE Transactions on*, 14(6):864–878, 1998. doi:10.1109/70.736771.
- F. Lombardi, G. Barbato, P. Chiabert, M. De Maddis, F. Ricci, S. Ruffa, G. D'Antonio, S. Milletari, and R. Siccardi. Metodo di recupero automatico dell'allineamento di un pezzo in lavorazione, in particolare una ruota dentata, e relativa macchina di lavorazione automatica. Patent application TO2014/A000066, 2014.
- C. J. Malmborg. Conceptualizing tools for autonomous vehicle storage and retrieval systems. *International Journal of Production Research*, 40(8):1807–1822, 2002. doi:10.1080/00207540110118668.

- C. J. Malmborg. Interleaving dynamics in autonomous vehicle storage and retrieval systems. *International Journal of Production Research*, 41(5):1057– 1069, 2003. doi:10.1080/0020754021000033887.
- C. Mançanares, E. de S. Zancul, J. Cavalcante da Silva, and P. Cauchick Miguel. Additive manufacturing process selection based on parts' selection criteria. *The International Journal of Advanced Manufacturing Technology*, 80(5):1007–1014, 2015. doi:10.1007/s00170-015-7092-4.
- R. Manzini, R. Accorsi, G. Baruffaldi, T. Cennerazzo, and M. Gamberi. Travel time models for deep-lane unit-load autonomous vehicle storage and retrieval system (AVS/RS). *International Journal of Production Research*, pages 1–19, 2016. doi:10.1080/00207543.2016.1144241.
- Markets and Markets. Manufacturing execution system market by deployment type, process industry (chemicals, food & beverages, life science, oil & gas, power), discrete industry (aerospace & defense, automotive, medical devices) and by region – Global forecast to 2020. Technical report, 2015. URL http://www.marketsandmarkets.com/Market-Reports/ manufacturing-execution-systems-mes-market-536.html. Last accessed 2016, March 14th.
- S. Mellor, L. Hao, and D. Zhang. Additive manufacturing: A framework for implementation. *International Journal of Production Economics*, 149(0): 194–201, 2014. doi:10.1016/j.ijpe.2013.07.008.
- MESA International. MES functionalities & MRP to MES data flow possibilities. Technical report, 1997. URL https: //services.mesa.org/ResourceLibrary/ShowResource/ 2cedfe75-daed-4b9c-b187-f421cf90fdd2. Last accessed 2016, March 14th.
- MESA International. Mesa's next generation collaborative MES model. Technical report, 2004. URL https:// services.mesa.org/ResourceLibrary/ShowResource/ aacc4930-6c55-41fa-a597-22ccdded594d. Last accessed 2016, March 14th.
- H. Meyer, F. Fuchs, and K. Thiesl. *Manufacturing Execution Systems (MES): Optimal Design, Planning, and Deployment*. McGraw-Hill Professional, 1st edition, 2009. ISBN 978-0071623834.
- E. Monk and B. Wagner. *Concepts in Enterprise Resource Planning*. Course Technology, 4th edition, 2012. ISBN 978-1111820398.
- R. Myers. *Classical and modern regression with applications*. PWS-Kent, Boston, 1990.
- K. Nagata, H. Himeno, H. Okitomo, K. Tsuruta, I. Murata, K. Kobayashi, S. Hosaka, K. Hironaka, and O. Torii. Method of centering hollow cylindrical workpiece. Patent number US4196523, 1980.
- L. Niewmierzycki. Workpiece alignment structure and method. Patent number US5452521, 1995.
- NIST, National Institute of Standards and Technology. Measurement science roadmap for metal-based additive manufacturing. Technical re-

port, 2013. URL http://www.nist.gov/el/isd/upload/NISTAdd\_ Mfg\_Report\_FINAL-2.pdf. Last accessed 2016, March 14th.

- T. Noda, S. Sakamoto, K. Ogura, and N. Kikuchi. Profile measuring instrument and profile measuring method. Patent number EP2149775, 2009.
- F. Nynke, B. M. d. K. Ren, and L. v. d. V. Steef. Linking warehouse complexity to warehouse planning and control structure: An exploratory study of the use of warehouse management information systems. *International Journal of Physical Distribution & Logistics Management*, 32(5):381–395, 2002. doi:10.1108/09600030210434161.
- T. Ohno. Toyota Production System: Beyond Large-Scale Production. Productivity Press, 1st edition, 1988. ISBN 978-0915299140.
- Omnify Software. ERP and PLM: defining their roles and creating a collaborative environment for success. Technical report, 2007. URL http://www.omnifysoft.com/Register/Default. aspx?type=1&id1=5. Last accessed 2016, March 14th.
- T. Osada. *The 5S's: five keys to a total quality environment*. Asian Productivity Organization, 1991. ISBN 978-9283311164.
- A. A. Perez, C. M. Haid, M. P. Doll, and F. W. Pieper. Automatic process control of additive manufacturing device. Patent number US2015/0045928A1, 2015.
- H. Perret and J. Philippi. Method for authomatic calibration of a device for generative production of a three-dimensional object. Patent number WO2014/180971A1, 2014.
- PLM Interest Group. PLM handbook for SMEs. Technical report, 2011.
- R. Ponche, J. Y. Hascoet, O. Kerbrat, and P. Mognol. A new global approach to design for additive manufacturing. *Virtual and Physical Prototyping*, 7 (2):93–105, 2012. doi:10.1080/17452759.2012.679499.
- R. Ponche, O. Kerbrat, P. Mognol, and J.-Y. Hascoet. A novel methodology of design for additive manufacturing applied to additive laser manufacturing process. *Robotics and Computer-Integrated Manufacturing*, 30(4):389– 398, 2014. doi:10.1016/j.rcim.2013.12.001.
- T. C. Poon, K. L. Choy, H. K. H. Chow, H. C. W. Lau, F. T. S. Chan, and K. C. Ho. A RFID case-based logistics resource management system for managing order-picking operations in warehouses. *Expert Systems with Applications*, 36(4):8277–8301, 2009. doi:10.1016/j.eswa.2008.10.011.
- QC software. Warehouse control system: orchestrating warehouse efficiency, 2008. URL http://www.qcsoftware.com/PDFs/ QCwhitepaperPreview.pdf. Last accessed 2016, March 14th.
- M. J. Rabbani, F. M. Ahmad, J. Baladi, Y. A. Khan, and R. A. Naqvi. Modeling and simulation approach for an industrial manufacturing execution system. In System Engineering and Technology (ICSET), 2013 IEEE 3rd International Conference on, pages 26–31, 19-20 Aug. 2013 2013. doi:10.1109/ICSEngT.2013.6650137.
- P. Rao, J. Liu, D. Roberson, Z. J. Kong, and C. Williams. Online real-time quality monitoring in additive manufacturing processes using heteroge-

neous sensors. *Journal of Manufacturing Science and Engineering*, 2015. doi:10.1115/1.4029823.

- Y. Rao, F. He, X. Shao, and C. Zhang. On-Line Simulation for Shop Floor Control in Manufacturing Execution System, volume 5315 of Lecture Notes in Computer Science, chapter 16, pages 141–150. Springer Berlin Heidelberg, 2008. ISBN 978-3540885160.
- A. Reiner. Industrie 4.0 Advanced engineering of smart products and smart production. In *Technological Innovations in the Product Development*. 19th International Seminar on High Technology, Piracicaba, Brasil, 19 October 2014.
- E. W. Reutzel and A. R. Nassar. A survey of sensing and control systems for machine and process monitoring of directed-energy, metal-based additive manufacturing. *Rapid Prototyping Journal*, 21(2):159–167, 2015. doi:10.1108/RPJ-12-2014-0177.
- J. Riezebos, W. Klingenberg, and C. Hicks. Lean production and information technology: Connection or contradiction? *Computers in Industry*, 60 (4):237–247, 2009. doi:10.1016/j.compind.2009.01.004.
- M. Rolón and E. Martínez. Agent-based modeling and simulation of an autonomic manufacturing execution system. *Computers in Industry*, 63(1): 53–78, 2012. doi:10.1016/j.compind.2011.10.005.
- D. W. Rosen. Computer-aided design for additive manufacturing of cellular structures. *Computer-Aided Design and Applications*, 4(5):585–594, 2007. doi:10.1080/16864360.2007.10738493.
- M. Rother and J. Shook. *Learning to See: Value Stream Mapping to Add Value and Eliminate Muda*. Taylor & Francis, 2003. ISBN 978-0966784305.
- J. Rowley. The wisdom hierarchy: representations of the DIKW hierarchy. *Journal of Information and Communication Science*, 33(2):163–180, 2007. doi:10.1177/0165551506070706.
- D. Roy, A. Krishnamurthy, S. S. Heragu, and C. J. Malmborg. Performance analysis and design trade-offs in warehouses with autonomous vehicle technology. *IIE Transactions*, 44(12):1045–1060, 2012. doi:10.1080/0740817x.2012.665201.
- P. Sagues, S. Gaudio, and T. Wong. Wafer aligner system. Patent number US6275742, 2001.
- E. Shehab, M. Sharp, L. Supramaniam, and T. Spedding. Enterprise resource planning: An integrative review. *Business Process Management Journal*, 10 (4):359–386, 2004. doi:10.1108/14637150410548056.
- J. M. Simao, P. C. Stadzisz, and G. Morel. Manufacturing execution systems for customized production. *Journal of Materials Processing Technology*, 179 (13):268–275, 2006. doi:10.1016/j.jmatprotec.2006.03.064.
- A. K. Sood, A. Equbal, V. Toppo, R. K. Ohdar, and S. S. Mahapatra. An investigation on sliding wear of FDM built parts. *CIRP Journal of Manufacturing Science and Technology*, 5(1):48–54, 2012a. doi:10.1016/j.cirpj.2011.08.003.
- A. K. Sood, R. K. Ohdar, and S. S. Mahapatra. Experimental investigation and empirical modelling of FDM process for compressive

strength improvement. *Journal of Advanced Research*, 3(1):81–90, 2012b. doi:10.1016/j.jare.2011.05.001.

- R. Soto, B. Crawford, E. Vega, and F. Paredes. Solving Manufacturing Cell Design Problems Using an Artificial Fish Swarm Algorithm, pages 282–290. Springer International Publishing, Cham, 2015. ISBN 978-3-319-27060-9.
- R. Sudarsan, S. J. Fenves, R. D. Sriram, and F. Wang. A product information modeling framework for product lifecycle management. *Computer-Aided Design*, 37(13):1399–1411, 2005. doi:10.1016/j.cad.2005.02.010.
- Y. Sugimori, K. Kusunoki, F. Cho, and S. Uchikawa. Toyota production system and kanban system. Materialization of just-in-time and respect-forhuman system. *International Journal of Production Research*, 15(6):553–564, 1977.
- Y. W. Sun, J. T. Xu, D. M. Guo, and Z. Y. Jia. A unified localization approach for machining allowance optimization of complex curved surfaces. *Precision Engineering*, 33(4):516–523, 2009. doi:10.1016/j.precisioneng.2009.02.003.
- R. Tavakkoli-Moghaddam, N. Javadian, B. Javadi, and N. Safaei. Design of a facility layout problem in cellular manufacturing systems with stochastic demands. *Applied Mathematics and Computation*, 184(2):721–728, 2007. doi:10.1016/j.amc.2006.05.172.
- R. Teti, K. Jemielniak, G. ODonnell, and D. Dornfeld. Advanced monitoring of machining operations. *CIRP Annals - Manufacturing Technology*, 59(2): 717–739, 2010. doi:10.1016/j.cirp.2010.05.010.
- J. R. Tumbleston, D. Shirvanyants, N. Ermoshkin, R. Janusziewicz, A. R. Johnson, D. Kelly, K. Chen, R. Pinschmidt, J. P. Rolland, A. Ermoshkin, E. T. Samulski, and J. M. DeSimone. Continuous liquid interface production of 3D objects. *Science*, 347(6228):1349–1352, 2015. doi:10.1126/science.aaa2397.
- E. J. Umble, R. R. Haft, and M. M. Umble. Enterprise resource planning: Implementation procedures and critical success factors. *European Journal of Operational Research*, 146(2):241–257, 2003. doi:10.1016/S0377-2217(02)00547-7.
- J. Valacich and C. Schneider. *Information Systems Today*. Prentice Hall, 5th edition, 2011. ISBN 978-0137066995.
- P. Valckenaers, H. Van Brussel, P. Verstraete, B. Saint Germain, and Hadeli. Schedule execution in autonomic manufacturing execution systems. *Journal of Manufacturing Systems*, 26(2):75–84, 2007. doi:10.1016/j.jmsy.2007.12.003.
- H. Van Brussel, J. Wyns, P. Valckenaers, L. Bongaerts, and P. Peeters. Reference architecture for holonic manufacturing systems: Prosa. *Computers in Industry*, 37(3):255–274, 1998. doi:10.1016/S0166-3615(98)00102-X.
- B. Vayre, F. Vignat, and F. Villeneuve. Designing for additive manufacturing. *Procedia CIRP*, 3(0):632–637, 2012. doi:10.1016/j.procir.2012.07.108.
- P. Verstraete, P. Valckenaers, H. Van Brussel, B. Saint Germain, K. Hadeli, and J. Van Belle. Towards robust and efficient planning execu-

tion. *Engineering Applications of Artificial Intelligence*, 21(3):304–314, 2008. doi:10.1016/j.engappai.2007.09.002.

- V. Vijayaraghavan, A. Garg, J. Lam, B. Panda, and S. S. Mahapatra. Process characterisation of 3D-printed FDM components using improved evolutionary computational approach. *The International Journal of Advanced Manufacturing Technology*, pages 1–13, 2014. doi:10.1007/s00170-014-6679-5.
- H. Wang, S. Chen, and Y. Xie. An RFID-based digital warehouse management system in the tobacco industry: a case study. *International Journal of Production Research*, 48(9):2513–2548, 2010. doi:10.1080/00207540903564918.
- M. L. Wang, T. Qu, R. Y. Zhong, Q. Y. Dai, X. W. Zhang, and J. B. He. A radio frequency identification-enabled real-time manufacturing execution system for one-of-a-kind production manufacturing: a case study in mould industry. *International Journal of Computer Integrated Manufacturing*, 25(1): 20–34, 2012. doi:10.1080/0951192x.2011.575183.
- P. Ward and H. Zhou. Impact of information technology integration and lean/just-in-time practices on lead-time performance. *Decision Sciences*, 37(2):177–203, 2006. doi:10.1111/j.1540-5915.2006.00121.x.
- E. Westkämper and L. Jendoub. Smart factories Manufacturing environments and systems of the future. In *The 36th CIRP-International Seminar on Manufacturing Systems*, Saarbruecken, Germany, 2003.
- Wohlers Associates. 3D printing and additive manufacturing state of the industry. Annual worldwide progress report. Technical report, Markets and Markets, 2014. URL http://www.wohlersassociates.com/ 2014report.htm. Last accessed 2016, March 14th.
- J. P. Womack and D. T. Jones. Lean thinking: banish waste and create wealth in your corporation. Lean Enterprise Institute. Simon & Schuster, 1996. ISBN 978-0684810355.
- J. P. Womack, D. T. Jones, and D. Roos. *The machine that changed the world*. Free Press, 1990. ISBN 978-0743299794.
- X. Wu, C.-H. Chu, Y. Wang, and W. Yan. A genetic algorithm for cellular manufacturing design and layout. *European Journal of Operational Research*, 181(1):156–167, 2007. doi:10.1016/j.ejor.2006.05.035.
- H. T. Yau and C. H. Menq. A unified least-squares approach to the evaluation of geometric errors using discrete measurement data. *International Journal of Machine Tools and Manufacture*, 36(11):1269–1290, 1996. doi:10.1016/0890-6955(95)00075-5.
- L. Zhang, A. Krishnamurthy, C. J. Malmborg, and S. S. Heragu. Variancebased approximations of transaction waiting times in autonomous vehicle storage and retrieval systems. *European Journal of Industrial Engineering*, 3 (2):146–169, 2009. doi:10.1504/ejie.2009.023603.
- R. Y. Zhong, Q. Y. Dai, T. Qu, G. J. Hu, and G. Q. Huang. RFID-enabled real-time manufacturing execution system for mass-customization pro-

### REFERENCES

duction. *Robotics and Computer-Integrated Manufacturing*, 29(2):283–292, 2013. doi:10.1016/j.rcim.2012.08.001.